



OTHER SYMBOLS:

6FX11132AA01, 6FX1113 2AA01, 6FX1113-2AA01

RGB ELEKTRONIKA AGACIAK CIACIEK SPÓŁKA JAWNA

Jana Dlugosza 2-6 Street 51-162 Wrocław Poland

➡ biuro@rgbelektronika.pl▲ +48 71 325 15 05





www.rgbautomatyka.pl

www.rgbautomatyka.pl www.rgbelektronika.pl

YOUR PARTNER IN MAINTENANCE



At our premises in Wrocław, we have a fully equipped servicing facility. Here we perform all the repair works and test each later sold unit. Our trained employees, equipped with a wide variety of tools and having several testing stands at their disposal, are a guarantee of the highest quality service.



SINUMERIK System 3

Service Manual

Part 1

for Versions 0 through 3: for Model 0 to 2: Up to Software edition 09 for Model 3: Up to Software edition 05

Edition 12/85

Siemens Corporation

Start-up Check List	4 ¹⁰
Lists and Tables	2
Prerequisites and Visual Checks	3
Voltage and Function Tests	doauto
Machine Data Input	4
Standard Start-up	5
Drive Adjustment	6
Machine Data Description	<u> </u>
Interface	8
Board Overview and Strappings	9
Engineering Panel and Test Boards	10
PC Information	<u>8</u> 11
Modifying PC Programs via the NC Operator Panel with Test Boards	12
Alarm Description	13
Version 0 Notes and Start-up Information	14 N. C.
	15
16 ² 16 ²	16

SINUMERIK R - Documentation

Key to editions

The editions listed below have appeared prior to the present issue.

The "Revisions" column contains a list of the sections which have been revised, in each caxe with reference to this present edition.

Edition A.12.85 <u>Ordern-No</u>. E80210-T147-X-A-7600

Revisions

System 3 (I)

Table of contens

Page

1	Start-up Check List	1-1
1.1	Check List	1-1
1.2	Machine Data	1-3
1.3	Options available according to shipping notice	1-11
1.4	Jumpers on the I/O Boards	1-15
	S ALC ALC ALC	
2	Lists and Tables	2-1
2.1	Standard Machine Data Bits	2-1
2.2	Machine Data List with Standard	
	Machine Data Axis-Specific	
	Machine Data (Test)	2-2
2.3	Machine Data Bits (Test)	2-5
2.4	Display (Test)	2-9
2.5	Setting Data: For Versions 0 and 2	2-10
2.5.1	Input and Display: Zero Offset, S-max., M19	
	Setting Data and R-Parameters	2-10
2.5.2	Setting Data Bits (Operator Data)	2-10
2.6	Setting Data for Version 3	2-11
2.6.1	Display and Input Zero Offset, S-max, M19 Setting	
	Data Bits and R-Parameters	2-11
2.6.2	Setting Data Bits Nr. 25 (Operator Data)	2-11
2.6.3	Setting Data Bits Nr. 26 (Operator Data)	2-12
2.7	Alarm List	2-13
2.8	Interface Signals	2-15
3	Prerequisites and Visual Checks	3-1
3.1	Preliminary Remarks	3-1
3.2	Prerequisites, which must be met by the customer	
	before Start-up:	3-3
3.3	Supply Voltages	3-4
3.3.1	Power Supply Components	3-4
3.3.2	Line Conditions	3-4
3.3.3	Power Supply Connection-Logic Components	3-5
3.3.4	Line Connection for Fans	3-6
3.3.5	Operator Panel Power Sypply	3-6
3.3.6	Machine Control Panel 03 630	3-7
3.4	Visual Inspection	3-8
3.4.1	Grounding	3-8

0-1

System 3	(I)	0−2	A.12.85	Reg
3.4.2	2	Position Coders	3-8	C. Sel
3.4.3	3	Cabling	3–8	
3.4.4	L. W. OF	Shielding	3-8	
3.4.5	5	Operator Panel	3-8	
3.4.6	5	General State	3-8	
3.4.7	7	Battery in Power Sypply	3-9	
3.4.8	3	Cables	3-9	
3.5		NC-Boards and Strappings	3-10	
3.5.1	A AT	Identification System and Generalities	3-10	
3.5.2	2	Logic Component Assembly	3-11	
3.5.3	3	Strappings	3-15	
3.5.4	l	Other SINUMERIK Input and Output Boards	3-15	1
3.5.5	5	SIMATIC S5 Input and Output Boards	3-15	
3.6		Software System	3-15	1
3.6.1	14	Executive Software	3-15	,
3.6.2	2	SINUMERIK System 3 - Software Designation System	3-16	<u> </u>
3.6.3	3	Operating System Version 0 and 2	3-18	
3.6.4	1	Diagnostic System Version 0 and 2	3-18	
3.6.5	5 ALIGO	Operating System Version 3	3-19	
3.6.6	5 - 4 - 4 - 4	Diagnostic System Version 3	3-21	
3.6.7	7	PC-Diagnostic System for Version 3	3-21	
3.7		Remarks Concerning Version 1	3-22	
3.8		Remarks Concerning Version 0 and 2		
		with Software Edition 04	3-22	
4		Voltage and Function Test, Erasing the Memories,		
		Machine Data Input	4-1	~ ~
4.1		Voltage Test	4-2	
4.1.1	· _3	Voltage Supply	4-2	
4.1.2	. WIGH	Power-up Phase	4-2	\checkmark
4.1.3	-15	Fuses	4-2	
4.1.4		DC Voltage	4-3	
4.1.5	i	Error Storage while the NC is Switched Off	4-3	
4.2		Function Test	4-3	
4.2.1		CPU Cycle Monitoring on PC Board 03100 at Power-	up 4-3	
4.2.2	alasar.	CPU Cycle Monitoring during Operation	4-4	
4.2.3	i	Check SUM Test of the System Program Memory	4-4	
4.2.4	:	Adjusting the CRT Brightness in Version 3	4-5	
4.2.5	•	Remarks Concerning the CRT in Version 3	4-5	office a
4.2.6	S	Emergency-stop and Limit-switch Test	4-5	
		arman areas areas		

System 3 (I)	0-3	A.12.
4.3 John 4.3	Clearing the Memory Ranges (Cancel)	4-5
4.3.1	Cancel 0	4-6
4.3.2	Cancel 2	4-6
4.3.3	Cancel 3	4-6
4.3.4	Cancel 4	4-7
4.4	Loading the Standard Machine Data	4-7
4.5	Establishing the Control Type without	
	Changing the Machine Data	4-7
4.6	General Information Concerning the Input of	
	Machine and Setting Data	4-8
4.7	Constructing and Handling the Machine Data Tap	e 4-10
4.7.1	Constructing the Machine Data Tape	4-10
4.7.2	Preparing for Reading the Tape	4-10
4.7.3	Loading the Machine Data Tape	4-11
4.8	Example of Machine Data of a Lathe	4-12
10 ¹¹¹ 5 10 ⁰¹	Manual Start-up with System Software	5-1
5.1	Machine Data	5-1
5.2	Control Polarity of Feed Axes	5-1
5.3	Closing the Velocity Control Loop	5-3
5.4	Manual Run	5-4
5.5	Drift Compensation	5-5
5.6	Testing all Manual Functions	5-5 🔊
5.7	Running a Program	5-6
6	Drive Optimization, Drive Monitor and Finishin	g
	Touches	6-1
6.1	Tacho Adjustment and Definition of the Maximum	
	Command Value	6-1
6.2	Mult-gain Factor	6-2
6.3	Position Control Loop Gain (Kv Factor)	6-4
6.4	Acceleration TEST Nr. 120-123	6-7
6.5	Cut-off Velocity	6-8
6.6	Position Monitors	6-9
6.7	Contour Monitouring	6-10
6.8	Analogue Spindle Speed	6-13
6.9	Finishing Touches	6-15
6.9.1	Function Tests with NC Test Program	6-15
6.9.2	Generating a Machine Data Test	6-16
		Ann IS

A.12.85

System 3 (I)	0-4	A.12.85
6.9.3	Machine Data Tape with Machine Standard Data	6-16
6.9.4	Brief Instruction to the Customer	6-17
6.9.5	Start-up Report	6-17
6.9.6	Scheck-list for the Log Book	6-17
6		
7	Machine Data Description	7-1
7.1	Generalities	7-3
õ 7.2 Ö	Axis-specific Machine Data	7-3
7.3	Common Machine Data	7-18
7.4	Machine Data Bits Description	7-39
7.4.1	Address Designations	7-39
7.4.2	Time Settings	7-40 🗸
7.4.3	Reference Point Approach Direction	7-41
7.4.4	Sign Change for Velocity Command Value	7-41 🗸
7 4 5	Multiplication Pactor for the Dartial Actual	· · · · ·
1.3.5	Value	7-42
7.4.6	Rotary Avia	7-44
7.4.7	Exclude the Avis in the Start Interlock	7-44
7 4 8	Spindle with BOD-Frror	7-44
7.4.9	Speed in 0.1 rpm NA07 B	7-44
7 4 10	Reference Point	7-44
7 4 11	Auxiliary Function Output	7-45
7 4 12	Evaluation of the Programmed Data with	7-46
0 / · · · / 2 0	Seandlog	7-40
7 / 13	Position Control Food-back and Innut Custom	
7	Notrig/Inch	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~
7 1 11	No Decoloration at Limit Switch	7-47
7 4 15	Tool Longth Componention for non-programed	/-48
8 7.4.13	Avia (only for 2 ^m)	
7 4 16		7-50
7 4 17	Diameter Programing (only for 27)	7 F 4
7 4 18	PC-Present	7-51
7 4 19	Handwhaal Progent	7-51
7 4 20	Feed Bate not on the Contour	7-51
7 4 21	Option 2nd Input (Output Board	7-51
7.4.22	NC-PC Interface Activation	7-57
7.4.22	Key Switch Active for Operating Meder	7-52
7 4 24	I/O Interface V24 (BG 222 C)	7 52
7.4.25	ETA Code for \mathcal{O} Sign (A)	7-53
0	21. Oue for e bign (A)	1-22
		- St

Sys	tem 3 (I)	0-5	A.12.85
	7.4.25	Name of Main Axis Associated (B)	7-53
C Non Store	7.4.25	DC Control Character without Parity (C)	7-53
	7.4.26	Option Bits	7-54
	7.4.27	- Actual Values of X-Axis at 3T	
		Displayed in Diameter	7-55
	7.4.28	Block End with Carriage Return and Line Feed	7-55
	7.4.29	14 Bit DAC	7-55
	7.4.30	Wear Input in Diameter	7-55
	7.4.31	Spindle Override Active in Threading	7-55
	7.4.32	Deceleration to the Velocity of the Next Block \diamond	7-56
	7.4.33	Test Bits	7-56
U.10	7.4.34	Display Bits	7-56
	, d		
\sim	8	Interface Text	8-1
\sim	8.1	Signal Display	8-1
-	8.2	Pin-Layout of the Input and Output Connectors	8-3
	8.3	Code Tables	8-4
	8.3.1	Operating Mode Selector Switch (Gray-coded) S15	8-4
	8.3.2	Axis Selector Switch S18	8-5
	8.3.3	Feed Override Switch (Gray-coded) S17	8-5
	8.3.4	Spindle Override Switch (Gray-coded) S16	8-6
	8.3.5	Gear ranges (Coded Input Signal)	8-6
	8.4	Data of Signal Transmission	8-7
	8.5	Coupling of the Machine Control Panel 3M to The PC	8-8
	8.6	Machine Control Panel Connector Board with	
·		Option for Connection of Additional Operating	Š
C		Function	8-8
	8.7	Interface Adapter-Connector and Adapter,	
N. GOC		Position Control Diagnostic Connector	8-9
18 ¹⁰	8.7.1	Interface Adapter-Connector	8-9
	8.7.2	Interface Adapter	8-9
	8.7.3	Position Control Diagnostic Connector	8-9
			tornor.
	9	Board Overview and Strappings	9-1
	9.1	Generalities	9-2
	9.2	CPU Board 03100	9-3
	9.2.1	Fixed Strappings	9-3
	9.2.2	Variable Strappings	9-3
C Joanto	9.2.3	Test Socket P1	9-3
	9.2.4	Switches on the CPU	9-3
			M.

System	3	(I)

tem 3 (I)		0-6		A.12.85
9.3	EPROM Memory 0320	1		9-4
9.4	RAM Memory 03210			9-5
9.5	Test Board 03220			9-5
9.6	Test Board 03221			9-5
9.7	RAM Memory 03260			9-6
9.8	Position Control	Loop (Actual Val	Lue) 03310/03311	9-6
9.9 🥈	Position Control	Loop 03315		9-8
9.9.1	Fixed Strappings			9-8
9.9.2	Fixed Strappings	for EXE		9-8
9.9.3	Variable Setting	for Probes		9-9
9.10	Position Control	(Command Value)	03320	9-9
9.11	Position Control	(Command Value)	0332/03323	9-10
9.11.1	Fixed Strappings			9-10
9.11.2	Variable Strappir	ngs		9-10a
9.12	Position Control	03325		9-11 -
9.12.1	Fixed Strappings			9-11
9.12.2	Variable Strappir	ngs Command Value	e Output	9–12
9.12.3	Variable Strappir	ngs Servo-Ready-S	Simulation	9-12
9.12.4	Variable Strappir	ngs for Probe		9-13
9.12.5	Settings for Comm	and Value Output		9-13
9.13	Position Control	03340		9–13
9.14	Position Control	03350		9-13a
9.14.1	Fixed Strappings			9-13a
9.14.2	Fixed Strappings	for EXE		9-14
9.14.3	Variable Settings	s for Probes		9-14
9.14.4	Variable Strappir	ngs for Servo-Rea	ady-Simult.	9-14
9.14.5	Variable Strappir	ngs		9-14
9.14.6	Settings for Comm	and Value Output		9-15
9.15	Intergrated EXE			9-15 🔍
9.16	I/O Board 03400			9-15
9.16a	I/O Board 03	2 Contraction		9-15
9.17	Input Board		- Setter	9-16
9.18	Output Board 0342	21		9–16
9.19	I/O Board 03450			9-17
9.20	Output Board 0346	51		9-18
9.21	Power Supply 0350	00		9-19
9.22	Power Supply 0351	10		9–19
9.23	Operator Panel 03	3700		9–19
9.24	Operator Panel 03	3710		9-20
arease	North Contraction		ALAN MILLS	WWW.

Lennedon L

and Lich

9.25	Machine Control Panel 03720	9-20
9.26	Operator Panel 03770	9-20
9.27	Operator Panel 03780	9-20
9.28	Couple Board 03800 A	9-20
9.29	Video Interface 03810	9-21
9.30	Video Interface 03811	9-21
9.31	List of Strapping Possibilities	9-22
	(Addressing) of the I/O Boards	
	according to the PC Program	
	and	
10	Engineering Panel and Test Board	10-1
10.1	Prerequisites	10-2
10.2	Test Board	10-3
10.2.1	Construction	10-3
10.2.2	Test Board Front Plate	10-3
10.2.3	Designation of Switches and LED	10-4
10.3	States	10-4
10.3.1	Normal NC Operation	10-4
10.3.2	Deactivated Engineering Panel Progr.	10-4
10.3.3	Activated Engineering Panel Progr.	10-5
10.3.4	Engineering Panel Program-Hold-state	10-5
10.4	Functions	10-5
10.5	Sequence	10-6
10.5.1	Engineering Panel Activation	10-6
10.5.2	Display	10-7
10.5.3	Break Points	10-7
10.6	Engineering Panel Instructions	10-9
10.6.1	Display and Change of Memory Content	10-9
10.6.2	Display and Change of Register Contents	10-10
10.6.3	Break Point Setting	10-11
10.6.4	Single-Step Instruction	10-13
10.7	Input List	10-15
10.8	NC Address Lists	10-17
10.8.1	Overview for Versions 0 and 2	10-17
10.8.2	Overview for Version 3	10-18
10.8.3	Board 03200	10-19
10.9	Access to PC with the NC-Engineering Panel	10-20

System 3 (I)	0-8	A.12.85
		. one
10.9.1	Converting PC to NC Address	10-20
10.9.2	Reading and Setting of Inputs and Outputs	S. A.
	of the I/O Boards	10-20
10.9.3	Direct Access of the I/O Boards through	
	PC-Interface without PC, with the Aid of	
	the Engineering Panel	10-21
0 10.9.4	Display ISTACK with the Engineering Panel	10-22
11	Notes to the PC	11-1
11.1	Prerequisites, Settings for 130 WA	11-2
11.1.1	Assembly	11-2
8 11.1.1.1	PC 130WA central processing unit 6ES5 - 921-3W	A 11-2
11.1.1.2	PC coupling board 03800 A	11-3 🤍
11.1.1.3	PC memory board 6ES5 - 350	11-3
11.1.2	Settings	11-3
11.1.2.1	CPU 6ES5 - 921	11-3
11.1.2.2	I/O boards 03400, 03410, 03450, 034560	11-4
11.1.2.3	S5 I/O boards 402-3, 445-3, 444-3, 432-3	11-4
11.1.2.4	Coupling Board 03800 A	11-4
11.1.2.5	Memory Board 6ES5 - 350	11-5
11.1.3	Function Test of the PC	11-6
11.1.3.1	PC-CPU	11-6
8 11.1.3.2	Coupling Board 03800 A:	11-6
11.1.3.3	User Program	11-7
11.2	Prerequisites, Settings for 130 WB	11-8
11.2.1	Assembly	11-8
11.2.1.1	PC 130W, CPU 6ES5 921-3WB	11-8
11.2.1.2	EPROM Modules	11-8
11.2.1.3	PC Coupling Board 03800 A	11-8
11.2.2	Settings	11-8
11.2.2.1	CPU 130 WB	11-8
11.2.2.2	EPROM Module	11-9
11.2.2.3	I/O Boards, 03800 A, as in section 11.1	11-9
11.2.3	Function test for the PC	11-9
11.2.3.1	PC-CPU	11-9
11.2.3.2	Coupling Board	11-9
11.2.3.3	User Program	11-10
11.3	Function Blocks	11-11
8 11.3.1	Overview	11-11

al and

System 3 (I)

11.3.2	Function Blocks Designation	ົ້ 11–12
11.3.3	Function Blocks Description	11-13
11.3.4	Designated Variables for Function Blocks	11-14
11.3.5	Flag List	11-15
11.4	Program Examples	11-16
11.4.1	Factory Program	11-16
11.4.2	Structure of the Factory Program	11-17
11.4.3	Shipping Program Printout	11-18
11.4.4	Example of Program Structure with FB12	11-20
11.4.5	Example of NC Oriented Function Blocks	11-21
11.5	Test Aids	11-22
11.5.1	Input/Output Signal Image on NC/CRT	11-22
11.5.2	PC Program Correction via NC	11-23
11.5.3	Access to PC via NC Engineering Panel	11-23
11.5.4	Service Unit 333-OBA	11-24
11.5.4.1	Application Area	11-24
11.5.4.2	Construction	11–24
11.5.4.3	I/O Board	11-25
11.5.4.4	Program Structure	11-26
11.5.4.5	Program Example	11-27
11.5.5	Service Unit 333 C	11-28
11.5.6	PG 670/PG675	11-28
11.6	Troubleshooting in the PC	11-29
11.6.1	Interrupt Stack (I-Stack)	11-29
11.6.2	Block Stack	11-34
11.7		
11.8	PC Lists	11-35
11.8.1	Memory Map PC 130 WA	َ [™] 11−35
11.8.2	Address List PC 130 WA	11-36
11.8.3	Memory Map Internal RAM 130WA/130WB	
11.8.4	System Range for Error Localization	11-38
	for 130 WA and 130 WB	
11.8.5	Memory Map 130 WB	s [∞] 11−39
11.8.6 🔊	Address List PC 130 WB	11-40
11.8.7	Instruction Sets	
12	PC Program Correction through NC Operator	
	Panel	12-1
10 1	Conoralition	× 10.0

System 3 (I)

A.12.85

AN ANY ANY ANY ANY ANY ANY ANY ANY ANY A		
		I I J
, i i i i i i i i i i i i i i i i i i i	Distance	14-5
14.6.2	Adaption of Travel Distance to Command	p
14.6.1	Drift Compensation	14-4
14.6.	Moving the Axes	14-4
14.5	Setting the Control Sense for the Axes	14-4
14.4.6	Maximum Speeds for Gear Range	14-4
14.4.5	Operating Mode Dependent Velocities	14-4
14.4.4	Velocity Adaption	14-3
14.4.3	Reference Point Values	14-3
14.4.2	Software Limit Switch	14-3
14.4.1	Maximum Axis Velocity	14-3
	Data before the First Travel	14-2
14.4	Adaptions of the Velocity Related Machine	^{9.4.} –
14.3	Adaption to the Machine	14-2
14.2	- Setting the Standard Machine Data	14-2
14.1	Prerequisites	14-2
14	Basic Version 0	14-1
9°-	the the the the the the the	
13.3	Alarm Description	13-4
13.2	Alarm List	13-2
13.1	Generalities	13-1
13	Alarm Description	13-1
	na ona ona ona ona	
12.3.2	Program Correction for Test Puposes	12-14
12.3.1	Display of ISTACK at PC-Stop	12-14
12.3	Examples of Use	12-14
12.2.8	Information Functions	12-13
12.2.7	PC-RAM Compression	12-13
12.2.6	Activation of Blocks Stored in PROMs	12-11
12.2.5	Search	12-10
12.2.4	Editor	12-7
12.2.3	PC Initial Clear	12-6
12.2.2	Up and Down Load of PC Program	12-5
12.2.1	Basic Display	12-4
12.2	Operation	12-4
12.1.3	Key functions	12-3
	Correction	12-2
12.1.2	Prerequisites and Activation of the Program	
12.1.1	Using the PC Program Correction	12-2

www.cobastonadka.pl

www.cpaitonadka.d

www.spauconastran

	System 3 (I)	0-11	À A.1
\smile			
	14.7	Functional Adaptions	14-5
	14.7.1	Handwheel Activation	14-5
	14.7.2	Device Coding for Data Input/Output	14-5
	14.7.3	Functional Options	14-5
	14.7.4	Automatic Determination of the Position	
	10 ²⁰¹¹	Loop Gain (Kv Factor)	14-6
	14.8	Remarks Concerning Erroneous Inputs and	
2		Erasing the Memory Ranges	14-6
	14.9	Conclusion	े 14−6
	14.10	List 1: Axis Specific Machine Data	14-7
	14.11	List 2: Common Machine Data	14-8
	14.12	List 3: Machine Data Bits	14-9
S.	14.13	List 4: Possible Devices for Data Input and	14-10
\sim	- d	Output	
		1.55 ⁴⁶	

1-1	1
-----	---

S	ΙN	UMER	RIK	SYSTEM	;
---	----	------	-----	--------	---

SINUMERIK 3T

Basic Version

3M

1.1 Check List

Note the preliminary remarks on page 3-1!

Fill out in pencil or complete a copy and place it in the log book.

Check each finished section with a "yes".

Enter all the required values in the appropriate places.

Explanations concerning the individual sections can be found in the start-up instructions.

OEM Start-up

End User Start-up

Name	Division	Date	Name	Division	Date 🗳
, <u>?</u>	0.01	from	И	્વે	from
Customer	Location	to	End User	Location	to
Star Star	and the second sec		ANNI C	and the second se	

- 1. Have the start-up prerequisites been met ?
- Visual checks : line connection, E-Stop,grounding, position measuring devices,cabling,shielding,operator panel,general state
- 3. Software system designation
- 4. Voltage Function Tests :

Input voltage on power supply 03500 :

Input voltage on operator panel 03700/03780 Input voltage on power supply components : 3x

- 5. Enter machine data, especially: travel range limitation, rapid traverse, feed rate, and spindle speed.
- 6. Position control loops of started axes : Axis velocity,tacho adjustment,mult-gain factor,position loop gain,acceleration,position monitors,position control loop monitor,analogue speeds;are all adjustments done and yes tested ?

Drive Adjustment

Axis	5	x	Y S	Z	4
Maximum velocity Vmax (mm/Min)			20		201
Tacho adjustment Umax at Vmax ()	1)		Ser 1		Se la companya de la
Position loop gain (m/min/mm):k	v	. 5	<u>्</u> य		

yes (

yes 🔿

V-

V-

٧~

yes 🤿

- 7. Have all manual functions been tested? Has the customer executed function tests with test tape ?
- 8. Make a machine data tape with printout. This tape should be placed near the control Include printout in the filled-out machine data list, and put in the log book Check the option list

Enter deviating strappings into the list

Has the customer been instructed concerning :

drift compensation, reference point adjustment , backlash compensation, entering these values into the machine data, generating a tape, and where this tape should be placed ?

Do you have a start-up form, and has it been signed by the customer ?

Has a copy of this check list been included in the log book ?

Signature

OEM Start-up End User Start-up yes

yes

yes

yes

yes

yes

yes

yes

1.2 Machine Data (complete, even if machine data tape and printout already exist)

MAG	HINE DATA S	YSTEM 3T	VERSIONS 0, 2		
Nr.*)	S	Explanation	Max. Value		
100	2	Position	32000		
101		tolerance	لس + سر		
110	28	Clamp limit	32000		
111	5	all	(+ mu		
120	······································	Acceleration	6000		
121		and the second sec	0.01 m/s^2		
130		Max. Velocity	15000		
131			mm/min +)		
140	20	Velocity command	2048/8192		
141		limit	VELO 1		
150	2	KV-factor	10000		
151		- Au	0.01 s ⁻¹		
160		Limit switch +	± 9999999		
161		J.C.	um N		
170		Limit switch -	± 9999999		
171			pum .		
180	14	Reference point	± 9999999		
181			µm .		
190	28	Backlash S	± 225		
191	4 <u>69</u>	compensation	µm +)		
200		Tool reference	± 9999999		
201		point value	µm (+)		
210		Reference point	± 9999		
211	24	shift	µm +)		
220	19.	Mult-gain	32000		
221		1	C x min/m		
230	S	Drift 🚫	+ 500/2000		
231	K.	compensation	VELO 1		

+) for the limit values for degrees, resp. inches, see the machine data description (sec. 7)

Machine Data Bits

		-1			Bi	t		
Nr.	7	16	5	4	3	2	1	0
N 400S	<u> -</u>				¢,			
N 401S				20				- 2
N 402S				5	0	0	0	0
N 403S	ĪŌ	0	8.				2	
N 404S	0	0	4				A.	
N 405S	0	0	0	0	0	0	0	0
N 406S	0	0	0	0	0	0	0	0
N 407S	8	0	0	0	0			
N 408S	9				to.			
N 409S	1	0		2			0	1
N 410S	Τ			6				20
N 411S	1		200				20	2
N 412S			250				NE	
N 413S	T	22				£	2	
N 414S	T	0	0	0	0	0	0	0
N 415S	1		1		1		1	
N 416S	0		0	0	0	1	1	1
N 417S	0	0	0	8			0	2
N 418S	0	0	0	0	0	0	0	0
N 419S	0	0	0	0	0	0	0	0

Nr.	S	Explanation	Max. Value
350		Cut-off	15000
		velocity	mm/min +)
351	200	Threshold for con-	15000
	and the second s	tour monitor 🚿	mm/min +)
352	100	Tolerance for	32000 +)
		contour	mm • Test 850
	S	. 8	125 • 1000
353		Dwell time for	16000
24		position monitor	ms
354		Velocity command	3000/12000
	200	value limit	VELO 1
355	S.	Circle end-point	32000 +)
	office.	monitor	Jum 🚽
356	J.	Threshold for com-	32000 +)
	S.	pensation motion	jum 🔗
357		Spindle drift	± 500
27.0		24	VELO 2
358		Dynamic smoothing	c
	2	exponent f.thread	2
359	S.	Maximum 🔬	0000
360	S.C.	speed for 🔊	7777
361	25	8	1/min 🔊
362	S	gear 🔗	8
363		ranges	and the second
364		39.	32
365			
366	2		2
371	and the second s	Manual feed 🛛 🔊	15000
372	S.	Man. rapid trav.	mm/min
373	a Ste	Ref.approach vel.	J.
374	8	INC speed	(+ گې
375		DRY feed	Sell .
376		Dwell time for	16000
		spindle inhibit	MS
377	Ś	Min. spindle	8192
	Nº.	motor speed 🛛 🔊	VELO 2
381	a construction of the second s	Software edition	(32000)
383	30	Increase update	30
	\$ ⁶⁶	time 📈	1/2 ms 🔊
385		2nd. software	± 9999999
255		limit switch X-	תינא_

 for axis-specific machine data, the 10°decade is the axis designation



1-5

(complete, even if machine data tape and printout exist)

Nr. †)	s	Explanation	Max. Value	Nr.	S	Explanation	Max. Value
100	<u></u>	Position tolerance	32000	250		Cut_off	15000
101	<u> </u>)2000 Mm)			velocity	mm/min +)
102	2	-		351	8	Threshold for con-	15000
103		- Ale			Nº I	tour monitor	mm/min +)
110		Clamp limit	32000	352		Tolerance for	32000 +
111			μm +)		150	contour	mm • Test 850
112		Se.	S	ð	20	S.	125 • 1000
113	Å.			353		Dwell time for	16000
120	1	Acceleration	6000	1		position monitor	ms
121		-	0.01 m/s ²	354		Velocity command	3000/12000
122	88	202	+)			value limit	VELO 1
123		- 	and the	355	· 23	Circle end-point	32000
130		Max. velocity	15000			monitor	Jum
131			mm/min	356	N.	Threshold for com-	32000
132		8°.	<u>(</u> ۲) +)	, Ô		pensation motion	jum 🔗
133	35	2	1997 - Contract - Cont	357	r	Spindle drift	± 500
140	22	Velocity command	2048/8192			10	VELO 2
141		limitation	VELO 1	358		Dynamic smoothing	5
142	28.				200	exponent f.thread	
143	[S.	S.	359	- Tez	Maximum 🔬	9999
150	<u> </u>	KV-factor	10000	360		speed for	
151		dpaule .	0.01	361	and the second s	gear	1/min 🔬
152			S '	362			. S
155	<u></u>	1	<u></u>	363		ranges	355
100		- Limit switch +	± 9999999	364			
101	~~~~~	~	hu 🖉	365	À	2	
102	2×.	NON C		300	100 N	Marriel Const	45000
170	÷	limit cuitch	+ 0000000	271		Manual Teed	15000
171	<u> </u>	- Switch -	± 9999999	272		Man. rapid trav.	
172		1000	P ^m	374	÷	TNC speed	
173		10°		375	<u> </u>	DRV feed	ALCO.
180	- Se	Reference point	± 9999999	376		Dwell time for	16000
181	1	-	UM			spindle inhibit	ms
182	8	~	r	377	8	Min. spindle	8192
183	2	No.X			NOX	motor speed	VELO 2
190	1	Backlash	± 255	381	18 N	Software edition	(32000)
191		compensation	µm 🚫	383	<u>3</u> 0	Increase update	30
192		1021	' ∞° +)		<u>e</u>	time 🔊	1/2 ms
193		1997 - 19	N. Co	A.		and i	all'
210	35 ⁵⁴	Reference point 🔬	± 9999	32			
211		shift	μm				
212		, à	+)				
213	201	NG	NG.				
220	l	Mult-gain	32000				
21			Cx min/m				
.22		. 800		6.			
2)				St.			
20			± 500/2000	350			
. <u>)</u> 22		compensation	VELU 1	1			
72	<u> </u>	. A					
11 1	19	NL ^O		1			

 +) limit values for degrees, resp. inches (see sec.?)
+) the 10° decade is the axis designation for axisspecific machine data ..0 : X-axis ..1 : Y-axis ..2 : Z-axis ..3 : 4th. axis

Machine Data Bits

Machin	e Da	ta B	its					
			E	it		en e		
Nr.	7	6	5	4	3	2	1	0
N 4005				8				
N 4015			Ex.		0	n	0	0
N 4025		~ ~	0					K.
N 4035		0					20	
N 4045	3	0				<	8	
N 4055	19. T	0				Stor.		
N 4065		0			4			
N 4075		0	0	0	0			
N 4085				ð.				
N 4095	1	0	1			0		0
N 410S		ć	0.					2
N 4115	[30						
N 4125	1.0	P			[S.	
N 4135					1	1	2	
N 4145		0	0	0	0	0	L	
N 4155	1			0	1	L	L	Ļ
N 4165		0	0	0	0	1	1	11
N 4175	0	0	0	×	ļ	0		1
N 4185	0	0	0	0	0			
N 4195	0	0	0	0	0		0	0

Do not change the given values.

ANNAL GOOLS

www.idautor

(complete, even if machine data tape and printout exist already)

N	· · · · · · · · · · · · · · · · · · ·	Evolution	Nay Value		· · · · · ·		Nov. V-1
Nr . =)	<u> </u>	Explanation	max. value		2	Explanation	Max. Value
100		Position tolerance	32000	359	- St	Maximum	9999
101		<u></u>	µm +)	360	-S ²	speed	1/min
110		Clamp limit	32000	361	<u> </u>	for	
111			<u>µm +)</u>	362	······	8 gear ranges	. S
120	and the second s	Acceleration	6000	363			and the second
121	19		0.01 m/s²	364		L.	24
130		Max. velocity	15000	365			
151	<u>~~</u>		mm/m1n +)	200	<u> </u>	- March 1 - C - L - C	45000
140		Velocity command	2048/8192	371		Manual feed	15000
141			VELUNI	372		Man.rapid traverse	e mm/min
120		KV-Tactor	10000	373	<u> </u>	Ket.approach vel.	
171		N		374		INL speed	, (° +)
100		LIMIT SWITCH +	± 99999999	375		UKY TEED	4(000
101	19- 19-	limit cuitch	μm + 00000000	1 270		-DWell time for	16000
170		LIMIT SWITCH -	± 99999999	700		spindle innibit	ms
1/1	<u></u>	0.6			13.8	Min. spindle	8192
100		Reference point	± 99999999	770		motor speed	VELU Z
101		Deathlash	um L 255	2/0	. S.	CUT-OTT Spindle	i/min
190		backlasn	± 200	770	<u></u>	speed for MIY	11:10/0
200		Tool reference point	+ 0000000	20		for M19	1/m1n/ 360
200		inor reference horn	± 999999999	700		Desition limit	degrees
210	1ª	Pafarance raint	μm +/	200		for M10	l Mail degree
211		chift	± 9999	701		TOF MIY	(72000)
220	<u></u>	Mult onin	<u> </u>	701	<u></u>	Jostopoo undate	()2000)
221		mult-yain	52000 [v min/m	202	and the	time	1/2 ==
230		Drift	+ 500/2000	1 205	. 6	2nd coftunes	+ 00000000
231		compensation	- JU0/2000		57	edition cuitch Y	- 999999999
350	3	Cut-off velocity	15000	396		Accoloration	72000
	ast and	cut-off verocity		397			12000
351		Ibrechold for	mm/min +)	388		for	4 IIIS
		contour monitor	15000	380			
352	R. Contraction	Tolerance for	32000	300		o year ranges	
1		contour	mmelest 850	301	20		
5		o o in court	125-1000	392	.6		
353		Dwell time for	16000	393	×		
		position monitor	ms	12221			
354	1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 -	Velocity command	3000/12000	and a			
	-24	value limit	VELO 1	5°			
355		Circle end-point	32000 +)	{			
	20	monitor	ALTR ALTR				
356		Threshold for com-	32000 +)	1			
5		pensation movement	μm				
357		Spindle drift	± 500				
		8	VELO 2	S.			
358	334	Dynamic smoothing	5	and and a	★) the 10°	decade is the axis	designation

+) limit values for degrees, resp. inches, (see sec.7)

(for Nr. 100 through Nr. 231)

Machine Data Bits

1	Bit										
Nr.	.7	6	5	े4	3_	2	1	0			
N 4005			2								
N 4015			2					\sim			
N 4025		S.			0	0	0	0			
N 4035	0						8				
N 4045	0				_	1991					
N 4055	0	0	0	0	0	0	0	0			
N 4065	0	0	0	0	0	0	0	0			
N 4075				ð.		<u> </u>	[
N 4085			Nº			<u> </u>	[
N 4095	1	0	87 °		1		0	2			
N 4105		8									
N 4115	20	ř.					S.				
N 4125	2					15					
N 4135				I	3	<u> </u>					
N 4145		0	0	0	0	0	0	0			
N 4155	1		1	5	1		1				
N 4165			0	2		1	1	1			
N 4175	0	0	0				0	1			
N 4185	0	0	0	0	0	0	0	0			
N 4195	0	្ល	0	0	0	0	0	0			

Do not change the given values

32

Array Good

1-9

(complete, even if machine data tape and printout exist)

Nr.*)	S	Explanation	Max. Value	Nr.	S	. Explanation	Max. Value
100	324	Position tolerance	32000	350		Cut-off velocity	15000
101			(+ nm (mm/min
102	S.	<u></u>	·	351	2	Threshold for	mm/min
03		Nº.			No	contour monitor	15000
10		Clamp_limit	32000	352	- C ²	Interance for	32000
11	f		и т (+)		. 5 ⁰	contour monitor	mm.Test 8
12		100	200				125,1000
13		1 × 1		353		Dwell time for	16000
20	- All	Acceleration	6000			Desition monitor	10000
21			$0.01 m/c^2$	354	·····	Volocity command	Z000/12000
22	8	~ ~ ~	· · · · ·		8	velocity command	
27	<u> 2</u>		+	755	X		VELU I
2)		May walaattu	45000	277	35	circle end-point	32000
70		max. velocity	15000			monitor	<u>NW</u>
21			mm/min	556	and the second s	Ihreshold for com-	32000
52		Š.	(S) +)		2	pensation motion	jum 🔗
55			24.	357		Spindle drift	± 500
40	The.	Velocity command 🖄	2048/8192	1		N.	VELO
41		limitation	VELO 1	358		Dynamic smoothing	
42	<u>à</u>	<u>à</u>			d'	exponent for	5
43	2	Nº.			Nº.	thread 💦	
50		KV-factor	10000	359	<u> </u>	Maximum	9999
51			0.01	360	. C	speed	1/min
52		1.20	s1	361	8	for set	
3		() ()		362			. N. P.
50	S.	limit switch +	+ 00000009	767			555
50 51	<u></u>		- 77777777	705		-	1
2			hu	204		-	
67	and the second s	26.		205	200		
0)				300			
/0		Limit switch -	± 9999999 99	371	<u>S</u>	Manual feed	15000
[1]		. SP	hw	372	<u></u>	Man.rapid traverse	mm/min
/2		180		373	52	Ref.approach vel.	8
73			A.	374		INC speed	Star Star
30	155	Reference point 🔬	± 9999999 9	375		DRY feed	355
31			μm	376		Dwell time for	16000
32	6	8	6			spindle inhibit 👌	ms
33	2	Ko x		377	N.	Min.speed	8192
90.0		Backlash	± 255	1		spindle motor	VELO 2
1		compensation	um so	378		Cut-off spindle	1/min
2		100 M				speed for M19	17 III 10
33		Ś	Ser	370	<u></u>	Gain-factor	1/10:0000
<u>í</u>	12	Reference point	+ 0000	1 10		for M10	
1		chift	- 7777	700			uegrees
12		20111	۲ ۳ ,	200		rosition limit	1/11 degre
	<u></u>	2	+/ 🔊		<u>@</u>	TOP MIY	
2	29	······································	70000	581	N.º	Software edition	(32000)
US		Mult-gain	52000	383	Ser.	Increase update	30
3		30	Cx min/m		<u></u>	time 🔊	1/2 ms
Ζ		100		386	50	Acceleration	32000 📈
3		8	18	387		time constant	4 ms 🔊
0	534	Drift	± 500/2000	388		for	. Start
1		compensation	VELO 1	389		8 gear ranges	-1-
2	~	x	N	390			
3	38	28.5		301			
	<u> </u>			1702	<u>N</u>		
- 60			· / 2.77 /	174/ 1		P 5. 9 P	

specific machine data ..0 : X-axis ..1 : Y-axis ..2 : Z-axis ..3 : 4th. axis

www.chad

Machine Data Bits

.53	6 ×	Bit											
Nr.	7	6	5	4	3	2	1	0					
N 4005				8									
N 4015			13	5	0	0	0	0					
N 4025			35					, Ô					
N 4035		10					×	51					
N 4045	- 23	2					Ser.						
N 4055	S.					1.5	5						
N 4065						14							
N 4075					1								
N 4085		L		<u> </u>									
N 4095	1	0		2	1	0		0					
N 4105			So										
N 4115		2	С.					S.					
N 4125		S					3						
N 4135	18						õ.						
N 4145	12	0	0	0	0	0							
N 4155	1			0	13								
N 4165		0				1	1	1					
N 4175	0	Ó	0	0		0	0						
N 4185	0	0	0	0	0	0	0	0					
N 4195	0	0	0	0	0	0	0	0					

Do not change the given values.

,adva.pl

Pres Yes	ent No	Order code	Options	3T. T	/mo M	3T T	/M2 M	3T T	/M3 M	
S.		A04	4th. axis	-	X	-	X	<u>51-</u>	Х	406:7
		B02	Paper tape reader w/o reels	х	Х	X	×	. X	Х	. Bashe
	444	B03	Paper tape reader with reels	х	X	X	х	х	X	a an i an i an
à		B05	NC w/o operator panel	×	Х	х	х	X	X	
		B06	Operator panel switch-over	х	х	х	x	X	х	S. Card
		B41	Inch-metric	х	X	X	×	х	Х	408:5,6,7
	- 4 ³ -	B61	3-D interpolation	-	4	10	-	-	X	416:5
à		870	Drilling/Milling patterns	S.	-	-	-	-3	Х	4(6:1
1º		B72	Drill pattern,bolt hole circle	° -	-	-	2	200	Х	416:1
		B76	Read/write system memory,@ 29	-	-	-ă	6 ³⁵	х	Х	416:4
	44	B78	In process gauging	-	1	4	-	х	X	416:3
\$		C33	Chamfers and radii insertion	×	Х	х	х	X	X	415:0
ter.		C43	Memory extension to 16 k ch.	X	Х	Х	X	X	Х	Card
		C44	Memory extension to 32 k ch.	х	х	X	X	Х	х	Spart O
	44	E31	Threading G33	S	X	S	х	S	X	415:1
8		E42	Oriented spindle stop M19	8	-	-	-	X	X	407: 4,5,

1.3 Options available according to shipping notice

= not possible = possible = standard

Х

S

Pres yes	ent no	Order code	Options	3T T	/MO M	3T T	/M2 M	3T T	/M
		F05	S-analog	S	Х	S	×	S	
	1.	F71	External data input	-	12. 12.	X	Х	·X	
%		J11	Operator dialogue	X	Х	Х	х	x	
0		J12	Automatic tool offset calculation	X	-	, X	S. S. S.	x	
		J22	German text display	-	-	, S	-	х	
	44	J23	French text display	-	122	-	-	Х	
29		J24	Italian text display	ò-	-	-	-	X	
		J25	Spanish text display	-	-	-	0	Х	
	3	J84	Machine control panel	Х	X	X	X	х	
	424	K11	Integrated EXE-times 10X	Х	X	Х	X	Х	
2		K12	Integrated EXE-times 10Y/Z	ेx	Х	Х	x	X	
		K51	Integrated EXE-times 5X	X	Х	х	X	х	
	.5	K52	Integrated EXE-times 5Y/Z	х	Х	X	X	х	
	14	K53	Integrated EXE-times 5Z	-	X	-	x	-	
2		K54	Integrated EXE-times 5-4th.axis	<u>§</u> -	х	-	х	(Pro	
		N20	PC memory extension 8k EPROM 0.5k RAM for 130 WA	-	-	x	X	х	

Options available according to the shipping list

= not possible = possible = standard

- Х
- S

Pres yes	ent I no	Order code	Options	3T T	/MO M	3T T	/M2 1 M	3T. T	/M3 ¹ 1 M
		N22	PC memory extension 8k EPROM 4.5k RAM for 130 WA	-	-	x	x	x	x
		N23	PC memory extension 12k EPROM 4.5k RAM for 130 WA		-	x	x	X	x
2.2	N24		PC memory extension 16k EPROM 4.5k RAM for 130 WA	<u>à</u> -	-	х	x	×	x
		N25	PC memory extension 20k EPROM 4.5k RAM for 130 WA	•	-	x	x	x	x
	And C.	N32	PC memory extension 8k EPROM for 130 WB	-	1	x	x	х	X
,0; ,		N34	PC memory extension 16k EPROM for 130 WB	9	-	x	x	X	x
		N6D	Digital input 32I PC board 420 - 3	، ــــ	-	X	X	X	x
	34	N65	Digital output 32 O PC board 445 - 3	-	424	x	x	X	X
3.2		N70	Digital output 16 O PC board 444-3	2 -	-	x	X	x	x
	and s	N81	Digital Input/Output 48 I,24 O PC board 03400	-	-	x	x	х	x
à	4	N82	Digital output 16 O PC board 03460	à.	-	x	x	X	x

Options available according to the shipping list

= not possible = possible

Х

Pres yes	sent no	Order code	Options	3T/MO T M		3T/M2 T M		3T, T	/M3 M
		N83	Digital input 96 I PC board 03410	I	-	X	X	. X	X
	a chart	N84	Digital output 48 O PC board 03421	-	12	x	x	x	X
10.9		N85	Digital Input/Output 32 I,32 O PC board 03450	0.	-	x	x	X	X
	and it	N90	Digital input 16 I PC board 432 - 3	-	-	X	×	X	x

Options available according to the shipping list

= not possible = possible Х

1.4 Jumpers on the I/O Boards

(not for version O)

, L

<u>À</u>	A. A	Had I	10.91	Had
OPTION	PC board type	Location Nr. 3T/M2 3T/M3	Address Byte Nr.	Jumper
Anna.	and the ph	enne Anne	mashart	
ol something	Surger Surger	AIGDOULC ANNALS		N N N N N N N N N N N N N N N N N N N
	automative.	AND DALOT MANY	automatike	N5 K 0 17 11 0 9 O 9 0 0 0 0 0 0 0 O 0 0 0 0 0 0 0 0 1 2 3 4 5 6 7 6
ý	automatika.pl	and Dautomany And	RUSSINGHAR	NELD 1718 00000000 00000000 12345678
		Ardbaltonadient ann		

4. Non-standard Strappings

Complete only in case of deviations

	No.9	A CONTRACTOR OF	² <u>s</u> d	
Designation	Board	Standard Strappings	Special Strappings	Yes
20 mA - Interface	03100	NC active	NC passive	P2
Probe output	03315 03325 03350	Relay contact or open collector of 0 A B o o open C D	Other probe outputs	12. 12.
Velocity control ready	03320 03325 03350	E H external Signal is used o o open P N	Signal is not used oo P N	4
Command value output	03325 03350	Command value ground CVG connected to NC-M	other CV circuit	42.4
PC outputs are locked in case of NC fault	03800 A	PC outputs not locked oo A B	PC outputs locked oo C B	the second

Lists and Tables

Contents

2

2.1 Standard machine data bits (automatically set bits)

2.2 Machine data list (Overview lists and standard machine data)

2.3 Machine data bits (Overview list)

- 2.4 TEST display list (following error, actual value,K_v,etc)
- 2.5 Setting data for versions 0 and 2
- 2.6 Setting data for version 3
- 2.7 Alarm list
- 2.8 Interface signals 3T, 3M

Code tables for switches, gear ranges, and external signal input (see sec. 8)

2.1 Standard Machine Data Bits

Like the standard machine data (section 2.2), these bits can be set simultaneously (see sec. 4.4 for operation sequence).

SINUMERIK 3T

	Machine Data Bits											
Nr.	7	6	Š	4	3	2	[1	0				
N 4005	1	1	1	1	0	1	0	0				
N 4015	1	1	1	1	0	0	1	1				
N 4025	1	1	1	1	0	0	0	0				
N 4035	0	0	0	0	0	0	0	0				
N 404S	0	0	0	0	0	0	0	0				
N 4055	0	0	0	0	0	0	0	D				
N 406S	0	0	0	0	0	0	0	0				
N 4075	0	0	0	0	0	1	0	0				
N 4085	0	0	0	0	1	0	0					
N 4095	1	C	1	0	•	1	0	0				
N 4105	1	1	1	1	1	1	\mathbb{P}^{1}	1				
N 4115	1	1	0	0	0	0	1	0				
N 4125	1	1	0	0	0	0	ື	0				
N 4135	0	0	0	0	0	0	Ó	0				
N 4145	0	0	0	0	0	0	0	0				
N 4155	1	0	1	0	1	Ó	1	0				
N 4165	C	0	0	0	0	••	1	1 -				
N 4175	0	0	0	0	0	0	0	0				
N 4185	0	0	0	0	0	0	0	0				
N 4195	0	0	0	0	0	0	0	0				

SINUMERIK 3M

	I		B	its		200			
Nr.	7	6	5	4	3	2	1	0	
N 4005	1	1	1	1	1	0	1	1	
N 4015	1	1	1	1	0	0	1	1	
N 4025	1	1	1	1	0	1	1	0	
N 4035	0	0	0	0	0	0	0	0	
N 4045	0	0	0	0	0	0	0	0	
N 4055	0	0	0	0	0	0	0	0	
N 4065	0	0	0	0	0	0	0	0	
N 4075	0	0	0	0	0	0	0	0	
N 4085	0	0	0	0	1	0	0	1	
N 409S	1	0	0	0	•	0	0	0	
N 4105	1	1	1	1	1	1	Ī	1	
N 4115	1	1	0	Ó	0	0	1	0	
N 4125	1	1	0	0	0	0	1	0	
N 4135	0	0	0	0	0	0	0	0	
N 4145	0	0	0	0	0	0	0	0	
N 4155	1	0	0	0	1	0	0	0	
N 4165	0	0	0	0	0	••	1	1	
N 4175	0	0	0	0	0	0	0	0	
N 4185	0	Õ	0	0	0	0	Ō	0	
N 4195	0	0	0	0	0	0	0	0	

For version 0: simultaneously and time On Simultaneously and time On For version 2 and 3 : Simultaneously and time On Simultaneously and time On

- O is set automatically (w/o PC) in version O
- In versions 2 and 3,1^{*} is set automatically (with PC)
- * In versions 0 and 2,set to^{*}0^{*} In version 3,set to^{*}1^{*}

2.2 Machine Data List with Standard Machine Data

AXIS-SPECIFIC MACHINE DATA (TEST)

			<u> </u>	
Manual input (with automati- cally set standard values)	Explanations	Input unit	Max. input value	Units
standard Values) 10* S 11* S 200 12* S 13* S 13* S 13* S 14* S 15* S 1666 16* S 17* S 9999999 17* S 9999999 18* S 19* S 20* S 21* S 22* S	Position tolerance +) Clamp limit +) Acceleration Max. Velocity +) Velocity comm. value lim. Kv-factor Software limit switch + Software limit switch - Ref. point value Backlash compensation +) Tool ref. point value +) Ref. point shift +) Mult-gain	MS MS IS IS - MS MS MS IS MS S S	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	یس پیس O.O1 m/s ² mm/min VELO 1 O.O1 s ⁻¹ پیm پیm پیm ریm
23* SD	Drift compensation	-	± 500	VELO 1

 +) for limit values and units for degrees or inches, see the machine data description (section 7)

Axis assignment :

*	3T	3M		
0	X-axis	X-axis		
1	Z-axis	Y-axis		
2	-	Z-axis		
3	- 5	^{VV} 4th. axis		

- 1) for versions 0 and 2 : max. 9 999 999
- 2) for 12 BIT DAC : max. 2048 for 14 BIT DAC : max. 8192
- MS = units for the measuring system IS = units for the input system

COMMON MACHINE DATA (TEST)

Manual input (with automati- cally set stan- dard values)	Explanations	Input unit	Max. input value	Units
350 S500	Cut-off velocity +)	IS IS	15 000 15 000	mm/min
352 5 0	Tolerance for contour monito		32 000	mm•Test 850
353 5 500	Duell time for nos monitor	_		125 • 1000
	Velocity comm value limit		3000/ 12000 3)	
755 9 10	Circle end-point monitor	TIS	32 000	
356 S10	Threshold for compensation	+115	32 000	
357 50	Spindle drift	4 ²	± 500	VELO 2
358 S0	Dynamic smoothing exponent for thread (2 ^x -1), sample time	-	5	2 ²
359 S500	Nioni , stori	- 50	C C C C C C C C C C C C C C C C C C C	
360 S1000	and the second se	<u>s</u> oo	1000	1.500
361 S2000	Maximum speed for	4 ³⁶ -	0,000	1/-:-
362 S 4000	8 gear ranges	-	9 999	
363 S 4000	Mar Marin	_		9 [.] 2
364 S 4000	· Strats		Sar, "Public	. 5
365 5 4000	Source Mainte	. CORULE	A BOUL	and the second s
366 S 4000	and a second sec		. AND I STATE	and the second
371 S 2000	Manual feed	+) IS	15 000	mm/min
372 5	Manual rapid traverse	+) IS	15 000	mm/min
373 S10000	Ref. pt.approach velocity	+) IS	15 000	mm/min
374 S500	INCremental speed	+) IS	15 000	mm/min
375 S2000	DRY feed	+) IS	15 000	mm/min
376 S1000	Dwell time for spindle inhibit	+) -	16 000	ms
377 S0	Minimum spindle motor speed	+) -	8 192	VELO 2
ELO 1 = <u>10 V</u> 2048 f	or 12 BIT DAC	VELO 2 :	$=\frac{10 V}{2048}$ for vers.0-2,	rom software from software f
ELO 1 = <u>10 V</u> f	or 14 BIT DAC	VELO 2 :	= <u>10 V</u> for vers.0-2, 8192 vers. 3,f	fr.software O6 r. software O2
) Limit values) for 12 BIT DA	and units for degrees and inc C,max. 3000	hes,see (machine data descrip	tion (sec.7)

		1947 - Cal.				
-	Manual input (with automati- cally set standard values	Explanations	Input units	Max. input value	Units	
	378 50	Cut-off spindle speed for	- vauto	9 999	min ⁻¹	
	379 S0	M19 3) Mult-gain factor for M19 3)	- I C	10 000	min ⁻¹ /360 •	
	380 S0	Position limit for M19 3)	32.00	1 000	1/11 degree	
	381 SD	Software edition	-	(32 000)	3	
	383 S0	Increase update time 4)	-	30	1/2 ms	
	385 S99999999	2nd.software lim. switch X ~ x),+)	MS	± 99 999 999 ¹⁾	Jum 🧹	9
	386 S0	Acceleration time const.for 1st.gear 3)	ararah	32 000	4 ms	
	387 S0	Acceleration time constant for 2nd. gear 3)	-	32 000	4 ms	/
	300 30	for 3rd. gear 3)		32 000	4 ms	
	389 50	Acceleration time constant for 4th. gear 3)	- John	32 000	4 ms	
	390 S0	Acceleration time constant for 5th. gear 3)	And and a second second	32 000	4 ms	
	391 SD	Acceleration time constant for 6th. gear 3)	-	32 000	4 ms	
	392 S0	Acceleration time constant for 7th. gear 3)	6	32 000	4 ms	
	393 S0	Acceleration time constant for 8th. gear	- Ardbaure	32 000	4 ms	
	1.	·	- CA	h		

x) only for the 3T

- +) for limit values and units for degrees or inches, see machine data description (sec. 7)
- 1) max. = 9 999 999 for versions 0 and 2
- 3) only for version 3, from software edition 02 up
- 4) for vers. O and 2, from software O6 up
 - vers. 3, from software edition O2 up

2.3 Machine Data Bits (TEST)

Nr.		- Charles	6	5 Starke	•	lit 3	²	1,00	0
400S	ČŎ	Strobe	signal dur	ation	1 0	Address (fer & too	of (inserte l tip (3T)-	d) radius & resp.mill (cham-1 3M)-radius
401S		Delay	time for st	l robe signal		Ad.for co	ding the too	d position	for 31 only
402S		Durati	on for MO2/	1 M3O signal	0	Address of	f 4th.axis,	only for 3M	0
4035	1. Axis	Omit 5 axis in the start interlock	Partial 6 actual value multiplied by 10	Rotary axis	Partial actual valu divided by 2	Partial actual value multiplied by 2	Sign change for partial actual value	Sign change for velocity	Ref. point approach in (-)
4045	2. Axis	" _" 5) 6)	11 17	17 11	11 11	11 11	" "	" "
4055	3. Axis	" " " ")	н п	17 11		11 11	п н	Г . н н
406S	4. Axis	•)	, ⁶)	11 H	нн	и и		Сп. н.	11 11
407S	14	NC-start w/o ref.point approach	•)	•)	3)	Spindle speed in 0.1 rpm 3)	Spindle encoder installed	Sign change for spindle actual value	Spindle actual value
4085		sudden stop on the limit switch	Input reset state for inch (G70)	O	Position control in inch	spindle control directly by NC	Aux.function autput prior to motion	Auxiliary fu output durin number search	nction () g sequence h (SNS)
4095	ALIO	NC-PC interface activation	•)	Feed rate referred to putter center	Hend uheel installed	PC installed	Dianeter ^(C90) programing X-axis (31)	2 nd •) measuring board	Length com- pensation adjusted on non-moving
4105	active at	DATA Start at MDA	Zero offset data	Tool offset	T.O. deta additive input	Part program edit	Dry run feed <u>rate</u>	SNS	axis - 1 Superimpose of S.M.T
4115	RSZ52 Input	Device co	l ode (i	input devic	e)	5	Baud rate	(input	device)
412S	RS232 Dutput	Device co	de (c	butput devi	ce)	5	Baud rate	(output	(device)
4135	@		EIA	code for	0	 -	2	ŧ	2 2
* 414S		DC control Character W/Rs232_4/		. Crashe		onabl		Name of associated the 4th. a	axis) With Wiss
4155	ALC:	CRC (3T) •)	•)	Analogue spindle speed (gnly 3 T)	•)	Teach-in playback MDA	•)	Thread and feed/rev. (only 3T)	4
4165		Block end with car- riage return and LF	ACTUAL VALLES X-exis displa in diame-(31)	•)	-)	•) 3)	NC Alarm Texts	Cycles and a	RS232 (V24.2)

active only after Power on-Reset

*) Single bits according to Start-up instructions or Control Data table.
	_	BIT										
Nr.		7	6	5	2 4	3	<u></u> 2	1	0			
4175		automativ		autorna	Decelera- tion to ve locity of next block 4)	pindle speed override active in threading()	Weer input in clianeter (31) 4)	in the second	14-BIT DAC 4			
4 I _. 85	19. 19. 19. 19.	5	-search -	97	-34 ²⁴	1.0°	A	AN. OT				
4195		H2	Q	-	Naid		23.9		120.01			

- 3) Only version 3, from software edition 02 up.
- 4) Versions 0 and 2, from software edition 06 up.
- Version 3, starting with software edition 02. 5) Versions 0 and 2, from software edition 07.
- Version 3, starting with software edition 03.
- 6) Only version 3, from software edition 03 up.

1 Address for radius, chamfer, and tool radius, as well as for tool mose

1		B	Lt		Name	
	3	2	1	0		
	0	0	1	1	À	
	0	1	0	0	В	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~
	0	1	0	1	· C	102
	0	1	1	0	U.	25
	0	1	1	1	v	C
	1	0	0	0	W	
	1	0	1	1	P	

3 Name of the axis to which the 4th. axis is subordinated (only for 3M)

B	it 👘	Name
1	0	
0	0	X
0	1	Y
1	0	Z

Times

2

i [B	ĺt	25	Time	[ms]
7	6	5	4	3т	3M
0	0	0	0	16	18
0	0	0	1	32	36
0	0	1	0	48	54
0	0	1	1	64	72
0	1	0	0.	80	90
0	1	0	1	96	108
0	1	1	0	112	126
0	1	1	1	128	144
1	0	0	0.	144	162 [·]
19	0	0	1	160	180
1	0	-1	0	176	198
1	0	1	1	192	216
1	1	0	0	208	234
1	1	0	1	224	252
1	1	1	0	240	270
1	1	1	P	256	288

Device coding

5

Meaning of bit

B	it	Number of
7	6	Stop-Bits
0	1.2	1 Stop-Bit
1	0	1 1/2 Stop-Bit
1	-1	2 Stop-Bit

Bit 5	Parity Type
0	odd
1.	even

Bit 4	Parity Bit
0	w/o parity
1	with parity

Bit 3	Ready for
	operation (USR)
~	evaluation
0.8	no
18	yes ·

7 Auxiliary function

	out	put during SNS.
1 B	it0	Output
0 0	0 1	none after cycle start
1	0 1	During SNS -

Baud rate

6

Bit	Baud
2 1 0	Sec
000	110
0'0 1	150
0 1 0	300
0 1 1	600
100	1200
101	2400
1 1 0	4800
1 1 1	9600

	M	lach	ine	da	tur	i (b	ina	iry)	HEX	Device
. Nº	B7	B6	BS	B4	B3	B2	BI	B0	34 ^{0,0}	<u></u>
Lorno -	1	1	0	0	0	1	0	0	C4	FACIT 4040 with P 81 (1200 BAUD)
	1	1	0	0	0	0	1,0	1	С3	FACIT 4070 with M 77 interface (600 BAUD) 54
	1		10	0	0	0		1 	C2	PT80 Siemens printer Start-up datum with STT104 Interface (300 BAUD)
and the state	1		0	0	Ó	1	0	10	C4	SANYO M25020 cassette unit with
C ^r	1	1	.0 .	10	0	1	0	0	C4	SME (1200 BAUD)
	Ī		0	10	0	1	0	10	C4 .	Coupling NC NC
	1	l I i	10	0	0		0	' 0 	C4	FACIT 4030 (1200 BAUD)
No.		1			22	[1	1	Carl	
680		[1	E.E.	1	1 1	1			A CONTRACTOR OF
2		ſ	1.05	р 1		1		1050	I	walter water
	0	0	0	0	0	0	0	0	00	Output : PT80 (300 BAUD) 4) Input : S-tape reader
	0	0	0	10	0	1	1		07	Siemens tape reader with and without reel (9600 BAUD)
Casha	0	0	0	i 0 1 1	lo <u>r</u> 1	1 1 1 1	1 I 1 1		OF	Siemens tape reader with and without reel (9600 BAUD)
<u> </u>	0	10	0		11.		0	10	18	Teletype ASR-33 full duplex 6) (110 BAUD) 10
	0	0 	1 	0	0		<u>р1</u> 1.	0	26	FANUC hand reader 4) DC1/DC3 controlled (4800 BAUD)
	0	0	1		10			10	36	FANUC system P/D 4) (4800 BAUD)
5	0		1 I Sall	10	0	1	+ 0 .8		24	Coupling NC NC 4) with control characters DC1-DC4 (1200 BAUD)
	0	0	1	0	0 				27	FACIT 4040 with P 81 4) with control characters DC1-DC4 (9600 BAUD)

5 and 6 : Possible Input/Output Device Connections (Selection Table)

4) For versions : - O and 2 from software edition O6 up
- 3 from software edition O2 up

6) For versions : 0 and 2 (from software 04 and 05 up), and 3 (from software edition 01), the following machine data must be entered for teletype ASR-33 : 1100 0000.

2.4 <u>Display (Test)</u>

ID Nr.	A 3T]	xis 3M	Display	Un: Metric	its Inches
800 S	X	X	Following error	μm	10 ⁻⁴
801. S	Z	Y	H 300	, d'at	In.
802 S	L.	Z	H		354
803 S	_	4th	11		
810 S	X	X	Actual value (position control)	μm	10-4
811 5	z	Ŷ	H MARY MARY	. Star	In.
812 S		Z	11	100100	
813 S	3 ² -	4th	H AND AND	and C	3
820 S	X	X	Velocity command value	VEL	_0 1
821 S	Z	Y	n	2048/819	92 VELO =
822 S	_	Z	11	= 10	v
823 S		4th	H Stor	10	
830 S	X	X	Partial actual value	л	10-4 _{In}
831 S	z	Y	n san	update t.	update t.
832 S	-	Z	11	Update t 3T 8	lme : 3 ms
833 S	[4th	H HON HON	30) ms
840 S	X	X	Contour deviation	Ļm	10-4
841 S	z	Y	H Kan		In.
842 S	and -	z	H	And .	
843 S	-	4th	TI		
850 S	X	X	Calculated position loop gain	0.001	0.001
851 S	z	Y	11	m/min	In/min
852 S	<u> </u>	z	11	mm	1000 In
853 S	<u></u>	4th	H ALO	ALCON.	
860 S	C ⁴	-	Spindle speed command value	VEL	.0 2
861 S	-	- 3	Spindle position	<u>360</u>)° 96

VELO 1 = $\frac{10 \text{ V}}{2048}$ for 12 bit DAC VELO 1 = $\frac{10 \text{ V}}{8192}$ for 14 bit DAC VELO 2 = $\frac{10 \text{ V}}{8192}$ for versions : $\binom{0}{3}$, 2 from software 06 up VELO 2 = $\frac{10 \text{ V}}{8192}$ for versions : $\binom{0}{3}$, 2 for software 04,05 VELO 2 = $\frac{10 \text{ V}}{2148}$ for versions : $\binom{0}{3}$, 2 for software 04,05 3, for software 01

2.5 <u>Setting Data : FOR VERSIONS 0 and 2</u>

2.5.1 Input and Display : Zero Offset, S-max.,

M19 Setting Data and R-parameters

<u></u>	0.0					1
Input Nr.	Address	Display/Input	conable	Sign	Nr. of decades	Units Metr, In
$1-4^{2}$	X,Y,Z,4th	Zero offset	X	±	7	µm 10 ⁻⁴
5	X,Y,Z,4th	Programable zero offset G59	-	±	7	µ ^m ¹⁰ ^{−4}
6	X,Y,Z,4th	External zero offset	- No	±	4	m 10 ⁻⁴
ī. 2 0	Succession	Spindle speed limitation G92	X		4	1/min
25	N	Setting data bits	X	}	see below	
100 - 149	R	R-parameters	X	±	7	è

2) Starting with software edition D6 , 4 ZO for 3T

2.5.2 Setting Data Bits (Operator Data)

also see the operating instructions, section 12.4

Identity	Nr. d		6		<u> </u>	
Input Bit	0		Closello.	1	Stranko.	
	Hand wheel ,increments pe	er divis	ion			
est the	1	10]		100	and the second
7	<u> </u>	C)	1	1	2
6		1	Ś		0 8	<u> </u>
5	Tape punch ISO-code	• 	Tape p	unch EIA	-code	
4		. S			NITO'	
3	Program start with %	1)	Progra	m start w	ith LF	<u>1) ()</u>
2	Tape block parity OFF	42	Tape b.	lock par	ity ON	252
1	Operator dialogue OFF		Operat	or dialo	gue ON	
D	Actual value display in reference to machine zero)	Actual refere	value d nce to p	isplay in art's zero	⁾ *)

*) Referred to W, without zero offset, without tool offset

1) Starting with software edition D6

2.6. Setting Data FOR VERSION 3

2.6.1 Display and Input Zero Offset, S-max.

M19 Setting Data Bits and R-parameters

Input Nr.	Address	Display/Input		Sign	Decade Nr.	U metr.	nits inch
1-4 2)	X,Y,Z,4th	Zero offset	х	±	8	۲m	10 <mark>-4</mark> 10 _{1n}
5	X,Y,Z,4th	Programable zero offset G59		±	8	۳u,	10 <mark>-</mark> 4 In
6	X,Y,Z,4th	External zero offset		±	4	J um	10 <mark>-4</mark> In
20	S	Spindle speed limitation G92	x	1	4	1/mi	п
22	S	Oriented spindle stop 1) M19	x	+	4	0.5	degree
25	N	Setting data bits see 2.6.2	x		8	¥	
26	N. COO	Setting data bits see 2.6.3	х		N. COMB		Sal C
100 - 199	R	R-parameters	x	t ±	8	19 ¹ 2	4

2) Starting with software edition 02, 4 ZO for 3T

2.6.2 Setting Data Bits Nr. 25 (Operator Data)

Input Bit	atent O test	He? 1 He?
7		
6	AND AND	and the second sec
5	Tape punch ISO-code	Tape punch EIA-code
4		
3	Program start with % 1)	Program start with LF 1)
2	Tape block parity OFF	Tape block parity ON
1	Operator dialogue OFF	Operator dialogue ON
D	Actual value display in reference to machine zero	Actual value display in *) reference to work piece zero

1) Starting with software edition O2

*) Referred to W, without zero offset, without tool offset

Input	C C	Carlon .	anasha 1	SCH 2.
<u>Bit</u> 7	Allen Allen		Real Providence	8
6		Sec. Sec.	and the second se	S. S. S.
5		<u></u>		
4	A CONTRACTOR OF	NO.S	12. ⁹¹	10. ⁰¹
3	2. Page 1	Carlo Carlo	and the second sec	87
2	Hand wheel increments per division	0	Hand wheel increments per division	100
1 4	Hand wheel increments per division	0	Hand wheel increments per division	10
16.90	Hand wheel increments per division	D	Hand wheel increments per division	Star 1

2.6.3 Setting Data Bits Nr. 26 (Operator Data)

*) can be cleared only with PORESET

Clear	Nr.: J. Decade 2. Decade	.8 81t 7	7 Bit 6	6 Bit 5	5 811 4	4 81L 3	3	2 Bit 1	1 Bit 0
	00		, Bar		Axis	80		Axis	Axis
	01		A. A.	ļ	Axis		3 MA	O Axis	⊕ — fs ^{Axis}
	02	. All		2.2	Axis		1	Axis	Axis
	63	S.S.		(Card)	Axis	afraid.		Axis	Axis
	10.000	Axis 1	. Ser		Axis	Axis	Axis	m Axis	Axis
	11	Axis	44	·	Axis	Axis	Axis	Axis	Axis
	12	Axis		્વે	z Axis	Axis	Axis	BB Axis	Axis
	13	5 Axis		Card Card	Axis	Axis	Axis	Axis	Axis
٣	22	Spindle encoder fault	NIGON		j.	Position con trol icop monitor of spindle	Emergency Stop	Control not ready	
	23	Time monitor V24 Interface	Overflow 2 Hardware Reader error	No stop-bit	Overflow	Parity	Control whito trans-	Overflow 1	Stop-bit - parity error
	24	2010-P		and the R		all a	, needuet.	Over- temperature	5 ²
	25,010		Block w/o LF or with more then 120 charct.	<u> </u>		Operating error V24 interface	Parity error in memory	Program not in memory	Block not in memory
	26		ACREAN!		AL CONTRACT		44	1)	SNS block not found
	27	Memory overflow	Discrepancy between menory and tape progr.	Tape format error	Tape input not alloued	Block with move then 120 cheract	Block parity error	Irrelevant EIA character	Character parity error
	28	Sub- routine error	Cutter point error	STR		automo		- alloffic	General decoding error
	29	· .	False input perameter	False block order	False GO2/GO3	False radius value	False angle value	No intersection	False input value

2.7 Alarm List

Clear	Ar.: J.rd. Dec.	8 811 7	7 815 6 - Ó	6 811.5	5 811 4		3 811 7	2	1 Bit D
	30	Circle end-point error	ANNI-GOOD		ANI CO	Zero or tool off set value error		Option not present	Circle not in selec- ted plane
4	31		Too many axes to be driven	No F-word or too large	4	Falsely programed lead	4		4
$\overline{\mathbb{N}}$	32	dre n		addro."		officially con		Functions not allowed with selec ted CRC	
	33				MIGDO			Sault	4
	34		10. 		4		44	~	14
·	35	Jean P	•	adka.Q		-atyles.P		all a	NC-start without ref.point
	50	2x axis,or more then 2 axes progr. rep.progr.f.	doauto	CRC/ contour error	Blueprint programing error	False block structure		More than 6 geometry parameters	General
	51		and and a second second	Complete block can't be displayed	Preselected block nr. cernot be found	Block with more then 120 characters	Memory overflow	Input inhibited	Input only in Reset- state
رسّا	52	Ku-factor not deter- mined	Ky-factor of axes is not equal Monitor —•	ster?	-	ater?	Hold at ¹⁾ thread	Nº.	Strobe- input error
	53	Ceneral input error	Last proor. not termine- ted	Driy 2 axes alloued at playback	Playback only allower at axes rest	Playback MDI inter- TOT	Playback in 1st. block not allowed	Program nr. already used	Block with nore than 120 charac- ters
Restart	70		and Contraction .		stated.		Sec. M.C.	False address code in mechine detum	44
	71	(a)		202					Battery alarm

- ** Input line (lowest line) must be completely cleared. Error 70X cannot be cleared with either RESET or CLEAR ! Error 71X can be cleared with CLEAR !
- 1) Alarm "Hold at Thread" Versions O and 2 up to software O5 Version 3,software O1

	N	C		NC -Interface Control or PC						PC				
	Test Nr.	Byte	7	6	بة 5	ta bit	3	2	et est	0	Dat , DW	ta Bl	lock #9 Relat. Byte-Ad	Flag Byte
	7	0	Operati D	ng Mode S C	elector B	Switch	Feed/Rap D	C C	lde Switc B	h 	350	н I 1 н	0	-0
1. I/O board for 3T,version 0,and interface PC-NC for 3T 2 and 3T 3.	7	11	Key àwitch	Dry Eun	Block delete	Single block	Sequence number search (SNS)	Spindle .C	Override B	Switch A	1		1	1
	7	2 .	Rapid traverse override ective	Rapid traverse superim- pose	Di X+:	rectional X-	Keys Z+	Z -5	*Decale- ration X	Control enable X	2	H H I	2	2
	7	3		2rd.soft- ere limit adton-X active	Data start 1)	Gauging 1)	Hand wheel X	NC start	*Decele- ration Z	Control enable Z	2		3	10 ¹ 3
	8	4	Gear C	Range C B	ode A	Spindle direction CW	Spindle enable	Feed enable		÷	3	H I	Å.	4
	8	 ⁵ 8	:> *E-stop	Read-ix enable	Mirror image Z-velues	Mirror image X-values	Axis lock	Without operator panel) 3		5	5
Dnly 3T 2	8	6	Strobe signal		Extern	al Data C E	Input ode Sign D	al for St	robe B	***	4	н I I	6	6
Interface PC - NC	8	7 	H	G	Extern F	al Data E	Input Datur D	c	B	λ	4	L L	7	8
N. LOS	9	8	Q	P	Extern O	al Data N	Input Datur M	L.	ĸ	I. A.C.	5	H I	8	702

1) Only 3T 3, from software edition 02 up

2.8 Interface Signals

Input Signals 3T

2-15

-3K2.9	N	с	NC-to	NC-to-Interface Control or PC							PC		
soutoman	Test Nr.	Byte	7	6	dipaliforna	Data B 4	it 3	2	1	0	Data DW	Block _# 9 ^I Relat. Byte-Ad	Flag Byte
1 AND A	10	0	Strot M	be Sign S S2 *)	al T	Spindle in ¶ position S1 *)		J.	Motion (Command X	.7н	1 14	12
) 1. I/C board for 3T,version D.and interfac	10		Program active	RS232 active 1)	NC Ready 2	NC Ready 1	Rapid traverse	Thread	M02/M30 Reset	Program stop MOO	7 L	15 	13
- PC for 2 and 3T 3	10	2 2	D	c	р1. В	BCD 0 A	utput D	c	I0 ⁰ В 4	A	8 H	16 	15
1 star	10	3	aller			340.9		and the	9		8 L		16
Only 3T 2 3T 3 nterface NC - PC	11	ALCONTO		, N	doautor.		. S	,5 ^{.01}		MIGDAULO	9 н	 18	17
\downarrow \diamond	11	<u>5</u>	D	10 C	3 B	BCD 0	utput D	1 ∵≁C	02 В	·. A	9 L	 19	14

1) Only 3T 3,from software O2 up

*) In version 0,the 4-decade S-function is output in two steps : S1 \triangleq 10³ and 10² decade (high byte),then S2 \triangleq 10¹ and 10⁰ decade (low byte) Output Signals 3T

2-16

	·															
	N	SY .	NC-	to-Inte	rface C	ontrol	or PC		.A		PC					
	Test Nr.	Byte	6	4	Da	ta Bit		1	2		Dat	a E	lock #9 ¹ Relat.	Flag		
No.			7	6	5	27.4	3	2.0	1	0	DI	4	Byte-Ad	Byte		
autor	7	0	Operati D	ing Mode C	Selector	Switch	Feed/R	apid Over	ride Swit B	ch o	1	н	1 0	0		
I/O board for 3M O,and interface PC - NC for 3M 2 & 3M 3	7	1	Key switch	Dry run	Block delete	Single block	Sequence number search (SNS)	Spindle C	Override B	Switch A	. 1	L		1		
	7	2	Rapid traverse override active	Rapid traverse super- impose	Directio	on Keys ·	Axis Sel switch B	code A	*)Dece- Leration X	Control enable X	2	н	2	2		
	7	3	4th. axis is main axis	. Second	Data start 1)	Gauging	ARAL GOO	NC start	*)Dece- leration Y	Control enable Y	2	L	3 3	3		
, ed	8	4	Gear C	Range C B	ode A	Spindle direction CW	Spindle enable	Feed enable	*)Dece- Leration Z	Control enable Z	3	н	4	4		
	8	5	*) E-stop	Read enable	Mirror Y	Image X	Axis lock	Without operator panel	*)Dece- leration 4.	Control enable 4.	3	L	 5	5		
Only 3M 2 3M 3 interface PC - NC	8	6 6	Strobe signal	Anna Mil		Extern Code E	nal Data signal D	for strop	B B	A	4	н	6	6		
	8	7	н	G	F	Extern E	Datum Datum D	Input C	 B [A	4	L	1 1 7 1	8		
	9	8	Q	P (0	Extern	Datum M	Input L.	<u> </u>	(I ^{QUIO}	5	н	8	2011 0 M		

1) only 3M 3, starting with software edition 02

(

2-17

Input Signals

ЗM

	N	с	NC-t	NC-to-Interface Control or PC								PC			
	Test		S. Contraction			Data Bi	t	all'an	1		Data				
	Nr.	Byte	7	6	5	4	3 0	2	1	0	DW	Relat. Byte-Ad	Flag Byte		
<u>^</u>	29		Strobe Sional		Spindle	Motion	Comman	Command			- 31				
1. I/O board for 3M O,and interface NC - PC for 3M 2 & 3M 3	10	0	M 52 *)	S	Т	S1 *)	4.	2	Y	X	.7H	<u> </u> 14	12		
	10		Program active	RS232 active 1)	NC ready 2	NC ready 1	Rapid traverse	Thread	M02/M30 Reset	Program Stop MOO	7 L	15 	13		
	10	2	D		01 ₿	BCD A	Output D	с	10 ⁰ В	A NOR	8 H	16	15		
1 and	10	3	star			Stra.Q		3	<u>,</u> ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,		8 L	17	16		
Only 3M 2 3M 3 interface NC - PC	11		50		(Sparton		Š	autonic		.55 ⁰⁰⁵	9 н	 18 	17		
	3ª			10)3	BCD	Output	1	02	e ^{da}					
	11	5	D	C	B	A	D.	C	В	A	9 L	1 19	14		

Output Signals 3M

2-18

1) only 3M 3, from software O2 up *) for version O (see page 2-16)

3. <u>Prerequisites and Visual Checks</u>

Contents

- 3.1 Preliminary remarks
- 3.2 Start-up prerequisites
- 3.3 Supply voltages
- 3.3.1 Power supply components
- 3.3.2 Line condition
- 3.3.3 Power supply logic components
- 3.3.4 Line connections for fans
- 3.3.5 Operator panel power supply
- 3.3.6 Machine data table
- 3.4 Visual checks
- 3.5 > PC boards and assembly
 - 3.6 Software system
 - 3.7 Information concerning version 1
 - 3.8 Information concerning versions 0 and 2 with software edition 04

3.1 A Preliminary Remarks

Plastic carpeting as well as the plastic or rubber soles of people's shoes can cause static charge accumulations of up to several kV.Integrated circuits are sensitive to such charges.For this reason,the circuit components and traces should never be touched before having discharged on a grounded part of the machine.

PC boards and power supply lines should never be connected or disconnected while the power is on.

Even when the control is switched off, one must be careful that no short circuits occur in the V_{cc} RAM traces , because these can lead to falsification of information in the buffered CMOS RAM memories, or even cause the traces to burn out.

3-1

WARNING ! Mind the safety instructions !

MOS WARNING !

Mind the safety instructions !

MOS is the technology used to manufacture highly integrated digital circuits. "MOS" is an abbreviation for Metal Oxide Silicon. The main advantages of the MOS method are :

- the simple construction of the transistor
- the high density
- the extremely low power requirements.



MOS

Logo on packaging Logo on the PC board

Μ

0

S

WARNING !

The PC board is assembled with MOS components.Potential equalization is necessary before the PC board is handled, in order to avoid destroying the MOS components. Take the PC board with its conducting foam out of the packaging box, and with your hand, touch a grounded part of the machine.Do not touch components or traces !

(Instruction included in the packaging)

Further Note :

Do not open the special packaging unnecessarily. Store only in the black (conducting) foam. Do not bring into contact with plastic materials (because of possible static charge build-up). Switch off the power supply before insertion or removal.

Prerequisites, which must be met by the customer before Start-up : The recommended machine data for the particular machine should be on hand.

Electrical and mechanical assembly of the machine must be completed, and the machine must be ready for operation (which should be confirmed by the customer). Also pertinent here is the note in section 3.3.13. Are the drives set up ?

The <u>interface and customer PC program</u> should be functional, tested according to the interface description, and connected to the machine and the NC (this should be confirmed by the customer). Also see the notes in section 3.3.13.

The position coders must be mounted and wired to the NC (visual check).

The <u>cables</u> to the interface and machine should be connected. The cable shields should be brought to the end point of the control, all according to the interface description.

The flexible grounding cables should be connected (visual inspection) :

Ground rail at the interface - SINUMERIK 10 mm² Ground rail at the interface - Machine base 10 mm² SINUMERIK (NC) - Operator panel 6 mm²

Check tapes should be available for testing the machine specific functions.

The customer must make available the personnel needed for <u>assistance</u> to work on the interface, machine operation, and the customer's PC program.

Recommendation : <u>traveling ranges</u> should be limited by moving the hardware limit switches (for larger safety distances).

If the customer has used <u>intermediate connectors</u> in the position control cables, check if the connections are properly made, check for strain relief, and especially for the required shields.

3.3 Supply Voltages

3.3.1 <u>Power Supply Components</u> (if applicable) Primary 3x 380/415 V / 50 / 60 Hz (tolerances + 10% - 15%) Secondary + 24 V

Type 6 EV 1350-5AK <u>20 A</u>

6 EV 1360-5AK 40 A

3.3.2 Line Conditions

The supply voltage for the logic components, including PC and operator panel, is designed for 24 V (DC).

This 24 V supply voltage must be generated from the line voltage by the components of the power supply.

Built-in power supply data :

-Stradyla	3T/3M Basic version	3T/3M Basic version	3T/3M Basic version
MARING COLUC	0,1	2	3
Rated line voltage	24 V_	24 V_	24 V_
Input voltage range including ripple	20 to 30 V	20 to 30 V	20 to 30 V
Input capacity	8100 µF	F بر 8100	⁸¹⁰⁰ µF
Current consumption 1) Logic components U _E = 20 V U _E = 24 V U _E = 30 V	IE = 3.9 A IE = 3.2 A IE = 2.6 A	IE = 5.0 A IE = 4.2 A IE = 3.4 A	$I_E = 5.4 A$ $I_E = 4.5 A$ $I_E = 3.6 A$
Current consumption 2) Operator panel U _E = 20 V U _E = 24 V U _E = 30 V	$I_{E} = 1 A$ $I_{E} = 0.8 A$ $I_{E} = 0.7 A$	IE = 1 A IE = 0.8 A IE = 0.7 A	IE = 1 A IE = 1.8 A I _E = 1.5 A

1) Current consumption of the logic components NC + PC without INPUT-DUTPUT of the I/O boards

2) without machine control panel

3.3.3 Power Supply Connection - Logic Components

This connection is made to the terminal strip on the front plate of the 24 V power supply 03 500 (NC) = wire gauge 1.5 mm² 24 V power supply to the extension rack = wire gauge 1.5 mm²

If multi-conductor cables are used, don't leave free any unused conductor, i.e. redundant conductors must be paralleled.

Wire gauge of the cable for Power On : 2 x 1 mm² shielded



The M-input terminal D1 of the NC connected internally to the chassis (grounded on back plane)

Note :

If external switch components are used for Power On, <u>no</u> latching switches may be used.Connections E - F (remove jumper) can be used as external enable (floating,e.g. relay contact).

3.3.4 Line Connection for Fans

Input voltage : 220 V, 50 Hz Terminal strip X1 Ð > ² + 22 U + V (or N) + not used 1 + X1 2 not used

Note :

There may be, in the fan section, a terminal strip X2, but it is not connected. Possible mislables (e.g. 24 V) must be removed.

3.3.5 Operator Panel Power Supply



Operator Panel Connection Board



<u>Warning !</u> Before switching the operator panel on, one must check that the power supply is hooked up with the proper polarity, and that there is a proper M-connection to the logic components. Also see the interface description, section 1.1.7. Improper connection may lead to the destruction of components (IC's) of the operator panel logic, and of logic components. Also check the 6 mm²ground connection to the logic frame.

			1.0
automaty	antomaty.	Basic version 0,2	Basic version 3
Operator	PB board type	03700	03780
6	24 V connector	X 7 00	X785
Panel	50 pole connector	X702	X781
Logic	PC board type	03100	03810
Components	50 pole connector	X10 2	X812

Operator panel to logic components connection :

3.3.6 Machine Control Panel 03 630

Faston connector 6.3 P24 Input voltage + 24 V_

3.4 Visual Inspection

3.4.1 Grounding

Clean grounding, for the dissemination of external noise, is essential for smooth operation. Special care should be taken that the required wire gauge be used, and that no ground loops are present (also see section 3.2).

3.4.2 Position Coders

Special attention should be paid that the scales (air gaps,etc) and pulse coder (coupling)are properly installed;also see the Heidenhain installation and adjustment instructions.Make sure the connectors are wired correctly and the connections are tight.Different brands of position coders can cause accuracy and surface quality problems, for which we take no responsability.

3.4.3 Cabling

The power and control cables should be separated.No ground loops are allowed.Poor grounding or ground loops become most apparent as low frequency noise on the velocity command value.This makes smooth runs impossible at low speeds.

Also check for any kinks, proper ducting, and cable tracks.

3.4.4 Shielding

The outer shields of all cables leading to or from the control must be grounded through the connectors <u>at</u> the control (see the interface description).Only the cable to the operator panel has a shield grounded on both ends.

3.4.5 Operator Panel

Check the switches, push-buttons, lamps, symbols, actual value and data displays.

3.4.6 General State

Are the PC boards fastened? Cover plates? Documents : log book and complete assembly parts list ? (The assembly parts list is included with the original shipping notice, and must be filed in the log book)

When components are exchanged or in malfunction cases, always check all the socket plugged IC's for proper location and connection.

WARNING !

The 24 V power supply 03500 and RAM memory 03210 should be disconnected only in cases of malfunctions, because machine data etc, are lost otherwise (battery in power supply).

3.4.7 Battery in Power Supply

The back-up battery for the NC and PC is within the power supply unit 03500; it can be exchanged from the front. The positive terminal is at the top, at the insulated contact ; ground shorts must be prevented ! The battery should be changed only while the control is on, so that memory information is not lost. The battery voltage is always checked at PORESET, and if it falls below 2.7 V, alarm 711 lights up.

Battery type : 3.4 V / 5 Ah TL 2200

IEC-R-14 (Baby battery)

Connection for the auxiliary battery : (only applicable to basic version 2 (with PC)

An external auxiliary battery can be connected to the 6.3 mm faston terminals (marked U-BATT and O V) for test purposes. This battery would be located on the backplane of the PC. The terminals can be accessed by removing the right-side (fourth) I/O board (see section 3.4.2).

WARNING !

For units with integrated PC, if the back-up battery voltage is too low, when the PC is switched on, it goes into Stop-mode. This also prevents the NC from starting, and the red light on the NC-CPU 03100 lights up or flashes (see section 4). In this case no alarm 711 display possible.

3.4.8 Cables

Check all the cables (according to the cable and devices overview in the interface description), and especially those supplied by the customer. At least one connector should be opened and examined closely, with particular attention being paid to conducting elastomere connections. If you find deviations from our guidelines, please inform the sales office concerned, and if necessary, correct the problem (see Interface section 1.1.5).

3.5 NC-Boards and Strappings

3.5.1 Identification System and Generalities

		0 3 3	
System Nr.	autorn	J.on T	TT TOT
Board type a)	AN ISON	ALCO LAND	ALS .
1 CPU	4	4	4 ²⁴
2 Memory			
3 Position control			Ľ
4 I/O boards			100
5 Power Supply	Store.		
6 Wired boards			A CAL
7 Operator panel board			14°
8 Coupling board			
9 -			
0 other <u>Version Nr. of a board ty</u>	pe_Q9 b)	- Ballon	
Innovation Nr. 09	and the second s	and the second s	24 ³¹
'Assembly version AZ			
Manufacturing edition A	2	, in the second s	

- a) For a combination of board types, the more significant board is used for identification.
- b) If the number of board variations exceeds 10, the innovation number is also used.

The example shown above is for the position control board in the assembly variety 03 310A. The front plate of this board has here the designation 03 310A/B, but only the left PC board is present ; the two connector locations on the top right of the front plate are covered over. This board can be used for 3T or 3M with 3 axes, without S analogue. For the 03 310B assembly variety, both PC boards are present for use with four axes and/or S analogue for the 3M. For connectors arrangement, see section 3.5.2. For board assignment, see section 3.5.2.

On the back edge, the NC boards have only one 96-pole connector for the NC bus. In versions 2 and 3, the couple board 3 800 has, at the top, an additional 48-pole connector for the PC bus. The rest of the boards have two 48- pole connectors.

On the bottom, in the frame at the slot location, an identification strip carries the number of the board to be connected.

Information concerning board handling can be found in section 3.1 .

3-10

3.5.2 Logic Component Assembly

Assembly of basic version O



Remark concerning basic version 0 :

Only type 03 400 can be used for Input/Output PC board.



*) Location of the 6.3 mm faston connector for the auxiliary battery (see section 3.3.12)

<u>In exceptional cases</u> (PC malfunction) for version 2, if it becomes necessary to operate without the PC, all PC boards -including the I/O and the coupling boards- must be disconnected. The basic functions of the interface (without the 3rd. and 4th. decades and external data input) can be tested with an I/O board to the 96-pole adaptor +) on the NC bus, (so that there is no connection through the upper 48-pole connectors to the PC bus). In this special case, machine datum 409 bit 3 must be set to O.

⁺⁾see section 9.1

Assembly of version 3



*) Position of the 6.3 mm faston connector for the auxiliary battery

Remarks concerning basic version 3 :

The cable to the operator panel must be connected to PC board 03 810, on the X812 connector. PC board 03 810 : connector X813 is present only in model 03 811, the

connector remains unoccupied.

PC board 03 100 : connector X102 remains free.

<u>In exceptional cases</u> (PC malfunction) <u>for version 3</u>, should it become necessary to operate without the PC,PC board 03 800 must be disconnected, and a 03 400 board can be plugged into the reserved slot. The basic interface functions (without 3rd.and 4th. decades and external data input), can be tested in this manner. In this special case, machine datum 409, bit 3 must be set to zero.

Remarks concerning versions 0 through 3

1) Power supply 03501 2 Fault LED's

03502 1 Fault LED (only output voltage fault)

- 2) Engineering panel 03220, normally not assembled.
- 3) PC board 03260 can also be installed for board 03210 :

Program memory 8000 characters = 03260 E

- 16000 characters = 03260 F
- > 32000 characters = 03260 G

On the front plate there are two jacks for the connection of 5 V. The data can be protected before disconnecting the power supply,by connecting a battery.

									the second s
PC board	03 310 A			03 31	O B	and the .	03	3 320	
position control -	Act. 1	Act. 2 X313	Act. 1	Act. 2	Act. 4	Com.1	Com.2	Act.3	
	CAJ IZ	<u>v212</u>	AJIZ	<u></u>	<u></u>		<u></u>	<u>^2J2</u>	
3T 🦨	Х	Z	Not us	X,Z,S	-	analog			
3M X,Y,Z	X	YONY	Not us spindl	ed with e encode	X,Y,Z	4.,S	Z		
3M X,Y,Z and 4th. and/or spindle encoder	Not use 4 axes with sp encoder	d from up,resp. indle	X	Y	4th.	S	X,Y,Z	4.,5	Z

4) Association of position control boards and connectors

For the associations involved with the use of position control boards 03315, 03325, 03350, see the start-up instructions for version 4.

- Note: The new position control boards comprise a 14 bit DAC. Therefore Bit 0 of machine data 417 needs to be set to '1'.
- 5) Input/Output board: For 03400, the 03410, 03420, 03450, 03460 PC boards can also be used, or the S5 boards 420-3, 445-3, 444-3. 432-3.

3.5.3 Strappings

The boards are shipped from the factory with standard strappings. <u>No changes</u> are necessary for a <u>standard start-up</u>.Only for the following applications, the strapping on 03310 and 03320 must be changed (see section 8 through 9.14) :

- 1.) No differential input for position coders (TTL)
- 2.) Velocity control Ready-signal is not brought back from the NC by the drive (alarm 222)

3.5.4 Other SINUMERIK Input and Output Boards

The following input and output boards may currently be used :

Designation	Input	Output	Word Code
03 410	96	-	N83
03 402	-	48 × 0.5 A	N84
03 450	32	32 x D.1 A floating	N85
03 360	-	16 x 2 A	N82

The mounting width of these boards corresponds to $1 \ 2/3 \ \text{SEP} \ (25 \ \text{mm})$ of that of 03 400 (for fuses, see section 4.1.3).

3.5.5 SIMATIC S5 Input and Output Boards

It is possible to install SIMATIC S5 boards (6ES5 4..-3...e.g. N60, N65, N70, N90) into the PC section of versions 2 and 3,instead of the 03 400. The +24 V supply and load must hereby be taken into consideration! The guide rails for input and output boards are equipped for grounding with grounding strips (which should be checked and retrofitted,if necessary). An assembly kit,which can be ordered with order nr. 6FC3 428-4QV, can be provided for mounting the boards and covering the gaps (to ensure proper air conduction).

3.6 <u>Software System</u>

3.6.1 Executive Software

Is the control equipped with a valid software system ? The software edition is printed on the EPROMs.The latest software edition information can be found in the service circulars. The actual software edition can be displayed under test 381S. When exchanging EPROMs, the appropriate tools must be used in order to avoid damaging them.Damaged EPROMs cannot be returned to the factory, which is also true for EPROMs not carrying the GWE label.

For 24...40 pole ICs, central ware house Fürth has available IC-removal tool L 30460-X281-X.

3.6.2 SINUMERIK System 3 - Software Designation System

General guidelines to the designation system of EPROMs/PROMs

Since the introduction of System 3, a new designation system has been used for PROMs; it is based on the 12-digit GWE key. It takes into account the requirements of GWE, of development, process, and service.

The following key is at the basis for the identification of PROM/EPROM designation :

		GE <u>54</u>	ⁱ t x tx	r × ×
GWE Product Gr	oup			S.
Software	A.M.	A. A.		3345
System family	0099	2.2 ¹	32	
	and the second second		de la companya de la comp	j.
System type	02			all ^o
Modification or	f system type 09		in the second	S.
PROM location		<i>A</i> ¹		4
Software Editi	on 0099	Ka.91		~.
				101
Explanations f	or the key :			MICHORN -
Locations 1 -	4 : These locations identification	always carry the s in GWE data process	same numbers fo sing	r solution
Locations 5 -	6 : For the identif O and 2 have th 15	ication of the system ne number 11,and ver	tém family,vers rsion 3 has the	ions number
Location 7	: These positions O is basic syst	designate the system 3 (common to al.	tem type 1 types)	

Location 8 : These positions designate the modification of the system type O English * only version 3, from

- 1 German *
- 2 French *
- 3 Italian *
- 4 Spanish *
- 6 Engineering panel

Location

9 - 10 : PROM locations are numbered in ascending order of addressing, so that each PROM location has a unique number within the total system.

Location

- 11 12 : The software edition of PROMs changed during revisions, is identical to the last two positions of the number of the revision service circular.

If no general re-translation of the software system is necessary for the revision, the system can contain PROMs with different software editions.

software edition 02

General

: For the identification of a PROM/EPROM (outside the GWE), only the last 8 locations must be written.

The PROM location nr. in positions 9+10, and the software edition , in positions 11+12 (the two positions in front -respectively behind- the last decimal point), can be read directly from the PROM plate.

As before, the service circular contains information concerning the extent, edition, and the assembly of the respective system.

Example

: Order designation of the system software for 3T and 3M Versions O and 2, software edition O6 : 548 811.00XX.06 XX = PROM location

Version 2, software edition 02, English 548 815.00XX.02

Remarks

: The following software editions are valid and can be ordered for versions 0 and 2 :

- 05 delivered as of approximately 4/82
- O6 delivered as of approx. 5/83
- 07 delivered as of approx. 12/83

 08 delivered as of approx. 4/84For version 3,the following software editions are valid and available :

01

04

- O2 delivered as of approx. 5/83
- 03 delivered as of approx. 12/83 04 delivered as of approx. 5/84

The controls are not automatically up-dated to the new software editions.

The machine data described in this edition of the service manual correspond to software edition 07, respectively 03.

Current software edition and check sum of individual PROMs EPROM designation GE 548 811 00 XX XX

	<u>_</u>	PROM	PROM	Edition	So	ftware Ed	ition	and Chec	k SUM	2
FUNCTION	On	Toca-	tvoe	1	ED.	Check	ED.	Check	ED.	Check
°6,	board	tion		04	05	Sum	06	Sum	07	Sum
Versions	03200	01	2532	04	05	838A	06	82A1	07	8304
0 and 2		02	2532	04	05	62AF	D6	7322	1994 - A	712E
7 7 8 7 4		03	2532	04	05	1E50	06	288A	1	2E4A
		04	2532	04	05	263C	06	35D2	1	25.80
+ Uptions	~	05	2532	04	05	4431	06	SEF 8	1	7400 🔿
	12	06	2532	04	05	5433	D6 🔿	7607	ſ	7892
4	54	07	2532	04	05	BC75	06	B5CA		8903
· · · · · · · · · · · · · · · · · · ·		08	2532	04	05	ADC8	06	98 21		7490
~35		09	2532	04	05	68BE	06	7E6A		AD48
S.		10	2532	04	05	7F15	06	761B	(Å	7032
and the second		11	2532	04	05	F03E	06	DDA5	1 I	D6A4
		12	2532	04	05 🖄	C438	06	A596	1.	D358
		13	2532	-	05	F941	06	2608		FE7E
	2	14	2532		05	10CA	06	09F5		E1B3
	Nº0	15	2532	-X2	05	81F7	06	80F9		9985
	5° 1	16	2532	<u> 1</u>	05	6894	06	9204	1	78C7
		17	2532	04	05	2A34	06	38X	1.3	0998
200		18	2532	04	05	48A7	06	4052	07	1600
		19	2532	04	-	÷	-		18 <u>1</u>	
574		20	2532	04	- 3	d°-] -	35	-	
		21	2532	04	-	-	-		-	
	~~~~	22	2532	04	-	-	-	<u>.</u>	-	~
	No.X	31	2532	04	05	7F 35	06	6035	07	4304
2	5.7	32	2532	04	05	8798	06	<b>5</b> C95	07	488C
Total Nr.	of PROM	s	1000	20	20	1000	20		20	

- PROM not present

3.6.4 Diagnostic System Version 0 and 2

EPROM designation GE 548 811 06 XX XX

FUNCTION	PC BOARD	PROM location	PROM	Software Edition			
	454		type	01	02	03	
Engineering Panel	03220	71 72 73 74 75 76 77 78 79 80	2532 2532 - - - - - - -	01 01 - - - - - -	02 02 - - - - - -	03 03 - - - - -	

All 3 software editions are valid for the engineering panel program.

# 3.6.5 Operating System Version 3

Current software edition and check sum of individual PROMs EPROM designation GE 548 815 DX XX XX

ani O.	4	PROM	PROM	Soft	ware Edit	ion an	d Check S	um.	3
FUNCTION	PC BUARD	Location	Туре 🚽	01	Check Sum	02	Check Sum	03	Check Sum
Version 3	03200	01	2532	01	79CF	02	92BC	03	8FFD
ST and SM	5	02	•	01	50FA	02	589E	d.	7C34
Ontions		03	1	01	4DE8	02	5301		52E7
operons		04		01	4320	02	5732	Ċ	5D40
		05	I I	01	6610	02	85F8		A5D4
		06	(	01	82A3	02	8C81	1	8CBA
24.	27.	07	2	01	17E5	02	FD04	(	2AE6
		08		01	2042	02	E3E1		BC 30
	1	09	2.	01	375A	02	6F6F	ાજે	359C
	fe.	10 🔄	20	10	5IDC	02	3526	X	4466
		$11^{\circ}$	1	01	COES	02	026F	5	FF9A
		12	1	01	ASDA	02	F68D	- 1	F728
		13		01	EDCB	02	BCD9	۰ <b>۱</b>	D30F
	de.	^{&gt;} 14	1	01	1372	02	B227	-	A37D
	and the second sec	15	j.	01	DE99	02	54C1	. 1	798A
		16		01	EB64	02	77F8	1	5C67
		17	2	01	BB41	02	ABC0	1.2	871A
	102	18		01	CIE7	02	CIB3	03	C116
	5	19	(	01	B819	-	-	d -	
		20		01	A8F2	-	°., -	-	
	*	25	l.		1 - 1	02	+.S~	03	•
	*	26		S.	-	02	E -	1	•
	12 A	27	1	1200	-	02	≥° <b>+</b>	1	+ _S ³
	*	28	1 3	- 1	-	02	*	1	• 3
		31	2532	01	4D27	02	2BB5	1	OE94
	R	32	2532	01	5B86	02 ·	2F60	03	185F
Total Number	of PROMs	Carl		22	Cont -	24	~	S.	

* Language EPROMs, optional

Language EPROMs

FUNCTION (Language)	Modification and Location	PC BOARD	PROM Type	Sof 02	tware Edi Check Sum	tion : 03	and Check Check Sum	Sum
English	025 026 027 028	03200	2532	02 02 02 02 02	1326 A710 F552 FCF2	03 , , ,	1959 9F3E 6011 464B	
German	125 126 127 128	03200	2532	02 02 02 02	2FD8 AB04 9625 9A1A		32BA A84A 993B 9639	onadhaid
French	225 226 227 228	03200	2532	02 02 02 02 02	0F60 A53F 2062 IA35		0957 AB44 1477 1BDD	201
Italian	325 326 327 328	03200	2532	02 02 02 02 02	1A07 A7C7 1182 FA47	1	13E0 AE27 F4E7 0D0C	ornadi
Spanish	425 426 427 428	03200	2532	02 02 02 02	14E4 A172 8CB8 76B0	' ( 03	1899 9F6A 7219 8913	28742.D

Anna Gastor

3.6.6 Diagnostic System , Version 3

EPROM designation GE 548 815 06 XX XX

FUNCTION	00 80400	PROM	PROM	Software Edition		
	PL DUARU	COMPTION	Type	01	02	
Engineering	03220	71	2532	01	02	
Panel		72	2532	01	02	
	Ŝ.	73	-	-	- 2	
	2	74	-	-	X ²⁰	
		75	-	- 8		
		76	-	3°	-	
	8	77	-	æ~-	-	
	A	78	- A	<b>-</b>	-	
	3525	79	- 34	-	- 4	
		80	-	-	-	

Both software editions are valid for the engineering panel program.

# 3.6.7 PC Diagnostic System for Version 3

EPROM designation GE 548 815 00 XX XX

FUNCTION		PROM	PROM	Software Edition		
	PL BUARD	Location	гуре	01	14	
Engineering Panel and Diagnosis	03220	71 72 73 74 75 76 77 78 79 80	2532	01 01 01 01 01 01 01 01 01 01	red sur	

The PC diangostic system can be installed independently of the software edition of the operating system.

#### 3.7 Remarks Concerning Version 1

Version 1 has been discontinued. The following are characteristic of version 1 : The same software system as for versions 0 and 2, The same logic frame width as for version 2 without PC , Only type 03400 can be used for input/output PC boards , Two input/output boards can be installed,unlike in version 0 ; when installing the 2nd. I/O board,machine data bit 409 bit 6 must be set. This results in 4-decade S-value output,and makes possible the "External Data Input" option (as in version 2).

3.8 <u>Remarks Concerning Versions D and 2, with Software Edition D4</u> The following differences exist in comparison with edition D5 :

- Machine data 365, 366, 385 are not present.
- Standard machine data cannot be loaded according to section 4.5 .
- Machine data must be entered for the 4th. axis, even if such axis is not present.
- As of software edition 04, there are option EPROMs in locations 17 22. Even functions such as "Cycle", "Cutter Radius Compensation", are not possible without these option EPROMs.
- With software edition 04,RAM board 03210 ¥can also be shipped with 4k program memory. This board type cannot be used with other software editions.

- The software edition contains machine data 382 (serial number). A machine data tape generated prior to the switch to software edition 4 causes,during the reading of datum 382,an alarm and reader stop. In this case,the data up to 381 are read correctly. The remaining data (385 through 419) must be entered manually. It is recommended, that a machine data tape be ultimately punched for the new software edition. 4. <u>Voltage and Function Test, Erasing the Memories,</u> <u>Machine Data Input</u>

#### Contents

- 4.1 Voltage Test
- 4.2 Function Test
- 4.3 Erasing the Memories (Cancel Operations)
- 4.4 Determining the Control Type (After Clearing the Machine Data)
- 4.5 Loading of Standard Machine Data
- 4.6 Machine and Setting Data Input
- 4.7 Constructing and Handling the Machine Data Tapes, Drift Compensation
- 4.8 Example of Machine Data for a Lathe
## 4.1 Voltage Test

#### 4.1.1 Voltage Supply

The current for the power supply 03500 has 24 V_. Ceck <u>before</u> connecting the power supply ! Check the rated input voltage on the terminal strip :

Line voltage+ 24 V_ (20 V_...30 V_)Temperature range0 through + 55°CTemperature monitor63°C ± 2.8°CFan line voltage220 V 50 HzFan monitorw/o monitor,E/F jumperNC ON push-button0

# Ë,F

Terminal C1. D1

G, H

## 4.1.2 Power-up Phase

The interface cables should not be connected yet.Axis movement should be inhibited, and the command value connector for the position control loop should be disconnected.

Switch the control on (activate the NC ON push-button for ca. 1 s) Is the control in operating state ? Can you see the basic display of the selected operating mode ? (See section 12 if test board 03220 is being used)

If these points are not satisfied, also check the voltage supply and fuses. During this test, the power supply must be separated from the equipment.

#### 4.1.3 <u>Fuses</u>

#### <u>NC</u> :

Mounting location	Designation	Rated current	
Power supply (03500)	F 30 F 161	16 A fast O.8 A medium lag	50 5
I/O board (03400)	F 1	1.6 A medium lag	
Output board (03421)	F 1	16 A FF	
I/O board (03450)	F 1	1.6 A medium lag	
Output board (03460)	F 1	1.6 A medium lag	
Operator panel (03700)	F 1	4 A medium lag	
Operator panel (03780)	F 1	2.5 A medium lag	

#### 4.1.4 DC Voltage

A 5 V supply should be measured on the power supply 03500 (jack 5 V against M).Adjust the rated voltage to > 5.15 V...5.25 V with potentiometer R145 on the front plate (clockwise = higher voltage).This ensures the IC supply voltage (voltage drops on the back plane and PC board wiring are taken into account).The 5 V supply is set correctly in the factory,and normally does not have to be adjusted during start-up.

#### 4.1.5 Error Storage while the NC is Switched Off

If power supply 03501 is used : 2 LEDs indicate whether the shut-off was triggered through the input voltage or the internal NC voltages. The cause for the shut-off is stored, and can be displayed while the control is off, via the "Error Monitoring" key. The shut-off cause remains stored until the next shut-off.

If power supply 03502 is used : only 1 LED is present, and it lights up only after activating the "Error Monitoring" key, if the shut-off was caused by any of the 3 output voltages (5 V,  $\pm$  15 V _). It does not light up for input supply errors of devective power supply.

## 4.2 Function Test

4.2.1 CPU Cycle Monitoring on PC Board 03100 at Power-up :

LED indicates : Version 2 up to software O6

Version 3 up to software O2

LED flashes with approx. 2 Hz : Version 2 up to software 07 Version 3 up to software 03

LED flashes with approx. 4 Hz Version 2 up to software 07 Version 3 up to software 03

LED lights up in all versions : CPU error EPROM error

Test board activated but not connected False machine data

PC does not start Battery alarm is displayed or general errors (see below)

PC not functioning

Battery alarm is displayed

Defective NC bus

Incorrectly strapped PC board (address, WAIT) Position control-,EPROM-,RAM-,PC interface defective

4.2.2 CPU Cycle Monitoring during Operation :

LED lights up : Hardware fault DMA to PC not possible

4.2.3 Check Sum Test of the System Program Memory

Operation : 1. Set switch S3 (inside) on the CPU in position 2 (top). This is the normal position, if no engineering panel mode is conducted with the test board.

2. System Reset (e.g.during power-up: PORESET)

This starts the check sum test procedure. If a defective EPROM chip is discovered, the display is :

> EPROM - ERROR - FOUND EDITION __ (Software edition) CHIP __ ACT/SET-SUM ____/___

(Chip number location nr. decimal)

If no error is found, the processor jumps immediately into normal system program.

3. Further defective chips are displayed by activating the page key.Eventually, the processor jumps into the normal system program if no other defective chips are found.

All connected system program chips are tested.

This PROM check is executed automatically with each PORESET (power-up). If any deviations are found in the check sum (PROM missing or in the wrong location), the display indicates the location nr., the correct check sum, and the actual check sum.

EPROMs with GWE stickers are checked automatically during "burning".

## 4.2.4 Adjusting the CRT Brightness in Version 3

CRT brightness can be adjusted with potentiomater R18, on board 03780.



#### ATTENTION !

High voltage of ca. 16 kV in the the CRT component on the high voltage transformer,anode wire, and anode connection on the CRT.

## 4.2.5 Remarks Concerning the CRT in Version 3

Contrast adjustment: normally handled by the manufacturer; if readjustment is necessary, note the following procedure :

Set optimal brightness (R18).Select actual value display (in large letters). Potentiometer R17 (03780) full counter clockwise turn.Subsequently turn R17 clockwise until optimal brightness,sharpness,and contrast are achieved.

CRT cleaning: the CRT is neither acid nor scratch-proof. Magnetic interference: if the CRT unit is exposed to magnetic fields, the CRT display can become subject to oscillations.Devices that generate electromagnetic fields should be located no less than 300 mm away from the CRT.

## 4.2.6 Emergency-stop and Limit-switch Test

With the control off,plug in the interface cables. The functioning of the E-stop and limit-switch are tested without drive enable (velocity command value disconnected). The interface test (see section 8) can hereby be used.

## 4.3 <u>Clearing the Memory Ranges (Cancel)</u>

The following situations require cancel operations : Exchanging the power supply 03500 Exchanging RAM memories 03210/03260 Exchanging the system software on 03200 If undefined displays appear in the selection display If certain memory ranges need to be cleared. Canceling is accomplished by pressing the "Cancel" 😥 and the appropriate number keys simultaneously. In this operation, the control is started again with a hardware reset, e.g. the reset key on 03500, or through new power-up. The keys must be pressed until the basic display reappears. The following ranges can be cancelled :

Power on reset

PC RAM memory (initial clear) _

Machine data cancel

Cancel user program (PP and SR) 👡

Cancel setting data (TO and ZO) 🔪

Subsequently, the system should run normally.

#### Remark :

If switch S3 on the CPU is in position 2 (down), and no test board is connected, undefined displays can also appear, but in this case cancel operations cannot be executed because the CPU goes into stop-state.

## 4.3.1 Cancel O (in Conjunction with the PC)

The following sequence must be adhered to :

- 1. Machine data nr. 409 bits 7 and 3 must be set.
- 2. PC switch on Stop.
- 3. Press the Cancel and O keys simultaneously, and activate hereby the hardware reset (on the power supply).
- 4. PC switch from Stop to Operation.
- 5. PC switch from Operation to Stop.
- 6. PC switch from Stop to Operation.
- 7. Hardware reset (on power supply).

#### 4.3.2 Cancel 2 (Machine Data)

Cancel 2 is not necessary before loading the standard machine data according to section 4.5 ,because all the machine data memory is overwritten with inputs 3 through 6.

# 4.3.3 Cancel 3 (Part Programs and Subroutines)

When using cancel 3, consideration must be given to the fact, that all standard and option cycles are cleared as well.

## 4.3.4 <u>Cancel 4</u> (TO and ZO)

As of software edition 02, version 3 has with options B76 or B78, a background memory of 100 R parameter values. For the B78 option, this background memory contains the machine data for in-process gauging. These data can also be cleared with "Cancel 4".

## 4.4 Loading the Standard Machine Data, and simultaneously,

## Establishing the Control Type

With this function, machine data which are firm-stored in the EPROM range of the control can be loaded in the machine data storage, with an operation during control power-up.

Operation for :

3T/0		8	2.		
3M/0		:	4_]	+ 文 +	Power On Reset
3T/2	3T/3	:	<b>[</b> ]	Ratton -	(or NC power-up)
3M/2	<b>3</b> M/3	:	6.		

The keys must be activated simultaneously, until the basic display appears. For stored standard machine data, see section 2.1 and 2.2 .Changing of specific individual machine data according to section 4.6 .

# 4.5 <u>Establishing the Control Type without Changing the Machine Data</u> (for Test Purposes)

After clearing the RAM memory with "Cancel 2" (for machine data see section 4.2) or after exchanging power supply units or the RAM memory board, the control type can be established <u>without</u> setting the <u>standard machine</u> <u>data</u>. The control type is then stored in RAM.



Power On Reset = 3 T without standard machine data transfer

Power On Reset = 3 M

Keys 1 or 2 and  $[\checkmark]$  must be pressed until the basic display appears. The loading of the standard machine data (section 4.4) transmits the control type automatically. 4-8

4.6 General Information Concerning the Input of Machine and Setting Data

Page TEST And MDI-SE-TE

Independent of operating mode, except DO/DI, with the module key in the TEST position, there is among other things, a group of pages for the display of machine data.

This group of pages has a preselect cursor that,after module key activation,will return to the last displayed page of this selection group.

It is possible to change the corresponding cursor by activating :

the page key

The cursor generally moves line by line; if it leaves the display, the next page will be shown.

Manual change of values can only be done in the MDI-SE-TE operating mode:

Operating mode : 👌

Page (Mode) 🍕

Complete value blocks cannot be cleared. The values are also protected by a data safety switch S1 on the front plate of the CPU. Only the setting of drift compensation values is independent of the data safety switch. Machine data are never entered with decimal point; some values can carry a sign.

The lowest input value is O, respectively 1.

The highest allowed input value is determined from the machine data list (see section 2).Do not used any values, not even 0, for unused input numbers; this would inhibit the tape input, and would cause an alarm. With MDI, these unused input numbers are locked.

#### Example :

## Select TEST/Axis-specific Machine Data Values

Select display page TEST and Ident-Nr. via keys ... or ... The display of machine data is possible in all operating modes, except DATA IN/OUT.



4-9

In the MDI of machine data bits, leading zeroes may be skipped, e.g. 403 : S 101010 is automatically complemented to 00101010. The "S" address character must be entered in front of the bit pattern, respectively numerical value.

## 4.7 Constructing and Handling the Machine Data Tape

#### 4.7.1 Constructing the Machine Data Tape

Read a standard machine data tape with already known values of the machine.

Tape construction :

#### Remarks :

"N" for the Ident-Nr. is mandatory for the machine data tape."N" does not appear during manual machine data input.In the service manual the Ident-Nr. is sometimes only indicated by "Nr.".

#### 4.7.2 Preparing for Reading the Tape

Load the standard machine data according to 4.4, and then set the data safety switch in the "up" position (S1 on the CPU front pLate as in the 03100 circuit diagram).

Reset

TE

MDI-SE-TE

Operating mode for MDI of required machine data

Check the required machine data nr. 409, 411, and 416 (see below). If the machine data are not entered acc. to 4.4, or if the input device does not agree with the machine data entered, inputs -respectively changesmust be made manually.

MDIs are made under Ident-Nr. 411 for the interface device designation and baud rate (see machine data bit list).For the operating mode selection switch to be functional,bit 7 under Ident-Nr. 409 and bit 0 under Ident-Nr. 416 must be set to 1.

If you are dealing with an integrated PC,bits 3 and 7 under Ident-Nr. 409 must be entered simultaneously, and be activated with PORESET (power on-off). After the device name and baud rate have been established in the control, the tape can be read.

## 4.7.3 Loading the Machine Data Tape

Operating mode selector switch in desired position

Data input Data Start



The statement "Control in action" appears in the bottom display line, until the tape has been loaded.

Note : If the Test board is active, "Control in action" does not appear in versions 0 and 2, but machine data tape loading is possible.

The values can eventually also be edited manually. A drift compensation must also be made.

Select : TEST and MDI-SE-TE

> Nr. 230 S + [3]

(see section 5.5)

After entering the machine data, return the data safety switch to its normal position (switch in "down" position on the CPU front plate). Standard alarms (e.g. position control loop, etc) do not inhibit the loading of the tape.

Example of Machine Dat	a of a Lathe	
Version 3, software ec	lition O2	
$Kv = 1 \frac{m/\min}{\min}$ , $X = 5 m$	n/min, Z = 10 m/min ,for	U _{max} = ±
STE े		¢ I
N100 S+20	X 1 X	
N101 5+20	N232 5+0	N400 S
N102 5+0	N233 S+0	N401 5
N103 S+0	N350 S+400	N402 S
N110 S+100	N351 S+0	N403 S
N111 S+100	N352 S+0	N404 S
N112 S+0	N353 5+500	N405 S
N113 S+0	N354 S+2400	N406 S
N120 5-80	N355 S+10	N407 S
N121 S-80	N356 S+10	N408 S
N122 S+0	N357 S+0	N409 S
N123 S+0	N358 S+1	N410 S
NI 30 5- 5000	N359 5+100	N411
N131 S-10000	N360 5+200	N412 S
N132 S+0	N361 5-400	N413 5
N132 5-0	N362 5-800	N414 5
N140 5-2048	N363 5-1600	N415
N141 5-2048	N364 5-3200	N416
N141 5+2040	N365 5-4000	N417
N142 3+0	N366 5-4000	N418
N145 5+0	N371 5,2000	N419
N120 3+1000	N372 5, 5000	M02
N131 3+1660	N372 3+3000	MUZ
NIJZ S+U	N373 3+3000	
N133 3+U	N374 3+2000	
N160 5+452000	N373 3+3000	
N161 3+1302000	N376 5+300	
N162 5+0	N377 5+10	
N163 S+U	N378 5+300	
N170 5-2000	N3/7 5+200	
N171 S+150000	N380 5+11	
N172 S+0	N 381 3+2	
N173 S+0	N 383 3+2	
N180 S+450000	N383 3-77777777	
N181 5+1300000	N385 3+0	
NIBZ S+U	N38/ 3+0	
N183 S+U	N388 3+U	
N190 5+3	N387 5+0	
N191 5+8	N370 3+0	
N192 5+0	N 371 3+0	
N193 5+0	N 372 3+U	
N200 S+0	N 373 3+0	
N201 S+0		
N2U2 3+U	Ne Ne	
N2U3 5+0	and the second sec	
N210 3+0	10 10	
N211 3+U	Service Service	
N212 5+0	5S	
N213 3+U N220 6 5400		
N331 6.3700	-2°	
19221 J+2/VU N333 8.A	N	
N222 5-0		
N227 5-2	and the second	
N921 C_1	die die	
いちようし コード		

4-12 + ----

9 V

S

4.8

## 5. Manual Start-up with System Software

#### Content

- 5.1 Machine Data
- 5.2 Control Polarity of Feed Axes
- 5.3 Closing the Velocity Control Loop
- 5.4 Manual Movement (with Corresponding Alarms)
- 5.5 Drift Compensation
- 5.6 Tests for all Manual Functions
- 5.7 Program Execution

#### 5.1 Machine Data

The standard machine data can be loaded according to section 4.4 , special machine data can be altered manually, or the corresponding machine data tape can be loaded according to section 4.6.3, or all the machine data can be entered manually.List all the machine data used as described in section 1.2, and file them in the log book.If possible,keep a copy of the machine data tape by the control.

#### 5.2 Control Polarity of Feed Axes

Setting the polarity incorrectly causes the axis to move uncontrolled, with maximum velocity.

For this reason, it is very important to check the position control and velocity control polarities before closing the control loop.

## Handling :

Keep in mind : The direction of the feed axis (based on customer's statement, or according to ISO standards).

What <u>polarity of the velocity command</u> <u>value</u> causes the axis to move in positive direction ? (customer's statement, resp. test with battery box)

Check the position control polarity : by moving the feed axis mechanically in positive direction.

Note the <u>direction of the actual value</u> <u>change</u> from the actual value display.

Set the machine data bits for sign change for <u>velocity command value</u> (Nr. 403...406, bit 1), for sign change for the <u>partial actual value</u> (Nr. 403...406, bit 2).

#### Example :

Axis motion in positive direction ; hereby, the polarity of the velocity command value :



In TEST nr. 403 - nr. 406, bit 1 and bit 2 (bit 1,sign change for velocity command value) (bit 2, sign change for partial actual value), the appropriate combination is entered for each axis.

## 5.3 <u>Closing the Velocity Control Loop</u>

Shut the control off,plug in the command value connector,and remove any interlocks of the particular axis (fuses,control inhibit).All other axes should be still locked.Power the control up.

Warning : Activate the emergency-stop if the feed axis start running away uncontrolled.

Possible causes for a run-away axis :

a) The position control loop <u>or</u> the velocity control loop has wrong polarity :

False machine data bits.

Characteristic : the axis moves with maximum velocity.

b) Position control loop not closed :
 Cause : the encoder does not follow the axis movement.
 Characteristic : the axis moves with low constant velocity.

Either a ground-short, interruption, or a short circuit trip the position control monitor.

c) The command value does not reach the velocity control : Characteristic : the axis runs with constant low velocity (drift).

d) Control loop error :

Causes : tacho feedback interrupted

improper polarity for tacho feedback
incorrect optimization

Kv factor too high

Characteristic : the axis oscillates strongly

# 5.4 Manual Run

Check that all the command value cables of the axes are connected, and the control has the correct polarity. Also, the position control loop should be closed, and the gains should be properly set.

The following alarms can also inhibit the motion of the axes :

Axis	Alarm									
6	223	E-stop (emergency-stop)		8	5					
3	222	Servo control fault - velocity control not ready								
1	001	Software limit	alle .	Lipita astabliabod						
2	011	switches		Limits established						
3	021	positive direction		with machine data						
4	031	approach	14	160 163	4					
100	002	Software			<u>,</u> ,					
2	012	limit switch		limits given by 🚿						
3	022	negative direction		machine data						
4	032	approach		170 173						
1	005		A.A.	and in the second s	25					
2	015	The interface has remove	d the contr	ol enable						
3	025	of a movino axis								
4	035	NOR NOR								
1	102	The velocity command val	ue is too h	ioh						
2	112	The trigger is set with								
3	122	machine data 354								
4	132				and a second					
1	101	Clamping error								
2	111	Axis is not in position								
3	121	The trigger is set with	machine dat	a 110 113	· .					
4	131	NOTION STREET								
1	103	Contour monitor	, Spar	Bar						
2	113	Trioger is set with mach	nine data 3	51 and 352						
3	123									
4	133									
1	104	Control loop hardware fa	ault	No. X	2					
2	114	The monitor of the posit	ion loop si	onal has tripped fo	r the					
3	124	axes.respectively spindl	le Sol-		-					
4	134	·,	Jon -							
1	108	Position control fault.	contaminati	on www.	344					
2	118									
3	128									
4	138									
	1 .00									

The following signals are also necessary for manual runs (no alarm trigger) :

Feed release No axis lock Control enable X, Y, Z, 4th.

After activating the direction key, if the feed release and control enable have not been given, the "Feed hold" lamp on the operating panel lights up.

## 5.5 Drift Compensation

The drift compensation should be made when the control loop is closed for all the axes, and the drives are under control. If this is not done, the axes may not move (indicator light "Machine running" stays on.)

Procedure :



Important : The drift compensation must be done for each axis individually. The data safety switch may be left in its normal position.

## 5.6 Testing all Manual Functions

Limit switch Increment Reference point approach

## 5.7 Running a Program

Here, only the principle programs must be tested, so that programs may be utilized as optimization aids.

The following interface signals are also necessary for this purpose : " Read Enable " ,

and, under specific order from the interface only : " NC Start" .

If the feed release is interlocked in the interface with the spindle speed,tool number input,auxiliary function input or others,then this output ought to be possible.

Check whether axes movement is possible through the program memory.

## 6 Drive Optimization, Drive Monitor, and Einishing: Touches

#### Contents

- 6.1 Tacho Adjustments and Definition of the Maximum Command Value
- 6.2 Mult-gain Factor
- 6.3 Position Control Loop Gain (Kv Factor)
- 6.4 Acceleration
- 6.5 Cut-off Feed Rate
- 6.6 Position Monitors
- 6.7 Contour Monitor
- 6.8 Analogue Spindle Speed
- 6.9 Finishing Touches

# 6.1 <u>Tacho Adjustment and Definition of the Maximum Command Value</u>

The axis-specific maximal velocities Test-nr. 130 - nr. 133 selected by the customer, must be associated to a particular tacho voltage.

Keep in mind, that another 10 % control reserve will be needed here. The natural limits are determined by the position control board (10 V) or the servo control of the drive.

#### Case A :

The maximal allowed input voltage for the drive-servo unit :  $\geq$  10 V. Value 2048 is entered in Test nr. 140 - 143 (up to 10 V can be entered for the velocity command value; 2048 VELO = 10 V).

The maximal axis velocity must, however, be reached at 9 V already (10 % needed as control reserve).

#### Tacho Adjustment

The adjustment should be made at low velocity and low velocity command values.

Measuring point : the velocity command value at the drive's servo unit, with a defined velocity (e.g. manual) generated by the NC.Adjustment at potentiometer, tacho adjustment at the control.

#### Case B :

The servo unit of the drive must be limited to a velocity command value voltage of less than 10 V.

Value 1024 is entered, for example, in Test nr. 140-143 (the maximum command value input is 5 V).

The maximal axis velocity must be reached at 4 V (for tacho adjustment see Case A).

The velocity command value voltage can be limited by the NC via the machine datum Test nr. 140 - 143.

Conversion : 10 V correspond to approximately 2048 units (VELO).

The limitation entered under Test nr. 140 - 143 may not be reached during operation.

Since higher command value voltages result in better control behavior, it is generally preferable to use case A if possible.

#### 6.2 Mult-gain Factor

Test nr. 220 - nr. 223

A mult-gain factor must be entered for the calculation of the velocity command value.

This allows axes to be driven with different maximal velocities, while using the command value input fully.

Axes which move jointly in contouring modes, must have equal position control loop gains. Such will be the case, if the value for each axis is derived according to the following formula :



For rotary axes :

$$MULTGAIN = \frac{3 \cdot 10^7}{V_{max} \frac{Grad}{Min}} \qquad \frac{U_{max} [V]}{10 [V]}$$

In inches (input system  $\frac{1}{2} \cdot 10^{-4}$  in)

$$\text{MULTGAIN} = \frac{3 \cdot 10^7}{\text{V}_{\text{max}} \left[\frac{\text{inch}}{\text{min}}\right]} \cdot \frac{\text{U}_{\text{max}} \left[\text{V}\right]}{10 \left[\text{V}\right]}$$

Vmax = Maximal axis velocity,as set under Test nr. 130 - nr. 133
 (maximum velocity)

 $U_{max}$  = Velocity command value voltage for  $V_{max}$  after tacho adjustment

_ V ma;	5 ⁰⁰				U
<u>Kin</u> (	4 7	5.V	8 7	<u>9 V</u>	
15	34	1000	1600	1800	-25
14	~	1071	1714	1929	{
13	NO.X	1154 🖉	1846	2077	ł
12	100	1250	2000	2250	
11		1354	2182	2455	
10	52	1500	2400	2700	~3
90		S 1667	2667	3000	5
8	Sec. Sec.	1875	3000	3375	And Carlow
~ 7	14	2143	3429	3857	24
6		2500	4000	4500	
5	S.	3000	4800	5400	
- 4	Nº Star	3750	6000	6750	
3	- Con	5000	8000	9000	
2	3 ⁰	7500	12000	13500	
1.8	12000	15000	24000	27000	
0.8	15000	18750	30000	32000	3Nº
0.75	16000	20000	32000		250
0.6	20000	25000			
0.5	24000	30000		8	
0.4	30000	32000		Stax.	

Table for Mult-gain Input Values :

6-3

Examples :

- a) Kv factors of all axes
   Maximum velocity of all axes
   Command value correction of all axes
   i.e. mult-gain for all axes
- b) Kv factors of all axes
   Maximum velocity of all axes
   Command value corrections for all axes

i.e. mult-gain for all axes

c) Kv factors of all axes Maximum velocity of all axes Command value corrections for all axes

i.e. mult-gain for all axes

= (X, Z=1 m/min/mm) = (X, Z=10 m/min) = ( Umax X, Z=8 V) = (X, Z=2400) = (X, Z=1 m/min/mm) ≠ (X=10 m/min, Z=15 m/min) = ( Umax X, Z=8 V)

≠ (X=2400, Z=1600)

= (X, Z=1 m/min/mm) ≠ (X=1m/min, Z=15 m/min) s ≠ (Umax X=4 V, Z=8 V)

 $= \text{ or } \neq (X=12000, Z=1600)$ 

## 6.3 Position Control Loop Gain (Kv Factor)

Definition :

 $K_{v} = \frac{Velocity}{Following Error} \left[ \frac{m/min}{mm} \right] (Kv unit according to VDI standards)$ 

## Generalities :

To keep contour deviations to a minimum during contouring operations, it is necessary to have a large Kv factor value.Kv values that are too high however, lead to instability, overshoots, and finally, to inadmiss bly high machine loads (wear).

The maximum allowed Kv factor value depends on :

Design, respectively rapidity of the drives (control response time, acceleration/deceleration ranges), and quality of the machine.

In practice with production machines, such values have been found empirically to lie within 1 and 1.5 m/min/mm, in 80% of the cases. In these cases, the empirically found value should be set and tested with checks of possible instability or overshooting.

<u>Important</u> : A good velocity control optimization is always the prerequisite for a correct adjustment of the K_v factor.

#### Procedure :

Lower the acceleration (TEST nr. 120 - 123).

The overshoot behavior is the determining factor in the estimation of the Kv value.For this reason, the acceleration should not be set so high as to allow the drive to reach its current limit.

If the drive is to achieve an acceleration of 1 m/sec², as a precaution, it is better to halve this value :

0.5 m/sec² ⇒ input : 50

#### Kv Value Adjustment

The position control loop gain is entered under TEST nr. 150 -153 according to the following conversion formula :

 $K_{V}(0.01 \text{ s}^{-1}) = \frac{5000}{3} \cdot K_{V} \frac{m/min}{m}$ - 1666 . K. - - -

Thus, the numerical value 1666 is entered for a Kv factor of 1.

To determine a proper positioning behavior and the maximum value, it is advisable to select the axis, used in the contouring operation, which shows the poorest dynamic behavior. The command value voltage n_{comm} at the velocity control should be measured with a storage oscilloscope or a ink-jet recorder. The machine should move at maximum feed rate.

comm [v] n act [ms]

Since overshooting is observed, the Kv factor must be too high. In most cases overshooting can already be detected from the following error (TEST nr. 800 - 803).

n comm fy: n_{act}

[ms]

The Kv factor is low enough, so that no overshooting occurs. This can be double-checked by observing the deceleration on the oscilloscope or ink-jet recorder, with higher vertical amplification.

The following factors can also cause overshooting : Acceleration is too high (the current limit is reached). Control response time of the velocity control loop is too long. Velocity control error (reoptimization may be necessary). Mechanical backlash. Skewing in the mechanical portion.

Load changes (vertical axes).

As a precaution, it is advisable not to select the highest possible Kv factor, but a value that is at least 10% lower than that. Axes which participate together in contouring processes, must have the same Kv factor.

# Testing the Position Loop Gain (Kv Factor)

The magnitude of the following error can be determined under TEST nr. 800 - 803. If the drift has been compensated for, the value displayed for positive and negative directions at equal velocities will be the same.

Finally, the Kv factor value set for all axes must be checked during driving, over the display of the following error. The accuracy of contouring operations is based on identical dynamic behaviors of all axes, i.e. at the same velocity, the following error must be the same. If there are any deviations, the differences in mult-gain factors or velocity control must be adjusted.

## 6.4 Acceleration TEST Nr. 120 - 123

The axes are accelerated and decelerated with the values entered :

a | 10⁻² m/(sec² |

This makes it possible to accelerate to velocity and decelerate into position rapidly, accurately, and with no undue strain on the machine.

The customer should provide information concerning the machine's proper continuous brake deceleration. This value, if the machine is not overloaded, should be entered under TEST nr. 120 - 123.

Generally, these values lie between :

0.3 m/sec² and

 $2 |m/sec^2|$ 

Check, respectively determination of the acceleration values :

Criterion :

TEST nr. 120 - 123 : acceleration without overshoots, respectively positioning at rapid traverse rate (maximum velocity).

Under worst-case load conditions (heavy workpiece on table) : Instruments : chart recorder or storage oscilloscope

Measurements: velocity command value,

and possibly current actual value and velocity control output.

After setting the acceleration, the machine is run at rapid traverse rate, and the current actual values, and possibly the velocity control output, are recorded. From these measurements, it is possible to see whether or not the current limit was reached. The drive may reach this limit momentarily, but only in the rapid traverse range. For an interval before positioning, the drive must again be within velocity control, because the axis will otherwise overshoot its position. Example of 6-pulse circulating current-free feed drive with current limitation control :



Example 1 : well selected acceleration

Example 2 : acceleration is too high ; the current limit is reached. Due to the larger following error, the command value overshoots within the position control loop.

The acceleration value entered should be at least 10% lower than the ideal value, in order to avoid reaching the current limit following only slight load changes (such as may result due to heavy spots or lube effects).

To protect the mechanical parts, the customer may want to set this value even lower than that.

The axes can have different acceleration values.

## 6.5 Cut-off Velocity

TEST Nr. 350

For reference point approach and positioning out of higher velocities, it is necessary to select an appropriate velocity.

Recommended values :

Nr. 350 S 500 mm/min

#### 6.6 Position Monitors

#### TEST Nr. 100 - 103 , Position Tolerance

The approached position is checked after the dwell time for position monitoring TEST nr. 353 has elapsed. If at this time the following error is larger than the value entered under TEST 100 - 103, the Y"Machine not in position" remains on; further motion is inhibited.

#### Setting :

The in-position accuracy depends on the quality of the position control and velocity control loops.

Normal deviations can be determined by monitoring the following error at standstill.

According to the customer's request and the positioning accuracy reached, the setting value should lie between 10 µm and 50 µm,but it should be at least twice as high as the maximum deviation of the following error at standstill.

## TEST Nr. 110 - 113

Alarms 101, 111, 121, 131 are displayed, if,after the elapse of TEST 353, one of the axes is pushed out of position at standstill (clamping and control inhibit). The machine manufacturer has the task of keeping this deviation very low, if possible below the position tolerance set under TEST nr. 100 - 103. The clamp limit under TEST 110 -113 should be set at twice the value under TEST nr. 100 - 103.

#### Recommended value :

TEST nr. 110 - 113 between 50 µm and 200 µm. This also applies if none of the axes are clamped. This locks the control (control inhibit) if the position control loop is faulty (drifting).

#### TEST NR. 353 Dwell Time for Position Monitoring

This machine datum affects the clamping limit determined for TEST nr. 110 - 113 (see the machine data description ,chapter 11).

If the clamp limit is checked too early (some following error still present) or if the drive overshoots, alarms 101, 111, 121, and 131 may be triggered.

The time interval set under TEST nr. 353 must be sufficiently large to allow the drive to come to a complete stop before the clamp limit is checked.

TEST nr. 353 is entered in 1 ms units.

Reasonable values are between 160 and 1600 ms.500 is considered the standard value.

#### 6.7 <u>Contour Monitoring</u>

Contour monitoring functions according to the following principle :

After an acceleration or deceleration process, the following error of a position controlled axis remains constant.Load changes of the drive (e.g. due to interrupted or heavier cuts), are controlled by the velocity control (PI behavior).At constant command velocity, changes in following error occur only when the velocity control reaches a limit due to drive overloads, e.g. if tools break.This change is used as criterion for triggering the contour monitoring.A tolerance range is established for the maximum allowed following error, in order to avoid triggering the contour monitor unnecessarily following slight speed changes, as would be caused e.g. by motor slot ripples.There also is a dwell time after any velocity change, which must elapse before the contour monitoring can be activated.The width of the tolerance range and this delay interval are inversely proportional to the position loop gain. Accurate contouring processes require that all axes which participate in the interpolation motions have the same position loop gain setting.Besides being set as machine data in the NC TEST nr. 150.. = Kv and TEST nr. 220.. = MULT-GAIN,the position loop gain is also determined by the tacho adjustment of the speed control,the actual value mult-gain factors ,the gear ranges,etc.

For this reason, the contour monitor is provided with a Kv computation. The position loop gain is calculated from the command velocity and the resulting following error. This calculation is executed at the velocity ( ± 25 %) set under TEST nr. 371 (manual feed rate). The corresponding axis must hereby run at constant velocity for at least 3 seconds. The computed Kv value is displayed under TEST nr. 850.. ,in units of m/min (velocity) • 1000.This mm (following error) dimension is known and commonly used by machine tool manufacturers ; usual values are between 500 and 1800. The calculated Kv value is cleared every time a machine datum is changed.After the Kv values have been calculated for all axes concerned, their equality will be checked. If the deviation is larger than 50,alarm 527 (Unequal Kv factors) will be triggered. This alarm is also displayed if any machine datum is changed. The machine can thus be operated without alarm display. (There are machines, which have to operate with unequal Kv factors, e.g. for rotary axes.) After power-up or machine data changes, ALARM 528 will indicate if any Kv factors have not been computed. The once determined Kv factors will remain for as long as no machine data are changed

Machine data TEST nr. 351 and TEST nr. 352 can be used to modify the contour monitoring.

The velocity at which the contour monitoring becomes active is entered in mm/min under TEST nr. 351. At axis standstill, the contour monitor will not be active even after an input of  $\beta$  .The standstill monitor checks in such cases for inadmissable axis movements.

The tolerance range for allowed contour deviations is entered under TEST nr. 352. This process also takes into account the calculated Kv factor, so that the tolerance band is derived according to the following formula :

 $\frac{\text{TEST nr. 352} \cdot 125}{\text{Kv} \cdot 1000} = \text{Tolerance band} (\text{um})$ 

Value 2000 is automatically entered with input Ø.Thus, for Kv = 1 for example, the resulting tolerance band will be 250  $\mu$ m, for Kv = 2, 125  $\mu$ m, etc.

The actual contour deviations can be displayed with TEST nr. 840....

If the position command value is changed, the contour monitor becomes inactive. This renders any contour monitoring inactive during circular interpolations. In order to protect the machine even in these cases, the signs of the following error, position command value and position actual value are continuously compared with each other. After the elapse of the Kv dependent safety time, if disparities have been found, the contour alarm (alarm 506) will be triggered.

#### 6.8 Analogue Spindle Speed

The output for the analogue spindle speed command value is on the position 07 -325, -350 control board 03 320 (This value can be displayed under TEST nr. 860 ;the unit is VELO (2048 velo = 10 V).



The conversion of the command value, taking into consideration the gear range, is done by the control with the use of the maximum valocity, which is entered under TEST nr. 359 - nr. 364. This input value (in rpm) must always correspond to the same motor speed (and thus to the same command value voltage U). For the set maximum spindle speed, when programing this speed and gear range, the control outputs 2048 units (approximately 10 V).

NOTE : The gear range must be acknowledged by the interface.

					Gear range		TES	TEST nr.		8 Byte 4	
TEST	54			Gear ran			Input		Signals		
	-4-					4	С	В	A		
Nr.	359	S	1000	2	1		0	0	0		
Nr.	360	S	2000		2		0	0	1		
Nr.	361	S	3000		3		0	1	0		
Nr.	362	S	3000		4		0	1	્રે		
Nr.	363	s	3000		5		1	0	0		
Nr.	364	s	3000		6		1	0	1		
Nr.	<b>36</b> 5	S	3000		7		108	1	0		
Nr.	366	S	3000		8		1	1	1		

Example : 3 gear ranges

For only 3 gear ranges, S  $\not D$  can be entered for nr. 362 - 366; the codes for gear ranges 4 through 8 may hereby <u>not</u> be used for input signals.

## TEST Nr. 377 Minimum Motor Speed

TEST nr. 377 determines the lowest admissable speed of the motor, by limiting output voltage in VELO.A drift of the speed control can be compensated for through the input of a compensation value under TEST nr. 357. This is done by commanding a low speed ; the speed must be the same in both directions.

# 6.9 Finishing Touches

## 6.9.1 Function Tests with NC Test Program

For testing the following functions : Actual value display Data display All S, T, M-functions Single block, deletable blocks, program-stop Program memory Tool offsets Thread Data input Data output

The program and the tape should be produced by the machine manufacturer.

6-15

## 6.9.2 Generating a Machine Data Tape

There are several possibilities : 🔿

- a) A tape of the data entered during the start-up can be created out of the TEST memory, using a tape punch. For the operation sequence, see the service manual, section 2.
- b) A tape can also be generated on a separate programing device.

The tape should be placed by the control. The printout of the tape, or the list filled out according to section 1.2, should be filed into the log book.

If subsequently changes are made in machine data, a new tape and printout must be produced. After the start-up is finished, the service switch should be returned to its normal position (S3 on the front plate of the CPU, <u>down</u> = inactive position), in order to prevent inadvertent machine data changes.

The loading of standard machine data, change of machine data, and the handling of machine data tapes are described in sections 4.4 through 4.7.

#### 6.9.3 Machine Data Tape with Machine Standard Data

A machine data tape must be created for each machine; such a tape could then be used as standard tape for other machines of the same type. Individual data,e.g. drift values, grid point shifts, must however be derived and entered for each particular machine.

A "data gathering" scheme can be found on the first pages of the service manual, in section 1.

This scheme also includes a set of short explanations; more extensive information is given in section 11 (Machine data description).

For machine data tape inputs :

SEE SECTION 4.6

A new drift compensation must be made (according to section 4.6.4) after loading the machine data tape.

#### 6.9.4 Brief Instructions to the Customer

The operators and maintenance personnel who will be dealing with the numerical control should be given as much information as is possible during a start-up procedure.

The following will remain to be executed by the customer :

- a) the reference point adjustment ,
- b) determination of the backlash ,
- c) entering these derived values into the machine data memory, as well as the punching of a corresponding tape (respectively insertion into the already existing machine data tape),
- d) drift compensation (see section 4.6.4).

The customer's personnel must be instructed concerning the input of these data, in order to make sure that things can be handled in the absence of service engineers.

#### 6.9.5 Start-up Report

The service call must be filed in the log book.

The customer should confirm the completion of the start-up and the functionality of the control ; this acknowledgement should be included in the start-up report (form).

#### 6.9.6 Check-list for the Log Book

Include the check-list, completely filled out according to section 1.1, in the log book.

# Machine Data Description

# Content

7

- 7.1 Generalities
- 7.2 Axis-specific Machine Data
- 7.3 Common Machine Data
- 7.4 Description of Machine Data Bits


#### 7.1 Generalities

The machine data are entered into the TEST data memory.Input is possible only with the data safety switch S1 enabled.In versions 0 - 2, display is possible in all positions except D0/DI.

Overview :

100 - 223	Axis-specific data for programed axes
350 - 393	Data common for all axes,respectively for spindle and such
400 -419	Machine data bits
	o, o

# 7.2 Axis-specific Machine Data

Data memory TEST

Axis number	Ident- number	Addr.	Sigr	,	Mail.	Displ	.ay/Input	Ś.	ster.
1 2 3 4	100 101 102 103	S		Juto	Pos (e>	ition act st	tolerance op limit	)	€
144 144 16		Posit cont	ion rol ts	Inp	out li	mits	Incre- ments	Ur	nits
Metric; do	egrees	1/2 • 1	0 ⁻³ mm	Ø	3 3	32 000	1500	1 µm; 1	0 ⁻³ deg.
Inch	JOR JOC	$\frac{1}{2}$ • 1	0 ⁻⁴ mm	Ø	3	32 000	Sall ^C 1	10	4 in

7-3

The position is considered reached, when the axis reaches the command position ± the set position tolerance (position approach).

Effects of monitoring :

If the command position is not reached within these limits, the position control lamp remains on, and further motion commands are inhibited. Remedy : drift compensation.

Example :

N100 S10



# Note :

The position tolerance limit is only checked at GO9, GOO, G6O, and single block.If it is not reached (under TEST 800...803S : check following error), the NC stops.

Axis number	Ident- number	Addr	Sign	Ca Ra	Displ	ay/Input		and a d
1 2 3 4	110 111 112 113	S	ANICO BALLO	C p a	Clamping tolerance and position control monitoring at standstill		$\mathbf{\mathbf{\widehat{s}}}$	
4	*3	Posi cor ur	ition ntrol nits	Inp	out limīts	Incre- ments	Uni	ts
Metric;	degrees	$\frac{1}{2}$ • '	10 ⁻³ mm	Ø	32 000	î	1 µum;	10 ⁻³
In	ch	$\frac{1}{2}$ •	10 ⁻⁴ mm	Ø	32 000	1	10	4 in

The NC monitors the position at standstill (holding the position).

The following possibilities exist :

- a) If the interface control inhibits the control enable of an axis, it means that the NC does no longer hold that axis in position. The interface must hold the axis in position itself, through clamping. In such cases, the axis can be mechanically pushed out of position.
- b) The axis can be pushed out of position following high mechanical forces or drive malfunctions.

The clamping tolerance must be set <u>higher than the position tolerance</u>. After the dwell time of the position monitor TEST nr. 353, if the clamping tolerance value is exceeded, alarms 101, 111, 121, 131 are displayed. If the alarm is triggered in the last block, it will be cleared immediately with M30.

Axis number	Ident- number	Addr	Sign	N. C.	Displ	ay/Input		Catolka.pl
1 2 3 4	120 121 122 123	S	AMIGDOUC	and a second	Accelerati	on facto	r www.	$\mathbf{\mathbf{x}}$
4	8	Posi cor	ition ntrol nits	Inp	out limits	Incre- ments	Uni	ts
Metric;	degrees	1 <u>2</u> • ·	10 ⁻³ mm	ø	6 000	. And the second	10 ⁻² m/s ²	;10 <u>deq</u>
In	ch	$\frac{1}{2}$ •	10 ⁻⁴ mm	ø	2 400	1	1 inch	n/s²

The acceleration factor is set independently for each axis. The values also apply to deceleration (for braking, see section 7.3). The axes need not have the same acceleration values. In contour operations, the control always deals with the lowest acceleration value of the participating axes.

Remark : Values between 50...100 ( =  $0.5...1 \text{ m/s}^2$ ) are common.

These values are inactive at thread ;the active value here is 3585.

Axis number	Ident- number	Addr	Sign		Displ	ay/Input	N.	à
1	130		- aut	ST. CO	Maximum v	elocity	~	Monable
2	131	5 <	WWW.CY				ANNAN! OF	
3	132	à						
4	133			SUGAR				tornative
ANA AND	2 ⁰	Posi cor ur	tion trol nits	Inp	out limīts	Incre- ments	Uni	ts
Metric; (	degrees	ે <del>ટ</del> ∙ 1	0- ³ mm	ø	15 000	1	1mm/min;	1deg/min
Ind	ch _{soft} ar	<u></u> 1⁄2 ∙ 1	0 ⁻⁴ mm	ø	6 000	109	0.1 ir	n/min

The entered value represents the limit velocity to which the axis can accelerate (<u>rapid traverse limit</u>). The axis moves with this velocity when programed with rapid traverse GOO.

Example :

Maximum velocity :

cy:	X axis	12 m/min
	Y axis	12 m/min
	Z axis	10 m/min
	4th axis	4 m/min

If the machine is programed with 10 m/min by program, the axes will move as follows:

X axis	10 m/min		
Y axis	10 m/min		10 ^{.2}
Z axis	10 m/min		at limit nr. 132
4th axis	°4 m∕min	- Walton	with limit nr. 133

Example : Y and Z axes under 45°, with programed rapid traverse (15 m/min) Both axes move with 10 m/min,which corresponds to 14.142 m/min contour velocity, because the Z axis has been limited to 10 m/min under nr. 132.

Axis number	Ident- number	) Addr.	Sign	Display / Input	Calykan
1 4	140		ANI DOUD	Velocity command limit	5°
2	141				
3	142	25	+	Rank Markan	
4	143		W.Goalto.	MIGDOLLO. MIGDO	510.

Position control loop board	Input Limits		Incre- ments	Units
03320	0	2 048	dil 1 mon	1 VELO 1= <u>10 V</u>
03325/03350	D	8 192 🔹	1	1 VELO 1= <u>10 V</u> 8192

Note : Exceeding this limit results in IPO stop; the drive oscillates.

This input defines the maximum voltage value which can be produced as velocity command value (output voltage limitation through interpolation stop).

This voltage value should lie approximately 10% above the voltage for maximum velocity, so that overshoots can be controlled. For 9 V velocity command value for rapid traverse, the value should thus be 2048 or 8192 (for 10 V corresponding to 10% control reserve, also see section 7.3, machine datum 354).

Axis number	Ident- number	Addr.	Sign	Display / Input	alfonio
1	150		and and a second	Position loop gain	
2	151	2		Here Black	
3	152	5	+ 101	tosuton"	
4	∍ 153		anner!!!	Kv factor	

				•
Ιηρι	ut Limits	Incre- ments	Units	ð
D	10 000	C. 1	0.01 sec-1	

#### Conversions :

 $Kv (0.01 \text{ s}^{-1}) = 1666 \cdot Kv (\frac{m/\min}{mm})$ 

Or

 $Kv (0.01 \text{ s}^{-1}) = 1666 \cdot Kv (\frac{mm/min}{um})$ 

The position loop gain is axis-specific.

The values entered for axes which do not participate to contour operations may be different than those for axes that do participate in such processes Axes which cooperate in contour operations must have the same Kv factor (equal following error at equal velocity =  $45^{\circ}$ ).

Data Memory TEST

Axis number	Ident- number	Addr	Sign	10.0	Displ	ay/Input		.Hail
1 2 3 4	160 161 162 163	S	t	5	Goftware li	mit swit	ch <u>plus</u>	$\mathbf{\mathbf{\widehat{S}}}$
44 19 10		Posi cor ur	tion ntrol nits	Int	out limīts	Incre- ments	Uni	.ts
Metric;	degrees	$\frac{1}{2}$ • 4	10- ³ mm	Ø	<del>99999999</del> +)	1	1 µm; 10	- ³ degrees
In	ch	$\frac{1}{2}$ • •	10 ⁻⁴ mm	Ø	<del>99999999</del> ⁺⁾	1	10	⁴ in

Remark: +) input limit for versions 0 and 2, 9 999 999

The software limit switch can supplement the common limit switch. The absolute position of the positive range of each axis must be entered. The software limit switch becomes active only after reference point approach.

When the positive software limit switch is reached, alarms 1, 11, 21, 31 are displayed.

#### Note :

There are no input signals for hardware limit switches. These can only act through :

- Feed hold (unsuitable due to acceleration ramp)
- Drive inhibit (most advantageous due to speed stop via jump functions)
- Emergency stop (fast with jump functions, but unsuitable due to side effects)

Software limit switches are overrun despite the automatic reduction (see section 7.4). Overrun dependent on approach speed.

Axis number	Ident- number	Addr	Sign	32	Displ		antan		
1 2 3 4	170 171 172 173	s.	shaulon ±	342	Software lin	nit switc <u>mi</u>	:h . <u>nus</u>	$\mathbf{\mathbf{\hat{s}}}$	
14	2	Posi con un	tion trol its	II	nput limīts	Incre- ments	Uni	ts	
Metric; degrees $\frac{1}{2} \cdot 10^{-3}$ mm Ø 9		<del>99999999</del> +)	<u>_</u> 1 ·	1 um; 10 ⁻³ degrees					
In	ch	1 2 • 1	0 ⁻⁴ mm	ø	<del>99999999</del> +)	1	10 ⁻⁴ in		

Remark : +) input limits for versions 0 and 2, 9 999 999

The software limit switch can supplement the usual limit switch. The absolute position of the negative range limit of each axis must be entered. The software limit switch is activated ony after reference point approach. Alarms 2, 12, 22, 32 (depending on axis) appear when the negative software limit switch is reached.

#### Note :

The software limit switch becomes active only after the reference point of the corresponding axis has been approached.

Axis number	Ident- number	Addr	Sign	K ^a l	Displ		Cardward C.	
1 2 3 4	180 181 182 183	S	tosuona"	F	Reference p	oint coo	rdinates	$\mathbf{k}$
		Posi cor ur	tion trol nits	Inf	out limits	Incre- ments	Uni	ts
Metric; degrees $\frac{1}{2} \cdot 10^{-3}$ mm Ø 999999999 +) 1		ິ <u>1</u>	1 um;10 ⁻³ degrees					
In	ch	1 · 1	0 ⁻⁴ mm	ø 999999999 +) 1 10 ⁻⁴ i			4 in	

Remark: input limits for versions 0 and 2, 9 999 999

The difference between the absolute machine zero point and the fixed reference point is set for each axis. These values are entered as actual values at reference point approach.

Axis number	Ident- number	Addr	Sign	Č,	Displ	ay/Input	2	Call A
1 2 3 4	190 191 192 193	S	++++++++++++++++++++++++++++++++++++++	JC III	Backlash co	mpensati	on and	€¥
di Li		Posi cor ur	tion ntrol nits	Inp	out limīts	Incre- ments	Uni	ts
Metric;	; degrees $\frac{1}{2}$ · 10 ⁻³ mm Ø 255 1 1 um; 10 ⁻		³ degrees					
In	ch	1 2 • 1	0 ⁻⁴ mm	nm Ø 255 1 10 ⁻⁴		4 in		

Backlash can be positive or negative; for this reason, a value of up to ± 255 um is entered for each axis. The value must be positive for positive backlash, and negative for negative backlash.

Positive Backlash (normal case)

<u>Negative Backlash</u>

, Table Backlash Encoder

The encoder actual value preceeds the real actual value of the table



The real actual value of the .table preceeds the actual value of the encoder.

Data Memory TEST

Axis number	Ident- number	Addr	Sign	Sign Display/Input				
1 2	200 201	S	undballon	8	Tool refér	ence poir	nt dhai	
3	202		-	N.				
4	203			S.		onab		onan
Arana.C		Posi cor ur	tion ntrol nits	In	out limīts	Incre- ments	Uni	ts
Metric; degrees		1 · 1	10 ⁻³ mm		<del>99999999</del> +)	APX .	1 um; 10 ⁻³ degrees	
Inch		$\frac{1}{2}$ • 1	0 ⁻⁴ mm	ø	999999999 +)	<u>۲</u>	10	-4 in

Remark:+)input limits for versions 0 and 2, 9 999 999

For the automatic determination of the tool geometry, see the operating manual, section 8.1.7 .

(Automatic tool offset determination available only for 3T with option J12)

Axis number	Ident- number	Addr	Sign		Displ	.ay/Input	12	Ŕ
1 2 3 4	210 211 212 213	5	**************************************	R	eference p	point shi	ft ward	$\mathbf{E}$
ANNALC .		Posi cor ur	tion ntrol nits	Inp	ut limīts	Incre- ments	Uni	ts
Metric;	degrees	2 • 1	0 ⁻³ mm	Ø	9 999	1 30 2	1 µm; 10	) ⁻³ deg
In	ch	1 2 • 1	10 ⁻⁴ mm	Ø 9 999 1 10		4 in		

The reference point of the position control system can be shifted with the reference point shift. Thus, instead of shifting the position coder mechanically (hence also the *DECELERATION cam), the reference point can be shifted electrically up to  $\pm$  9999  $\mu$ m.

Positive reference point approach direction : If the input is positive, the axis moves beyond the reference point in positive direction (2000 µm after zero pulse).

Negative reference point approach direction :

If the input is negative, after approaching the zero pulse, the axis moves by the value resulting from the difference of 2000  $\mu$ m + input value. For reference point shifts larger than ca. -2000  $\mu$ m, after zero pulse approach the software recognizes that motion is in the wrong direction, and reverses it.



Reference point approach is possible even if the cam is on the deceleration switch.

Axis number	Ident- number	Addr.	Sign	Display / Input	Catha.P
1,8	220		ALODOULC	Multiplication factor for	
2	221	14		the position loop gain	
3	222	S	+	.e	
4	223		N.GDOULOT	WH. Charles	<u>0</u> *

Іпрі	ut Limits	Incre- ments	Units
1,50	32 000	101	<u>3 • 10</u> 7 Vinex <mm min=""></mm>
1	32 000	1	3 • 10 ⁷
			Vmex 0.1 in/min

For accuracy, and because of the different conversion factors, this value must be entered as follows :

 $MULT-GAIN_{input} = \frac{3 \cdot 10^7}{V \max \frac{mm}{\min}} \cdot \frac{Umax [V]}{10 [V]} , respectively$ 

V_{max} = the maximum axis velocity as given under nr. 130 - 133,is entered as maximum velocity

 $U_{max}$  = command value voltage for  $V_{max}$  (tacho adjustment)

Example :

 $V_{max} = 10\ 000\ mm/min;\ U_{max} = 9\ V$ MULT-GAIN =  $\frac{3 \cdot 10^7}{10000\ mm/min} = 2700\ [min/mm]$ 

If the MULT-GAIN factors are entered in the described manner, the Kv factor set under N150-N153 corresponds to the value active on the machine, in the appropriate units.

#### MULT-GAIN Table - different input values

Vma	× solution solution			
m		1	9 V	U _{max}
min	4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4			
à15	8	1600	1800	
2 14	No.x	1714	1929	
13	18 18 18 18 18 18 18 18 18 18 18 18 18 1	1846	2077	
12	10, 10, 10, 10, 10, 10, 10, 10, 10, 10,	2000	2250 ြ	
1 11	Service Service	2182	2456	
10	l S. L S.	2400	2700	
9	3	2667	3000	
8	1.	3000	3375	
7		3429	3857	
8 6		4000	4500	
5	100	4800	5400	
4	- Stree	6000	6750	
3	alle alle	8000	9000	
2	S. S.	12000	13500 🚫	
1 1	12000	24000	27000	
0.8 🚿	15000	30000	32000	
0.75	16000	32000		
0,6	20000	3		
0.5	24000	Ko. Ko.		
0.4	30000		d	

#### Inch System :

inch/min	9 V
600	4500
500	5400
400 🔜	6750
<b>30</b> 0 🚫	9000
200	13500
100 🛸	27000

A> nun	kis Nber	Ident- number	Addr	Sig	n .	Disp	1		
and the second	1 2 3 4	230 231 232 233	S		2	Drif	$\mathbf{E}$		
L	Pos loo	ition co p boards	ontrol	Inpu	t Li	lmits	Incre- ments	Units	AL.
	03	320		0		500	doa ⁵¹	VELO 1= <u>10 V</u> 2048	
	03	325/0335	50	0	2	000	1	VELO 1 = $\frac{10.V}{8192}$	4

To eliminate analog drift values software-wise, it is possible through MDI, to bring the following error at standstill to zero. It is also possible to do an automatical drift compensation in the operation mode MDI-TE-SE and the TEST mode under nr. 230, using **S A** . The compensation must be done for each axis individually.

If the values becomes larger than 100 (03320),respectively 400 (03325/03350) during automatic drift compensation,alarms 105, 115, 125 or 135 are displayed (see section 4.) .

#### 7.3 Common Machine Data

Jata Mer	pory_	ES I	×			Ke.		J.C.X		10×
Axis number	Iden: numbi	t- er	Addr	Sig	, north	Display/Input				STOWAR'S
And A	350	) Scht	S	Annal +		Cut-off velocity				$\mathbf{E}$
Starting Charles			Posit contr units	Position control units		Input limi		Incre- ments	Uni	ts
Metric		1/2	. 10-	.3 _{mm}	Ø	Ø 15 000 1 1mm/min,			,1º/min	
Inch		1/2	• 10	-4in	Z	K.	6 000	000 1 0.1 inch/m		ch/min

The velocity entered with 350S is active at :

- a) Positioning from program at cut-off feed rate.
  The velocity selected for approaching the reference point and for cut-off from a higher velocity should allow proper positioning.
- b) Reference point approach, as long as the "Deceleration at reference point" signal is active.

#### 351 S: Threshold Feed Rate for Contour Monitor

351 S	Position Control Units	Input Limits	Incre- ments	Units
Metric, degrees	12 • 10 ⁻³ mm	0 -> 15 000	1	mm/min
Inches	<u>1</u> • 10 ^{−4} in	0 - 6 000	<u>1</u>	0.1 in/min

352 S: Tolerance Range for Contour Monitor

352 S	Position Control Units	Input Limits	Incre- ments	Units	
Metric, degrees	$\frac{1}{2}$ • 10 ⁻³ mm	032 000	r ^{an} 1	mm • TEST 850 125 • 1000	
Inch	½ • 10 ^{−4} in	032 000	1	0.1 in • TEST 850 125 • 1000	

The Kv value is determined (display TEST 850...853 in 0.01  $\frac{m/\min}{mm}$ , normal values between 500 and 1800) only after at least 3 seconds of constant velocity. This remains stored until new machine data are entered. After the Kv has been determined for all axes (alarm 528 is otherwise triggered), an equality check is made. Deviations larger than 50 trigger alarm 527.

Tolerance band derivation :  $\frac{\text{TEST N352} \cdot 125}{K_V \cdot 1000}$  [µm]

For further description of the contour monitoring, see section 6.7 .

Axis number	Ident- number	Addr.	Sign	Displa	Display / Input				
wanter of	353	S	enterballoone	Dwell monito	Dwell time for position monitoring				
<u>Note</u> :		01 41	Input I	Limits	Incre- ments	Units			
Standard value : 500		:	D _{so} rté	16 000	1 10 Martin	1 ms	AS COSTINOT		

The entered dwell time is active at :

- The clamping limit (nr. 110 113) becomes active during position approach (digital zero), only after the elapse of this dwell time. The interval selected must allow the largest following error to be reduced, without triggering alarms 101, 111, 121, 131.
- 2. Delay time for the output of the control inhibit signal, after E-Stop and other faults which lead to an immediate stop of the axes motion.
- 3. Delay time for the output of the control inhibit signal, for cases in which the interface revokes the control enable of a moving axis.
- 4. Delay time for alarms 101...131 (standstill monitors), in cases where the maximum velocity command value has been exceeded (nr. 141 143).

Axis number	Ident- number	Addr.	Sign	Display / Input	
20, 10,	354	S		Velocity command value limit monitor (position control or drive fault)	$\mathbf{E}$

Position control loop boards	Input	t Limits	Incre- ments	Units	
03320	0	3 000	1	1 VELO 1= <u>10 V</u> 2048	
03325/03350	0	12 000	1300 ⁰⁰⁰	1 vello 1= <u>10 v</u> 8192	

If the velocity command value generated is too high (position control or drive malfunction), this monitor triggers alarms 102, 112, 122, 132. The input value must be higher than the largest value set under nr. 140 - 143 as maximal velocity command value.

Recommendation : approximately 20 % higher



Axis number	Ident- number	Addr	Sign		Display/Input			
10. 10.	355	5 Steric	and the second	C	ircle end-	nitor		
		Position control units		Inp	out limits	Incre- ments	Uni	ts
Metric; degrees $\frac{1}{2}$ • 10		0- ³ mm	ø	32 000	1	1 um, 1	0 ⁻³ degrees	
In	Inch $\frac{1}{2} \cdot 10^{-4}$ mm		ø	32 000	+ 1	10	-4 in 🔬	

The input value determines a ring (tolerance ring) equidistant to the <u>programed</u> <u>circular arc</u>, independently of the programed end-point. If the programed end-point lies within the tolerance ring, the machine will move along the incorrectly programed contour until the end-point can be radially approached. If the programed end-point falls outside the tolerance ring, this will be already recognized within the buffer, the block will not be released for machining, and alarm 308 will be displayed.

The same holds true if the radius is properly programed (M=M'), but the end-point is programed incorrectly.

Not to scale : e.q. 🗙 = 10 µm М End-point outside tolerance ring limits End-point within .... required contour (with M & E) tolerance ring : tolerance ring: actual contour (with wrong M') no machining, machining, no alarm alarm 308 E=end-point (correct) A=start-point M=required center point d=input value M' = programed center point (incorrectly programed parameter or R)

Axis number	Ident- number	Addr	Si	gn	Display/Input					
	356	S	+	autome	Threshold movements compensatio	nsatory radius	$\overline{\mathbf{S}}$			
	N.	Posit contr units	ion ol	Inp	ut limits	Incre- ments	Uni	ts ded		
Metric;		ø	32 000	1	1 um, 1	0 ⁻³ degrees				
Inch	Inch		ø	32 000	1	10-4	inch			

For transitions from circular contours to linear contours or to further circular contours,1 or several intermediate blocks are inserted (see programing instructions) for linear compensatory motion(s).During these compensation movements, the programed feed rate for the machining of the work piece contour is maintained on the cutter radius center point.This results in feed rate differences.In order to prevent feed reductions over very small distances, the compensatory movements below threshold "d" should be minimized or omitted as follows :



For transitions with only linear interpolation, the threshold is inactive.Compensation movements are executed without exception (see the programing instructions). ----- Cutter center point ////// Contour deviation **∆**X1, **∆**Y1 **<** d:



Reduced compensatory motion A-C



No compensation.Proper compensation is only reached at the end (E) of the block.

7-23

Axis number	Ident- number	Addr	Sign		<u>à</u> c	)isplay/I		10 ⁰
Spindle	357	5	Gloaut	Clore .	Dr sp	ift compe indle	$\mathbf{\widehat{S}}$	
Output edition			Input limits			Incre- ments	Units .	
3T/MD-2 software 04, 05 3T/M3 software 01			0	0 250		1	$VELO 2 = \frac{10 V}{2048}$	
3T/MD-2 software 06, 07 3T/M3 software 02, 03			0		500	1	VELO 2 =	<u>10 V</u> 8192

This machine datum determines the drift compensation value for analog spindle speed output.

At small command values, this value must be changed in the respective direction, positive or negative, until the spindle's actual speed is equal for both turning directions.

#### 358 S Dynamic Smoothing Exponent for Thread

Input	: Limits	Increments	Units
0	5	1	(2 ^x -1) up-date time

This affects the feed drive's ramp time at thread cutting for the following purposes : - to achieve short lead-in distances for thread cutting,

- to compensate for the ramp time of the spindle.

The time base for this is the actual value up-date time, according to the following equation :

(2^x-1) • up-date time

; (x = input value)

Input value	0	1	2	3	4	5
Up-date time mult.	0	×1	3	7	15	31
Ramp function	Jump		R	am	P	

Axis number	Ident- number	Addr	Siç	JU	Display/!	Input		matykala
Spindle	359 360 361 : 366	S	Indon-	X	Maximum sp 8 gears			
Spe	Speed valuation			Inp	out limits	Incre- ments	Units	
1 - 9	1 - 9999 mm ⁻¹			16	9 999	1	min ⁻¹	
0.1 - 999 m ⁻¹				16	<u>)</u> 9 999	10	0.1	m ⁻¹

The machine data determine the individual spindle speeds of each of the 8 gears at 10 V command value. If no gears are present, the maximum allowed spindle speed is entered under 359, and  $\emptyset$  is set under 360...366. If fewer than 8 gears are present, set  $\emptyset$  where no gear value exists.



Designation:

Gear range	1.2	2	3	4	5	6	7	8	
Input number	359	360	361	362	363	364	365	366	Ì

For the input signals of gear ranges, see section 8. (Interface test input signals, 3 inputs for gear range code)

Axis number	Ident- number	Addr	Sig	gn	e de	)isplay/In		ather			
wanter of	371	371 S +				Manual feed					
-automathe		Posit contr units	Inp	ut	limits	Incre- ments	Units .				
Metric;	degrees	$\frac{1}{2} \cdot 10^{-3}$ mm		Ø		15 000	1	1 mm/min;deg./m			
Inch		<u>1</u> •10−	⁴ in g			6 000	1	0.1 in/	min		

The input value determines the manual feed rate for all axes, unless the value is limited through the input under Test nr. 130 - 133, at 100% feed rate override.

Axis number	Ident- number	Addr	Sign		Displ	.ay/Input		615H8.P
AND	372	S	undbaute		Manual rapi	d traver	se rate	$\mathbf{E}$
44		Posi cor	tion ntrol nits	In	put limits	Incre- ments	Uni	ts
Metric; (	degrees	$\frac{1}{2}$ • 1	0- ³ mm	Ø	15 000	্রী	1 mm/min	;deg/min
In	ch	1/2 • 1	0 ⁻⁴ in	ø	6 000	1	0.1	in/min

Unless limited through the input under Test nr. 130 - 133, the value entered determines the manual rapid traverse velocity for all axes, at 100% rapid traverse rate override. This value is not used with programed rapid traverse GOO.

The programed rapid traverse GOO is determined by the maximum velocity set under Test nr. 130 - 133.

Axis number	Ident- number	Addr	Sign	and	Displ	ay/Input		(addra.g)
and a second and a second and a second and a second a s	373	S	endballon	F	Reference p velocity	oint app	roach	Ð
2	J.	Posi cor ur	ition ntrol nits	Inp	out limits	Incre- ments	Uni	ts
Metric;	degrees	$\frac{1}{2}$ • $\frac{1}{2}$	10 ⁻³ mm	Ø	15 000	5 No.	1 mm/mir	;deg/min
In	ch	$\frac{1}{2}$ • 4	10 ⁻⁴ in	ø	6 000	1	0.1 i	.n/min

Unless limited through the input under Test nr. 130 - 133, the value entered is valid for all axes, at 100% feed rate override and rapid traverse override ON.

Axis number	Ident- number	Addr	Sign	201 201	Displ	.ay/Input		natyka.p
AND	374	S	+ +	Clark P	Incrementin	g feed ra	ate	$\mathbf{E}$
79	2	Posi con un	tion trol its	In	put limits	Incre- ments	Uni	ts
Metric; o	legrees	1/2 • 1	0 ⁻³ mm	ø	15 000	J. C 1	1 mm/min	;deg/min
Inc	:h	1 2 • 1	0-4	Ø	6 000	1	0.1 in	/min

The entered velocity is active only during "increment" mode. Resonable input values: up to  $1000 \frac{mm}{min}$ .

Axis number	Ident- number	Addr	Sign	R.	Displ	ay/Input	Ś	al Marin
annan Goolf	375	S	spautoman	4. 	Dry run fee	d rate	www.grantos	$\mathbf{k}$
	20142.01	Posi con un	tion trol its	r Ir	nput limits	Incre- ments	Uni	ts
Metric;	degrees	1/2 • 1	0- ³ mm	ø	15 000	1	1 mm/mir	;deg/min
In	ch	<u></u> 1⁄2 • 1	0-4	ø	6 000	1	0.1 ir	n/min

Unless limited axis-specifically by the input under nr. 130 - 133, the entered value is activated with the dry run mode switch, and replaces the programed feed rate.

The feed rate override switch is active.

Whether or not the dry run switch is interlocked with the key switch depends on machine data bit nr. 410, bit 2.

Axis number	Ident- number	Addr.	Sign	Display / Input	Sautomats
Spindle	376	S	Marth.	Dwell time for spindle inhibit *)	$\overline{\mathfrak{S}}$

Inpu	t Limits	Incre- ments	Units	Ş
Ø	16 000	ANNO 1	ms	

After this dwell time has elapsed, a spindle command value of  $\not D$  revokes the control enable (* control inhibit), which prevents the spindle from creeping.

This dwell time is active at :

- removal of the spindle enable signal ,
- MØ5
- E-stop
- activation of the position control monitor

Axis number	Ident- number	Addr.	Sign	Display / Input	.onatyka.t
Spindle	377	s	Anna Goaton	Minimum spindle motor speed	$\mathbf{\widehat{X}}$

Edition	Inpu	ut Limits	Incre- ments	Units
3T/MO-2 ed.04&05 3T/M3 software 01	0,000	2 048	and source	1 VELO 2= <u>10 V</u> 2048
3T/MD-2 ed.06÷08 3T/M3 ed. 02÷04	0	8 192	1	1 VELO 2= <u>10 V</u> 8192

This machine datum determines the minimum motor speed, below which the spindle should not go,e.g. at constant surface feed and increasing turning diameter. This means that from this point on, the surface speed is no longer constant, but increases with the turning diameter. The motor can run smoothly down to this speed.

#### Example :

Motor maximum speed = 3500 rpm, corresponds to the maximum spindle speed. Motor minimum speed = (e.g.) 50 rpm

Input value :  $\frac{50 \text{ rpm}}{3500 \text{ rpm}} \cdot 8192 = 120$ 

Axis number	Ident- number	Addr.	Sign	Display / Input	automaty
Spindle	378	S	ANNAN COLLOC	Cut-off spindle speed for M19	$\overline{\mathbf{S}}$

Valuations	Inpu	ut Limits	Incre- ments	Units
1 - 9999 rpm	0	9999	1	1 min ⁻¹
0.1 - 999 rpm	O	9999	1	0.1 m ⁻¹

This machine datum determines the spindle speed to which the spindle speed (M19) is reduced for spindle positioning, and with which the spindle moves until it is positioned with the set position control characteristic curve (see MD 379). Only for 3T/M, starting with software edition 02 and option E42.

Axis number	Ident- number	Addr.	Sign	Display / Input	S CTURE
Spindle	379	S	M.Obaltonat	Gain factor for the position control loop (M19)	$\mathbf{E}$

Speed Value MD 407 Bit 3	Input Limits		Incre- ments	Units
	0	10000	6 ^{41.0} 1	<u>1/min</u> 360°
1	0	10000	1	0.1 <u>min⁻¹</u>

Recommended value: 50 to 500 - dependent upon performance of drive and speed controlles.

In oriented spindle stops (M19), the spindle is in closed position control loop. The gain factor is described by the positioning slope to the cut-off position. The slope is defined as the spindle speed (in rpm) at a position deviation of 360°.

Only for 3T/M 3, starting with software edition D2 and option E 42.

Axis number	Ident- number	) Addr	Sign			Display/I	anatha a		
Spindle	380	S	10 ⁴	+ Position limit f		imit for f	or M19		
Š.	pattomart		, S	Inp	out	limits	Incre- ments	Uni	ts
			S. S.	0		1000	1	1/11 d€	egree

The position limit is entered in increments of the spindle encoder. One (1) increment represents 360/4096 degreees.

In oriented spindle stops (M19), the "POSITION APPROACHED" flag is outputted to the PC as soon as the position deviation is within these limits. Only for 3T/M 3, starting with software edition O2 and option E 42.

Axis number	Axis Ident- number number Addr.		Sign	Display/Input	to natyko
tra.	381	S	Ann+100	Software edition	
2	² Q	140.01		ashead	*)

The software edition is written by the manufacturer into the EPROM, and is transferred to 381S with Power-On-Reset. (Input limits: 0... 32 000)

*)Values entered inadvertently can always be overwritten, and replaced with the value set in the PROM, with Power-On-Reset.

Axis number	Ident- number		Sign	Display/Input				
9. 44	383.	S	+	Increase up-date time			$\bigotimes$	
	astari		Ιηρι	ut limits	Incre- ments	Un	Units	
			0 30		1 <u>1</u> 1		is . 5	

Under normal circumsances, machine datum 383 is set to  $\emptyset$ ; the standard, fixed position control up-date time is active. It can, however, be increased with the use of this machine datum.

If the set up-date time (MD 383 =  $\emptyset$ ) is insufficient, it should be increased, but only after consulting GWE-TN4 (engineering).

Axis number	Ident <del>-</del> number	Addr.	Sign	S. B.	Displ	HORNAUNA.		
And a second second	384	S	± t		2nd. softwa n X (-) di only for 3 minus dire	$\mathbf{\mathbf{F}}$		
J. S.		Position control units		Inf	out limīts	Incre- ments	Units	
Metric; degrees		$\frac{1}{2} \cdot 10^{-3}$ mm		ø	±9999 9999	1 ¹⁰ 1	1 um; 10 ⁻³ deg	
Inch		1 • 10 ⁻⁴ in		ø	±9999 9999	1	10 ⁻⁴ in	

* In versions 0 and 2, the input limits are : ± 9 999 999

This 2nd. software limit switch is activated with a "high" ("H") signal (+ 24 V),on bit 6,input byte 3.

Axis number	Axis Ident- number number Addr		Sign	T	Displ			
Spindle	386 387 388 389 390 391 392 393	S	*	A	Acceleration time for 8 gears		onstant	$\mathbf{\overline{S}}$
<b>.</b>	onardhait	<b>[</b> 1		Inp	ut limīts	Incre- ments	Uni	its
ANI GOOD			Por.	Ø	32 000	1	4 п	າຣ

For acceleration, the control generates the command value in the form of a ramp, in dependency of this machine datum. The MD acts like a variable ramp generator.

The setting is determined by measuring the time it takes to accelerate the motor from speed  $\not 0$  to the maximum speed.

This time interval is entered as the machine datum, after unit conversion. Only for 3T/M 3, starting with software edition 02.

Example : Gear range 1 Acceleration time: 400 ms -- 386 S 100 Gear range 2

Acceleration time: 580 ms -- 387 S 145
#### 7.4 Machine Data Bits Description

The individual machine data bits are described in the order of input numbers, and further, starting with bit 0 through 7. Under circumstances, several input numbers, respectively bits, are described jointly.

#### 7.4.1 Address Designations

N400 Bit 0 - 3:

Address (name) for radii and chamfers to be inserted (see the programing instructions for 3T and 3M, section 6).

This address will also be used as address for tool nose radius (3T), respectively cutter radius (3M). (See Operating Instructions 3,section 8.1 .)

Normal designation for 3T: 8

3M: P

#### N401 Bit 0 - 3:

Address (name) for the coding of tool position (tool nose position), only for 3T, normally A (see programing instructions for 3T, section 4.2.2, and the operating instructions 3, section 8.1.1).

#### N402 Bit 0 - 3:

Address (name) of the 4th. axis,option AD4,only for 3M (see the programing instructions 3M, section 2.1).

Address coding :

	B	Name		
3	2	1	0	. s.S.
0	Õ	1	10	λ
0	1	0	0	В
0	1	0	1	С
0	1	1	0	U
0	1	1	1	v
1	0	0	0	W
1	0	1	1	Р

The bit combination determines the address character of the corresponding key: N400 bits 0 -3 for the 58 key (3rd. line,4th. key),and N401 and 402 bits 0 - 3 for the 4th. key in the 2nd. line. When these keys are activated,the designated character appears in the display.

The same designation applies for programing (MDI or tape).

#### 7.4.2 Time Settings

N400 Bits 4 - 7:

Time (duration) of strobe signal

N401 Bits 4 - 7:

Delay time for strobe signal

N402 Bit 4 - 7:

Duration for signals MO2, M30 (program end)

Time setting 3T/M O and 2 software 04,05

2	Е 6	317 5	Г 4	Time 3T	[ms] 3M
0	0	0	0	18	20
0	0	0	1	36	40
0	0	1	0	54	60
0	0	1	1	72	80
0	1	0	0	ິ 90	100
0	1	0	1	108	120
0	1	1	0	126	140
0	1	1	1	144	160
$\mathbf{\tilde{i}}$	0	0	0	162	180
1	0	0	1	180	200
1	0	1	0	198	220
1	0	1	1	216	240
1	1	0	0	234	260
1	1	0	100	252	280
1	1	1	0	270	300
1 4	1	4	4	200	220

Time setting 3T/3M 0 and 2 software 06,07,08 and 3*M/T3 01+04* 

7	BI 6	T 5	4	Time 3T	[ms] 3M
0	0	0	0	16	18
0	0	0	1	32	36
0	0	1	0	48	54
0	0	1	1	64	72
0	1	0	0	80	90
0	1	0	1	96	108
0	1	1	0	112	126
0	1	1	1	128	144
1 1	0	0	0	144	162
1	0	0	1°	160	180
1	0	d ⁰	0	176	198
1	0	1	1	192	216
N.	1	0	0	208	234
1	1	0	1	224	252
1 1	1	1	0	240	270
1	1	1	1	256	288

The times can be set in 16 steps, with 2-fold the up-date time *), common for all switch and auxiliary functions. The transfer of the switch and auxiliary functions to the interface control should be accomplished with the rising edge of the strobe signals.

The strobe signal and delay times are set in the same manner for the PC.

*) (actual value up-date time)



Example of a programing of M, S, and T-word in one block :

7.4.3 Reference Point Approach Direction (Axis-specific)

(Reference point approach in minus direction) N403 through 406, bit  $\emptyset$ .

Direction of approach	Bit Ø
Minus	1 , tomatio
Plus	Ø

# 7.4.4 Sign Change for Velocity Command Value, Partial Actual Value (Axis-specific): see also chapter 6.

N403 through 406, bit 1. (Sign change for velocity command value)

Velocity command value at positive axis motion	Bit 1
Negative	1.0110
Positive	Ø

N403 through 406, bit 2: (Sign change for partial actual value)

Positive partial a value is calculate	actual Bit 2 ed	
Negative Positive	1 Ø	4

### 7.4.5 Multiplication Factor for the Partial Actual Value (Axis-specific)

N403 through 406, bits 3 and 4 and 6.* (Partial actual value to be multiplied, respectively divided, by 2 or 10

Possible combinations:

		5.5 S	6.5
Factor	Bit 6 *	Bit 4	Bit 3
1 0.5 2 5 10 20	0 0 0 1 1 1 1	0 1 0 1 0 0	0 0 1 0 0 1

With this factor, the increment resolution of the position control system is adapted to the interpolation resolution of the control. See the table on the next page.

For encoders with pulse numbers deviating from those given in the table, or for ball screws with different lead values, the adaption must be done on the machine (e.g. through a gear).

only for version 3, starting with software edition 03

1	1.		4	Linear ax:	is		2		4	Rotary axis
Interpolator unit	140 ^{,2}	C. BON	30,	0.5	hw	Califica . O		Carles!	2	0.5 x 10 ⁻³ 0
Factor (Machine datum)	1/2	1/2	1	Sauto	2 6	2	1	2	1	2
Resolution	0.25µm	0.25µm	0.5µm	0.5µm	1µm S	1µm	0.5µm	1µm	0.5µm	1 x 10 ⁻³ 0
Pulse valuation of pos.con- trol (input pulse is quadrupled)	0.25µm/p	0.25µm/p	0.5µm/p	0,5µm/p	tµm/p	1µm/p	0 <b>.5µm/</b> p	1µm/p	0 <b>.5µm/</b> p	1x10 ⁻³ •/p
Pulse valuation acc.to pulse shaper circuit	1µm/p	1µm/p	2µm/p	2µm/p	4µm/p	4µm/p	2µm/p	4µm/p	2µm/p	4x10 ⁻³ °/P
Pulse shaper circuit	-	S		-		-		5-fold	5 <del>-</del> fold	5-fold
Encoder pulse nr./rev. or grid constant	200	2500	2000	2500	2000	2500	5000	20µm <	20µm	18 000
Max. encoder frequency per channel	100KHz	100KHz	100KHz	100KHz	100KHz	100KH	z	25KHz	12KHz	25KHz
Feed screw lead mm	2	2 5	4	5	8 _6	10	10	-	-	directly
Vmax. of axis,depen- dent on encoder m/min	6	6	12	12	24*	24*	12	30*	15	30 000 °/min
Electrical encoder limit speed	<b>3000</b> rpm	2400 rpm	3000 rpm	2400 rpm	3000 rpm	2400 rpm	1200 rpm	Cardle .	-	83.3 rpm

* Travel velocity for rapid traverse = Vmax axis, however, maximum 15 m/min The factor (machine datum) is set axis-specifically

Rotary encoder for linear axes 6FC9 320-3C

Table for the Selection of the Position Encoder

7-43

#### 7.4.6 Rotary Axis

N403 through 406, bit 5 :

Position control system programing	Bit 5	Use
Degrees	- ³⁶ 1	Rotary axis
mm or inch	0	Linear axis

With this bit,after 256 revolutions,the actual value is reset to  $\emptyset$ . With this bit,the rotary axis is recognized with degree programing.

#### 7.4.7 Exclude the Axis in the Start Interlock

N403 through 406, bit 7: -starting with software edition 07 for 3T/M 0 & 2 -starting with software edition 03 for 3T/M 3

The start interlock for individual axes can be excluded, even if 407 bit 7 was not set. For instance, 3M with only X and Y-axes; a short-circuit (*dummy*) connector must be installed for the Z-axis.

If bit 7 of 405 is set and bit 7 of 407 is not set, only the X and Y-axes must be brought to the reference point for interlock of NC start.

#### 7.4.8 Spindle with ROD Encoder

N407, bit Ø, gear range ratio: (spindle actual value,2-fold)

Spind multi	le actual value olied by	Bit O
à	2	1 D

In order to achieve higher spindle speed, it is possible to gear the ROD encoder down (with a ratio of 2:1) to the spindle, in which case the spindle actual value must be multiplied by 2. The maximum allowed spindle speed can thereby be doubled (to max. 9999 min⁻¹).

Limit values for the ROD encoder (ROD 426): max. 100 kHz,corresponding to 6000 min-1 (rpm).

N407, bit 1: (sign change for spindle actual value)

Positive partial actual value is calculated	Bit 1
Negative	1
Positive	0

The measured spindle actual value takes into account the sign change.

N407, bit 2: (pulse coder installed)

Installed pulse coder	Bit 2
yes	1 44
no	D

This bit activates the hardware monitor of the spindle encoder (alarm 224), and the display of the spindle actual value.

#### 7.4.9 Speed in 0.1 rpm, N407 Bit 3

The spindle speed is programed 10-fold. Example: for 99 rpm = S990 programed. The speed is displayed correctly (99 rpm). The maximum speed when this bit is set, is 999.9 rpm. Only for 3T/M 3, starting with software edition 02.

#### 7.4.10 Reference Point

(NC-start release without reference point) N407, bit 7:

Bit 7
1 NOT 1
» ^с О

If bit 7 is not set, the reference points of all axes must be approached after the control is switched on, because otherwise the NC-start will be inhibited in the MDA and AUT operating modes (alarm 351). Also see section 11.4.7.

#### 7.4.11 Auxiliary Function Output

N408, bit 0 and 1:

(Auxiliary function output during sequence number search)

Auxiliary function output	Bit
during SNS	O 1
None	0 0
After NC-start	0 1 ⁻
During SNS	1 0

According to machine manufacturers, the output of the auxiliary function must be determined during sequence number search; also see the interface description for system 3, section 3.5.

N408, bit 2:

(Auxiliary function output prior to travel)

Auxiliary function output	Bit 2
Before movement During the movement	1

According to the machine manufacturers, it must be determined whether the auxiliary function output occurs before or during the travel of the axis.

#### 7.4.12 Evaluation of the Programed Data with S-analog

N408, bit 3:

(Short-circuit for S-input data)

Evaluation of programed speed and direction	Bit 3
Internally in NC	1
Through interface control	0

The two possibilities are explained in detail in the interface description system 3, section 3.9.1 . When bit 3 = 1, the NC interprets the programed spindle speed, respectively surface velocity and MO3, MO4, MO5 internally, and outputs it as analogue spindle speed command value. Overwriting via the interface for S and direction is possible through the "External Data Input". The overwritten values are active until "RESET" or end of the program; during this interval, the programed data are suppressed.

The programed S-values are always processed by the PC, even when bit 3 = 1, if the FB21 and FB22 function blocks of the PC are active; this is described in the following paragraph for bit 3 = 0.

When bit 3 = 0, the interface control will decode the BCD data outputted by the NC, and will return them via the "External Data Input" to the NC. The interface control can thereby modify the data for special functions such as gear change and chip braking. The NC internal processing of programed data for the spindle is inactive.

#### 7.4.13 Position Control Feed-back and Input System, Metric/Inch

N408, bit 4 (position control system), bit 6 (reset state of input system) N408 bit 5 must always be set to  $\emptyset$ .

These machine data bits become active only after PORESET.

Option B41 is required for cases where bits 4 and 6 are not equal.

N408, bit 4:

(Position control system installed on the machine)

Position control system	Bit 4	Units
Inch	ر ۲ ۱	<u>1</u> • 10 ^{−4} in
Metric	0	12 • 10 ⁻³ mm
197 - 197 - 197 - 197 - 197 - 197 - 197 - 197 - 197 - 197 - 197 - 197 - 197 - 197 - 197 - 197 - 197 - 197 - 197		3.4

N408, bit 6:

(Reset state of the input system)

Reset stat syst	e of input em	Bit 6	Units
Inch	<ul><li>▲ G70</li><li>▲ G71</li></ul>	1	1 • 10 ⁴ in
Metric		0	1 • 10 ⁻³ mm

For operation and programing, the relationship of different machine data on certain units must be taken into consideration.

T	he following machi	ne data are dependent	on <u>bit 4</u> (position co	ontrol system):
	N100103	N170173	N220223	
	N110113	N180183	N352	
	N150153	N190193	N385	
	N160163	N210213		
т	he following displ	ay values depend on <u>b</u>	<u>it 4</u> :	
	800 - 801			
	810 - 813 💍			
	830 - 833			
	840 - 844			
۰T	he following machi	ne data depend on <u>bit</u>	4 (input system) :	
	N120123	N350	N356	
	N130133	N351	N371375	
	N200203	N355		
A	lso dependent on <u>b</u>	<u>it 6</u> (input system),b	ut <u>not</u> on programed G	70/G71 :
	Actual value dis	play, 🛇		

Zero offset (1st.through 4th. and external),

Surface velocity G96,

Feed rate G94,95,

Tool offset,

Incrementing - increments,

The programed zero offset G59 depends on G70 or G71; the content of the setting datum is processed according to the input system.

#### 7.4.14 No Deceleration at Limit Switch

(Quick-stop at the software limit switch) N408, bit 7:

Digital velocity command value	Bit 7	
0 Deceleration ramp	0 0	10 10 10

(for effects, see the next page)

7-48



- A Actual value counter content = position value of software limit switch
- B Start-point of deceleration
- C Stop-point without deceleration
- D Stop-point with deceleration
- E Position value of the software limit switch (N160...163 and N170...173)
- F Following error (dependent on Kv and v)
- R Computer timing (actual value up-date time)
- 0 Overrun

If the actual value counter shows the same value as the position value of the software limit switch (point A), due to the computer timing R (actual value up-date time), at low deceleration at point B :

- When the bit is set, the digital velocity command value  $\emptyset$  is outputted, and only the following error A is eliminated. The overrun D (E - C) results from the computer timing R; it can be practically  $\emptyset$ , or it can be negligeably small (see the note for the worst case value, on next page).
- When the bit is not set, the digital velocity command value is brought to Ø according to a ramp function. The overrun (E - D) results from the computer timing R and from the value for acceleration and decelaration set under N120...123.

7-49

The overrun portion caused by the computer timing is negligeably small, especially relative to the total deceleration distance.

#### Note:

The distance that can be traveled between actual value scannings is negligeably small; in the worst case, the <u>maximum</u> distance can be at 10 ms up-date time, and axis velocities from 15 m/min:2.5 mm, and from 1 m/min: 167 µm.

#### 7.4.15 Tool Length Compensation for: Non-programed Axis (Only for 3T)

N409, bit O (only for 3T) :

Length compensation can be executed even for axes which are not programed.

Length compensation for non-programed axis	Bit O
yes	1
no	O

If the bit is set, then during the selection, cancellation, or offset number changes, even if one axis is not programed, the tool length compensation will be executed for it as well (see the programing instructions for 3T, section 4.2.1 ).

#### 7.4.17 Diameter Programing (only for 3T)

N409, bit 2:

(Diameter programing of X-axis with G90)

Distance X pr	rogramed in:	Bit 2
Diameter	Jan Street	1
Radius		O series

The bit should be set according to the end-user's specifications.

#### 7.4.18 PC Present

N409, bit 3:

The bit must be set if an integrated PC is present.If this bit is set, N409 bit 7 must also be set simultaneously.

#### 7.4.19 Handwheel Present

N409, bit 4: The bit must be set if a handwheel is present.

#### 7.4.20 Feed Rate not on the Contour

N409, bit 5:

For radii, if this bit is set, the programed feed rate is maintained in respect to the tool nose or cutter radius (and not in respect to the part contour, in order to prevent inadmissable feed rate changes when the radii are too small). This may be used on lathes, where small radii are often programed, and where the cutter radius is relatively large.

#### 7.4.21 Option 2nd. Input/Output Board

N409, bit 6:

When a second I/O board is present, this bit must be set. If bit 6 is not set, the S-values will be outputted in two sequences  $(10^3 \text{ and } 10^2, \text{ then } 10^1 \text{ and } 10^0)$ .

This bit ought to be set only for version 1.

#### 7.4.22 NC - PC Interface Activation

#### N409, bit 7:

This machine datum activates the interface. If the bit is set, interface signals can be transferred. Attention should be given to the fact that an interface be present, and that machine datum N409 bit 3 be properly set.

The operating modes are activated with the mode selector switch only if this bit is set. This is a prerequisite for the down-loading of the machine data tape.

#### 7.4.23 Key Switch Active for Some Operating Modes

N410, bit 0 - 7:

(according to customer's request)

The appropriate function is interlocked with the key switch when the corresponding bit is set. The following functions can be interlocked:

Bit #	Function	System 3 Operating Ins- tructions (section with explanations)
ο	Superimposing	7
1 _	Sequence number search	7.2
2	Dry run	1.7
3	Part program editing *)	7.4
4	TO-data, incremental input	10 ² 8
5	TO-data, absolute input	8
6	ZO-data (ZO, SE)	8, & 9.3
7	DATA-start in MDA	4.2 & 4.3
1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1		

*) and cancel

#### 7.4.24 I/O Interface V24 (RS 232 C)

N411 and N412, bits 0 through 7: (Baud rate and coding of the input and output device) This specifies the designation of the input and output devices.

N411 indicates the input, and N412 indicates the output of device connected to board 03 100, connector X103. The interface can be operated as V24 (RS 232 C) or as 20 mA full duplex interface; N416 bit 0 must thereby always be set .

See section 2 for the meaning of the 8 bits.

#### 7.4.25 EIA code for @ Sign (A)

N413, bits O through 7:

The EIA code contains no @ sign, it therefore becomes necessary to select a function key to represent the @ key.This tape bit pattern must be set here.

**(B)** 7.4.25 Name of Main Axis Associated with the 4th. Axis

Bit		Name
1	0	4
0	0	X
0	1	Y S
1	0	Z

#### 7.4.25 DC Control Character Without Parity (C)

Versions 0 and 2, starting with software edition 06 ,starting with software edition 02 Version 3 If the bit is set to O, the DC signals are sent with parity.

#### 7-53

#### 7.4.26 Option Bits

	Š			Bit	30			3
	7	6	5	4	3	2	1	O
N415	1	x	1	X	1	Х	1	X
N416			<ul> <li>X</li> </ul>	X	X	1	1	1

The following bits must be set for the <u>3M</u>:

		Bit							
	7	6	5	4	3	2	1	0	
N415	1	х	x	x	1	Х	X	X	
N416			X	× X	Х	1	7	1	

X Individual bits are set according to start-up instructions and data sheet of the control (check list in section 1.1)

- N415, bit 1: Thread and feed per revolution (only 3T) contained in the basic control.Bit 1 must be set to "1".A spindle encoder must, of course, be present.
- N415, bit 3: Teach-in, playback and MDA (see operating instructions, section 4) are contained in the basic control.Bit 3 must be set to "1".
- N415, bit 5: Analogue spindle speed (only for 3T).For the 3T,bit 5 must be set to "1",and bit 1 of 415 must also thereby be set to "1".
- N415, bit 7: Tool nose radius compensation for 3T, respectively cutter radius compensation for 3M, is contained in the basic control. Bit 7 must be set to "1".
- N416, bit D: V24 interface (RS 232 C) on X103 connector of CPU 03100 for the devices described in section (contained in the basic control).
- N416, bit 1: Cycles for turning, respectively drilling operations, according to the programing instructions, section 7. The cycle programs are stored in the RAM memory of the control (basic).
- N416, bit 2: Alarm texts additional to the coded displays are contained in the basic control version.

#### 7.4.27 Actual Values of X-axis at 3T, Displayed in Diameter

N416, bit 6: When the bit is set, the actual values for 3T are diplayed in diameter. N40**9** bit 2 must, however, be set. Versions 0 and 2, starting with software edition 06 Version 3, starting with software edition 02.

#### 7.4.28 Block End with Carriage Return and Line Feed

N416, bit 7:

The program output is normally with LF,CR,CR. An output of CR LF is necessary for DNC operation. CR LF is the output if the bit is set.

7.4.29 14 BIT DAC (Digital/Analogue Converter) : Very important ?

#### N417, bit O:

The bit may not be set when position control board 03320-03323 is used. 1 VELD =  $\frac{10 \text{ V}}{2048}$ 

The <u>bit must be set when position control boards 03325 and 03350 are used</u>. 1 VELO =  $\frac{10 \text{ V}}{8192}$ 

The following machine data must be checked in conjunction with this : Nr. 140-143, nr. 354/nr. 230-233 Versions 0 and 2,starting with software edition 06

Version 3, starting with software edition 02

#### 7.4.30 Wear Input in Diameter

N417, bit 2:

Only 3T and X-axis.

The incremental input value is divided by 2 before it is entered into the tool offset memory.

Versions 0 and 2, starting with software edition 06

Version 3, starting with software edition D2

#### 7.4.31 Spindle Override Active in Threading

#### N417, bit 3:

If the bit is set, the spindle override will be active even if G33 or G63 is programmed.

Versions 0 and 2 starting with software edition 06 Version 3, starting with software edition 02

#### 7.4.32 Deceleration to the Velocity of the Next Block

N417, bit 4:

The velocity is changed to that programed for the next block, according to the deceleration ramp at contouring (G64).



Wood work applications

Versions 0 and 2, starting with software edition 06 Version 3, starting with software edition 02

#### 7.4.33 Test Bits

N418, bits 0 through 7:

Since some bits are used for test purposes, these machine data bits must be set to  $\emptyset$ .

#### 7.4.34 Display Bits

N419, bits 0 through 7:

If the timing sequence of the NC software is interrupted (e.g. the NC cannot access because of the access of the PG 670), these bits are set to "1".

The bits can be reset only through PORESET.

#### Content

- 8.1 Signal Display
- 8.2 Pin-out of Input and Outout Connectors
- 8.3 Code Tables
- 8.4 Data of the Signal Transmission (External Data Input)
- 8.5 Coupling of the Machine Control Panel to the PC
- 8.6 Possibility of Connecting Additional Operating Functions
- 8.7 Interface Adapter Connector and Adapter, Position Control Diagnostic Connector

Note:

For measuring and separation adapters, see section 9.13

#### 8.1 <u>Signal Display</u>

The interface test is used during malfunctions, to find out whether the cause lies inside or outside the SINUMERIC, and whether the VDI signals are properly connected.

Important: The output stages are not checked for the output signals of the digitalinput/output board 03 400. When in doubt, the corresponding connector pin should be checked with a measuring device or a diagnostic program.

Interface Test - Procedure

Select TEST

TEST

Nr.

The input and output signals are selected with the page key

DO/DI, w	ith the	mode	key	8	to	page
test 🕢	3 ⁰⁰					
an io	í					
16 ¹⁶	1	<b>F</b>				
•						

Operating mode selector switch.except in

to

The signals are produced in the same signal form in which they can be found in the integrated PC interface. The name designation for the signals is adapted to the PC interface.

11





#### Note :

- Outputs <u>cannot</u> be set with the interface test; this can only be done with the engineering panel (operator panel and test board).
- The operating mode selector switch can only be tested for each individual position, because the mode (display range) is changed with each switching.

See section 2.8 for lists of the interface signals

#### 8.2 Pin Layout of the Input and Output Connectors

The connection is made separately for inputs and outputs, on a 50 pole subminiature connector.

Cable: SINUMERIK standard cable 6FC9 340-2W.

Input connector X402, IN

		Conr	necto	r Pir	1			
7	6	5	4	Bit 3	2	1	0	
8	7		F			N. GDC		
16	15	14	12	12	2	2		
							9	
220	24	23	22	21	20	19	18	
33	32	31	30	29	28	27	26	
[°] 41	40	39	38	37	36	35	34	
49	48	47	46	45	44	43	42	
	7 8 16 25 33 41 49	7       6         8       7         16       15         25       24         33       32         41       40         49       48	Conr 7 6 5 8 7 6 16 15 14 25 24 23 33 32 31 41 40 39 49 48 47	Connector 7 6 5 4 8 7 6 5 16 15 14 13 25 24 23 22 33 32 31 30 41 40 39 38 49 48 47 46	Connector Pir         7       6       5       4       3         7       6       5       4       3         8       7       6       5       4         16       15       14       13       12         25       24       23       22       21         33       32       31       30       29         41       40       39       38       37         49       48       47       46       45	Connector Pin         Bit         7       6       5       4       3       2         8       7       6       5       4       3       3         8       7       6       5       4       3       12         16       15       14       13       12       11         25       24       23       22       21       20         33       32       31       30       29       28         41       40       39       38       37       36         49       48       47       46       45       44	Connector Pin         Bit         7       6       5       4       3       2       1         8       7       6       5       4       3       2       1         8       7       6       5       4       3       2       1         16       15       14       13       12       11       10       25       24       23       22       21       20       19         33       32       31       30       29       28       27         41       40       39       38       37       36       35         49       48       47       46       45       44       43	Connector Pin         Bit         7       6       5       4       3       2       1       0         8       7       6       5       4       3       2       1       0         8       7       6       5       4       3       2       1       0         9       25       24       23       22       21       20       19       18         33       32       31       30       29       28       27       26         41       40       39       38       37       36       35       34         49       48       47       46       45       44       43       42



#### Output connector X403, out

Byte	Connector Pin									
nrs.in refe-	-									
board)			<b>)</b>		3	2		0		
0	16	15	14	13	12	11	10	9		
1 300	25	24	23	22	21	20	19	18		
2	33	32	31	30	29	28	27	26		
P24 (+24V)		À	1	2	1 3 f	rom :	inter	face		
Mext (OV)			34	35	ر ۲	rom	inter	face		



#### .÷.3

#### Code Tables

# 8.3.1 Operating Mode Selector Switch (Gray-coded) S15 (acc. to diagram 03720)

TEST Nr. 7 Byte O

Code Table

200		10aur		Co	de	25	Sional 🔊
Position	tion	Symbol	D	C	B	*	Name
10.0 1	and a start	$\mathbf{O}$	0	0	0	1	DO
2 3 _ 🖉	5 S	] 🔊	0	0	1	3 ¹	DI
4 5	7		0	0	1	0	MDA
	11		1	0	1	Q	JOG
7	13	1.00	1	0	1	1	
8 🦽	15	10	1	0	0	ςš¶	
9	17	100	1	0	0	0	I INC
10 🚿	19	1000	1	1	0	0	15 ²⁴
_े 11	21	10000	1	1	0	1	5
12	23		0	1	0	1	MDI-PP
13	25		0	1	0	0	MDI-SE-TE
14 🔊	27		0	1	10	0	AUT 🚿
15	29		1				. www.
16	31		1	1	1	0	REF

Software generated clear-functions (at switch-over):

- *) : RESET (Clears all unbuffered controls and memories, except for block nr., actual value, and machine data memories, like at program end and reset)
- +) : CLPROAC (Clears "Program active" and "Feed hold")

Also, when switching over from MDA to AUT and vice versa : RESET -to prevent interferences in the automatic program with TEACH-IN and PLAYBACK.(RESET occurs only within program mode.)

8.3

8.3.2 <u>Axis Selector Switch S18</u> (only 3M, acc. to diagram 03720) TEST nr. 7,byte 2

Code Table	Nº	Nº.			
Position	ode Table Disition Connection	Symbol	Co B	de	jo N
1	ی ۱	X 🔊	0	0	]
2	3	Y	0	1	
3	5	Z	1	0	jó
4000	7	e ⁵⁵ 4	1	18	8
and the second s				A.	

8.3.3 Feed Override Switch (Gray coded) S17 (acc. to diagram 03720)

TEST nr. 7, byte 0

Code	Table
12	

2	24	Symbol	4	Co	de	
Position	Connection	F 🐝 \$	D	C	B	
danto'r	1 .50	0	6	0	3 ⁶ 0	0
2	3	1	0	0	0	1
3	5	2	0	0	1	1
4	7	<u>_</u> 4	0	0	1	0
5 💉	9	6	0	1	10	0
6	11	8	0	1	59	1
7	13	10	0	્લેં	0	1
8	15	20	0	1	0	0
9	17	40	1	1	0	0
10	19	60	1	1	0	1
- 11 J	21	70	1	1	1	21
12	23	° 80	1	10	50	0
13	25	90	1	Ő	1	0
s ⁶ 14	27	100	1	0	1	1
15	29	110	1	0	0	1
16	31	120	1	0	0	0

TEST nr.7, byte 1

to x	NAS.X	Symbol	C	ode	
Position	Connection	sΩ*	C	B	*
1 martin	1	50	1	1	100
<u></u> 2	3	60	1	1	0
3	5	70	0	1	0
4	7	80	0	1	1
5	9	90	0	0	1
6	11	100	0	0	0
7	13	110 .	1	0	0
<u>_</u> 8	15	120	10	0	1
			. N.		

8.3.5

Gear ranges (Coded Input Signal)

TEST nr. 8, byte 4

			8
Gear	C	<b>B</b> S	
1	0	0	0
2	0	0	1
à 3	0	1	0
4	0	1	1
5	1	0	Э
6	a li	0	1
7	1	1	0
8	1	1	A
Dit-N		6	5

Bit	Code	3T Signal	Code	3M Signal	103	Data wo	10 ² ord - 1	10 ¹ 3CD -	10 ⁰	
Datum	340 ^{,9}	4 3 2 1 0 E D C B A	2	43210 EDCBA		654 PON	03210 MLKI	7654 HGFE	3210 DCBA	340.9
Function	Axis		Axis	JOR	14-	Maximum	numerical	value per	decade	
Incremental	x	00001	x	00001	<u>+</u>	7	9	9	9	
(additive calcu- lation. no memory	z	00010	Y	00010	±	7	9	9	9	2
of its own)	No.X	-	≤ Z	00011	± '	7 20	× 9	9	9	Nº.
nout on		-	4.	00100	±	7.00	9	Stornal 9	9	рт. Г
Incremental	X	01001	x	01001	<u>+</u>	7	9	9	9	
zero offset	z	01010	Y Y	01010	±	.7	9	9	9	
(additive calcula- tion, absolute in	100		z	01011	±	7 240	9	9	9	Nº.9
its own memory)		-	4.	01100	±	all and	9	Stornau 9	9	5
S U/min; m/min		11000	 !	11000	Salah.	9	9	9	9	
F mm/min		11001		11001	1	9	9	9	9.1	
% Program nr.	A2.8	11010	1 ²	11010		9 🖋	9	9	9	Nº.S
PC alarm indica- • tion		10000		10000	0	7 BIT	ASC II- Code	0 7 BI1	ASC II- Code	

Maximum value for tool offset and zero offset:

<u>+</u> 7.999 mm ; **±**0.7999 "

*) Only for version 3, starting with software edition O2 FB22 must have edition O2 See interface description section 3.10

8-7

Data of Signal Transmission (external data input)

.8

₽C Incuts		. 8		👌 Dat	<b>c</b> a bit	6		6
Byte-Adr.	7	146	5	200	3	2	1	0
	Operat	ing Mode S	elector Swi	tch	Feed/Raj	pid Trave	rse Overri	Je Switch
	D	۲ <b>С</b> ,	86	A 5	<b>D</b> 4	С з	1 ° 2	A 1
	Key switch	Dry run	Block delete	Single block	SNS 12	Spindle	e override	switch
adka ?	Rapid tra verse over ride 25 active	Rapid tr. superimpo sition ²⁴	Direction 23	al keys *) 22	Axis selec Co B 21	tor swt ^{*)}		112
	Spinlde OF	Spindle ON 32	Feed HOLD	Feed Start 30	•)	Cycle STARI ₂₈	free 27	free <b>26</b>
	free 41	free 40	free 39	free 38	free 37	free '36	free 35	free 34
and the R	free 49	free 48	free 47	free 46	free 45	free 44	free 43	free 42

#### 8.5 Coupling of the Machine Control Panel 3M to the PC

For 3T: for the directional keys X+,X-,Z+,X-,and the handwheel switch
 X,see the interface description,section 4

1... 49 pin numbers of the 50-pole input connector

8.6 Machine Control Panel Connector Board with Option for Connection of Additional Operating Functions



26, 27 and 34 through 49 are free

Soldering points for the connection of additional functions of the machine control panel.

#### 8.7 <u>Interface Adapter-connector and Adapter, Position Control Diagnostic</u> <u>Connector</u>

#### 8.7.1 Interface Adapter-connector

The following separation adapters are available for test purposes for the wires connected to the boards:

15 <del>-</del> pole:	Ident-Nr.	400	91	337	
25-pole:	Ident-Nr.	400	91	350	
50-pole:	Ident-Nr.	400	91	374	

#### 8.7.2 Interface Adapter

Interface adapter for output signals with switch, test jack, and switchable LED per each output; in housing 220 x 130 x 50 mm³, with 50-pole connector (male) ,and cable 0.4 mm long with 50-pole connector (female) for connection in between an output line of 03 400. Device designation : 6FC9 330-OBA Order number: Ident-Nr. 706 88 203

#### 8.7.3 Position Control Diagnostic Connector

This connector (designated MKDS in the System 8 diagnostic case, jumper connector for the actual value) is used to test the control without connected position encoder, respectively in the absence of an axis. One connector is required for each axis.

Order number : 🔥 Ident-Nr. 400 91 279

This jumper connector can be produced in the following manner: Sub-miniature connector, 15-pole, female (complete), order-nr. 6FC9 341-1EC.

Wiring:

	••	43		100	0	D D		7.2	v	
_	1	2	3	52	4	5	6	7	8	
	2	Ŷ	0		2	<u> </u>	0	3	0	7
	6	7	5	0	0	ō	Ś	0	ر د	ſ
	9	1	0	11	12	<u>(</u> 1)	8 1	4	15	
	Z	*	Z						M ( 0	V)

as seen from the wiring (back) side

Characterizes: System 3 & 8 MKDS

The connector and adapter can be ordered from : SIEMENS AG, ZN Nürnberg-Werkstatt, Würzburger Strasse 121, 8510 Fürth 9. Board Overview and Strappings

- 9.1 Generalities
- 9.2 CPU board 03100
- 9.3 EPROM Memory 03201/03202
- 9.4 RAM Memory 03210
- 9.5 Test Board 03220
- 9.6 Test Board 03221
- 9.7 RAM Memory 03260
- 9.8 Position Control (Actual Value) 03315
- 9.9 Position Control (Actual Value) 03315
- 9.10 Position Control (Command Value) 03320
- 9.11 Position Control (Command Value) 03320/03323
- 9.12 Position Control 03325
- 9.13 Position Control 03340
- 9.14 Position Control 03350
- 9.15 Integrated EXE 03390/03395
- 9.16 I/O Board 03400
- 9.17 Input Board 03410
- 9.18 Output Board 03421
- 9.19 I/O Board 03450
- 9.20 Output Board 03460
- 9.21 Power Supply 03501/03502
- 9.22 Power Supply 03410
- 9.23 Operator Panel 03700
- 9.24 Operator Panel 03710
- 9.25 Machine Control Panel 03720
- 9.26 Operator Panel 03770
- 9.27 Operator Panel 03780
- 9.28 Couple Board 03800
- 9.29 Video Interface 03810
- 9.30 Video Interface 03811
- 9.31 List of Strappings of Input/Output Boards, According to the PC Program.

#### 9.1 <u>Generalities</u>

Strappings do not have to be changed during start-up, except for the addressing of I/O boards according to the user's program. When spare parts are delivered, it must be checked that the spare board strappings coincide with the "fixed strappings", according to the service manual. The variable strappings must coincide with the exchanged board according to the PC program.

The strappings are divided into the following categories :

- Fixed strappings May not be changed
- Strappings according to the PC program
   The I/O boards must be strapped by byte addresses, according to the PC user's program.
- Variable strappings, which can be :

Standard strappings (the control is shipped with these)

Special strappings (standard strappings modified at start-up)

For strappings of the 6ES5 boards, see section 10.

For measurements on the boards or on the NC bus connector, it is strongly recommended that adapter 548 187 9001.00AS (ES902-Adapter 96-pole) be used; you should thereby be careful not to drop the board.(Bolt with thread M4)

Adapter Ident-nr. 706 77 558.

The following separation adapters are available for the testing of the wires connected to the boards :

15-pole: Ident-nr. 400 91 337 25-pole: Ident-nr. 400 91 350 50-pole: Ident-nr. 400 91 374

The adapters are available from:

Siemens AG, ZN Nürnberg Shop, Würzburger Strasse 121, 8510 Fürth also see section 8.7

<u>Note</u>: Remove the strapping sockets perpendicular to the board, in order to avoid bending the pins; insert with care. The pin sequence of the strapping socket as seen from the soldering side (note the marker notch):

#### 9.2 CPU Board 03100

### 9.2.1 Fixed Strappings

Туре	Designation	Signal	Closed/Open	Remarks
Single junper	R-S	RDY	closed	offers
SPILL SPILL	AN-AM	Wait 1	" Sault	13 ¹¹¹
•	AP-AQ	Wait 4		for operator panel
n 11	М-N Q-Р АА-АВ	- Wait 3	open "	33 ²⁴
	AC-AD	Wait 3		2. ×
•	AE-AF	Wait 2	"	S. S. S.
"	AG-AH	Wait 2	" Latter	walke.
н ₂₁ 67	AK-AL	Wait I	" <u></u> (5'	A.C.
<b>H</b> 3 ^{ch}	ÁR-AS	Wait 4	<b>H</b> 3 ²⁴ .	34
**	AT-AU	Wait 0		2
•	AV-AW	Wait 0		×.
Soldering pins	AV-AW	Wait O	"	S. Con

#### 9.2.2 Variable Strappings

Туре	Designation	Signal	Standard NC active	Special NC passive
Soldering pins	C-D	M	closed	open
	E-F	+ 12 V	• <u>5</u>	н 50°
• 5	G-H	+ 12 V	н	
94	K-L	м	<b>9</b> 0	
п	н-к	TTY 3	open	closed
"	C-F	TTY I	••	"

9.2.3 Test Socket P1 (factory test for the board)

9.2.4 Switches on the CPU

S3: Test board switch (push-pull switch on board)
Reset state: upper position (test board/engineering panel routine inactive)

#### 9.3 EPROM Memory 03201

#### Fixed Strappings

Туре	Designation	Signal	Closed/Open	Remarks
Single jumper	Wait 2	Wait 2	closed	8
" Solle	Wait O	Wait 0	open	~3 ⁵
	Wait I	Wait 1	"	ALCO .
n 44	Wait 3 🚿	Wait 3	. 4	42
**	Wait 4	Wait 4		~

Remark: There is no designation for the soldering pins on board 03200 Arrangement:



EPROM Memory 03202

Replaces type 03201

Туре 🗠 🔬	Designation	Signal	Closed/Open	Remarks
Strapping socke	1-8	S ^o	closed 🔊	Adr. 17
X1	2-7		" sechi	Adr. 18
- 0-0 8 - 0-0 7	3-6			Adr. 19
• <b>0 - 0</b> 0	4-5	Å.	open	free
Single jumper	W1	Wait I	closed 💉	<u></u>
H	wo 🚽	Wait 0	open 🖉	
•	W2	Wait 2	н зай	
**	W3	Wait 3	"	
11	W4	Wait 4	<b>Q</b> 11	

#### 9.4 RAM Memory 03210

#### Fixed Strappings:

Туре	Designation	Signals	Closed/Open	Remarks
Single jumper	BI SO	RVCC RAM 0	closed	John
" (S ²	B0 🔗	+ 5 V 🔬	open	, SP
IC .	2W	Wait 2	closed .	

#### 9.5 Test Board 03220

Fixed Strappings:

Туре	Designation	Signals 👌	Closed/Open	Remarks
Single jumper	W3	Wait 2	closed	
87	W1,W2,W4,W5	W0,1,3,4	open	- -

#### 9.6 Test Board 03221

Replaces type 03220

Fixed Strappings:

Туре	Designation	Signal	Closed/Open	Remarks
Single jumper (	S16	1.2.2	open 🔊	Adr. 16
"	517	Self-	•	Adr. 17
" JON	S18 🔊		closed	Adr. 18
	S19			Adr. 19
Single junper	W3	4	closed	Wait 2
н 2	W1		open	Wait 0 💦
••	W2		• *	Wait 1
	₩4		<b></b>	Wait 3
	W5		n	Wait 4

#### 9.7 RAM Memory 03260

Replaces type 03210

Fixed strapp	ings:	8	6	č.
Туре	Designation	Signals	Closed/Open	Remarks
Strapping socke	rt I-16 🔬	5	Open	Adr. 16
XI	2-15		SH .	Adr. 17
*0°0+	3-14	354		Adr. 18
-003	4-13		closed	Adr. 19
-005	5-12	W 3_0	open 🔊	Wait 3
-0-0-	6-11	W2	· And	Wait 2
	7-10	W1	closed	Wait 1
AN. CO	8-9	wo	open	Wait O
Single junper	W4	W4	open	Wait 4

## 9.8 Position Control Loop (Actual Value) 03310 /03311

Fixed Strappings:

Туре	Designation	Signal	Closed/Open	Remarks
Strapping socket	1-16	8	open 👌	Adr. 0
X5	2-15	aller.	" "	Adr. 10
-6.9*	3-14	ST 10	" ₁₀ 110	Adr. 6 5
	4-13		20 ⁰⁰	Adr. 9
-0 05	5-12	.A.		Adr. 7
-0-0* -0-0*	6-11	4	closed	Adr. 19
,	7-10	6	"	Adr. 8
	8-9	adlar	open	Adr. 5
Strapping socket	1-16	5	open	Adr. 18
X6	2-15		J. CH	Adr. 11
-6.9+	3-14	44		Adr. 17
	4-13		**	Adr. 12
-005 -005	5-12	La.P	"	Adr. 16
• 0-0 = • 0-0 =	6-11	Cathle	closed	Adr. 13
	7-11 🔊	51	+, 3 ⁰	Adr. 15
AL COL	8-9		j ch	Adr. 14
Single jumper	P-R	SCLKSM	open	5 MHz

Fixed strappings				3	
Туре	Designation	Signals	Closed/Open	Remarks	
Single jumper	WO	Wait 0	open	à	
··	WI	Wait I	closed	, torn	
	¥2	Wait 2	open	, dbab	
Single jumper	C-D	3	open	3310 A	
••	C-D		closed	3310 B	

Remark: If there are more than 3 position actual values, board 03310 B is sandwiched to board 03310 A.Both boards have the same strapping, except for pins C-D.

#### Variable strappings

Туре	Designatior	Signals	Standard, Position act. val.diff.input	Special, Pos.act.val.assy metrical inputs
Strapping socket	1-16	*DMA	closed	open
X1 und X2	2-15	•DMB		"
Standard St.	3-14	+DMZ	<b>•</b>	н <u>"</u>
- 6-0 =	4-13	IDMA	•	closed
	5-12	IDMB	H JON	"
• • • • •	6-11 8	DMA	open	• <u> </u>
-00-	7-10	DMB	A	• 4 ⁶
	8-9	DMC		**
Special St. (not used) - • • • • • • • • • • • • • • • • • • •	al souther	ternativent	and toballomatike of	www.chautomaska.pl

### 9.9 Position Control Loop 03315

#### 9.9.1 Fixed Strappings

Туре	Designation	Signal	Closed/Open	Remarks
Single jumper "	53 54	SFBG1 SFBG2	open "	Addresses "
Single jumper "	WS1 WS2	M _{evt} -M _{int}	open "	AKO.P
Single jumper "	W1 W2		open "	Wait "
9 ¹⁰	W3	AND STREET	closed 🔹	9 ¹⁰
**	<b>W</b> '4	34	open	2

* Connection W3 on the trace of the Pcb.

#### 9.9.2 Fixed Strappings for EXE

Туре	Designation	Signal	w/o Option integrated EXE	with Option integrated EXE
Strapping socket	1-14	-	closed	open
X3 - X-Achse	2-13	•SDMB	5	11 JS
XI - Y-Achse	3-12	SMDB		a na star a s
X2 - Z-Achse	4-11	*SDMA		- 11
X4 - 4. Achse	5-10	SDMA		"
w/o Option	6-9	*SDMZ	" All	"
	7-8	SMDZ	and a second	
with Option	www.dballond	Kalit Wand	oneskant	, oballonatika.p

9.9.3 Variable Setting for Probes

Туре	Designation	Signal	Standard	Remarks
Soldering pins "	A-B/R <b>32</b> C - D <b>/R33</b> E - F	se acomaco	A 1.5k B o	For other settings, depending on probe output, see the sec- tion on interface

#### 9.10 Position Control (Command Value) 03320

Fixed Strappings

Туре	Designation	Signal	Closed/Open	Remarks
On trace	W1	Wait 1	Closed	Ś
Single jumper	АА-ВВ	5 MHz	Open	Call the

Variable Strappings

Position coder actual value strapping socket  $\gamma$  see 9.11.2.

Туре	Designation	Signal	Standard strappings Signal, Veloc Ready	Special strappings ity Control	
	×.		from drive	not monitored	
Soldering pins	W-X	Servo- Read <b>y</b> simulation	Open	Closed	2

Test Points

Туре	Designation	Signal	Remarks	Charles
Pin 🦿	R,T,V,B,D	CVS 15	Command values axis	15
Pin	Q,S,U,A,E	(internal)	Enable axis 15	
Pin	L,M,P,F,G	*SEEN 15	Control enable axis	15
Pin 🔬	N	*NC Ready	NC-Ready simulation	tomats
## 9.11 Position Control (Command Value) 03322/03323

- 1 Actual value input
- 5 Command value outputs
- 12 BIT-DAC 1VELD =  $\frac{10 \text{ V}}{2048}$
- applicable in conjunction with PC board 03310

## 9.11.1 Fixed Strappings

Туре	Designation	Signal	Closed/Open	Remarks
Single jumper " "	AF-AE AD-AC AH-AG AA-BB	1 Wait O Wait 2 Wait SCLk5M	closed open "	5 MHz
Strapping socket X2	1-16 2-15 3-14 4-13 5-12 6-11 7-10 8-9		closed open closed open " "	Addr. 19 Addr. 10 Addr. 8 Addr. 9 Addr. 5 Addr. 5 Addr. 6 Addr. 7 Addr. 3
Strapping socket X2	1-16 2-15 3-14 4-13 5-12 6-11 7-10 8-9	shan han	closed open closed " open " "	Addr. 13 Addr. 12 Addr. 15 Addr. 14 Addr. 17 Addr. 16 Addr. 18 Addr. 11

### 9.11.2 Variable Strappings

Туре	Designation	Signal	Standard strapping Posi- tion Coder Ac- tual value Dif <del>r</del> ferential Input	Special strapping:Posi- tion Coder Actu- al value w/o In- verting Input ( Asymetric Signals)
Strapping socket X1 Standard strappings	1-16 2-15 3-14 4-13 5-12 6-11 7-10 8-9	*DMA *DMB *DMZ IDMA IDMB DMA DMB DMZ	closed " " " " open "	open " closed " " "
Special strapping (not used)	icheand annuclea	Comaskan	Annioballonable pl	MANNIE BOUCONSTRANC
10 ²	estre ?	marghe P	585H0.9	Cather of

Туре	Designation	Signal	Standard	Special
A.C.	All A	4	Signal Veloci Ready	ty Control
8	6	Ó.	from drive 📎	not monitored
Soldering pins	P-N	Servo Ready Simulation	open	closed

#### Test Points

	Туре	Designation	Signal	Remarks	12. À
2	Pin " " "	T,W,Z,D,G S,V,Y,C,F Q,R,U,E,L X	CVS 15 (internal) *SEEN 15 * NC-Ready	Command values axis 15 Enable axis 15 Control enable axis 15 NC-REady simulation	

Remark: PC board 03320 (replaced by the new layout 03322) can also be used; strappings as per section 9.10

## 9.12 Position Control 03325 Replaced 03323

9.12.1 Fixed Strappings

Туре	Designation	Signal	Closed/Open	Remarks
Single jumper	S1	S board	open	Address
"	S2	S board	open	"
Single jumper	L-K	M-EXT	"	Cast and
"	R-S	M-EXT	"	
"	T-U *	D V	closed	

- 1 actual values input
- 5 command value outputs
- 14 BIT DAC, 1 VELO =  $\frac{10 \text{ V}}{8192}$
- ULA IC
- applicable in conjunction with PC board 03315
- * PCB track

Fixed Strappings

Туре	Designation	Signal 💉	Closed/Open	Remarks
Single jumper	ษพา	~	open	Wait
"	WW2	AKO.P	п зех	H
II Jornal	WW3	5	closed	H SUST
11	WW4		n saint	U Same
Single jumper	W13	Å	open	h
11	W14	1	11	

* Connection WW3 is on the tracer

9.12.2 Variable Strappings - Command Value Output

Туре	Designation	Signal	Meaning	Standard	Special
Single jumper	W7	CVG1	1st. axis	closed	see section
п	W8	ον	1st. axis	-H	"Interface"
11	W11 🦽	CVG2	2nd. axis	11	40 Me
11 500	W12	οv	2nd. axis	<b>1</b> . Š	~
н заба	W3	CVG3	3rd. axis	11	and a set
H	W4	o v	3rd. axis	11	
TT	W9	CVG5	4th. axis	JUN .	14 ³ .2
"	W10	o v	4th. axis	11	SC BEN
11	W1	CVG5	5th. axis	11	1 ¹⁰
n	W2	οv	5th. axis	11	
11	W5	CVG6	reserved	11	4
, IT	W6	o v	reserved	н	12 Con

Note: There are no soldering pins for these jumpers, also, they are not soldered when delivered by the manufacturer.

9.12.3 Variable Strappings Servo-Ready-Simulation

Туре	Designation	Signal	Standard Signal, Velocit Ready	Special y Control
CT	an ior		from drive	not monitored
Soldering pins	P-N		open	closed

### 9.12.4 Variable Strappings for Probe

### No function

Туре	Designation	Meaning	Standard Setting	Special Setting
Soldering pin " " " "	A-B/C33 A-B/C49 A-B/C19 A-B/C41 A-B/C7 A-B/C27	Com.val.1 " " 2 " " 3 " " 4 " " 5 reserve	open II II II II	see section "Interface" chapter 8
Soldering pin " " "	A-B/R26 A-B/R36 A-B/R16 A-B/R31 A-B/R11 A-B/R21	Com.val.1 " " 2 " " 3 " " 4 " " 5 reserve	A - C-B	see section "Interface" chapter 8

### 9.13 Position Control 03340

This board was shipped in low quantities.

Characteristics of version A: 3 command values, 3 actual values.

Probe input for 3T and 3M with 3 axes, one PCB.

Version B: 2 command values, 2 actual values, for 3M with more than 3 axes. Version A or B: 12 BIT DAC,ULA component, no integrated EXE, command value output as 03320; actual value - only differential input.

Туре	Designation	Signal	Closed/Open	Remarks
Single jumper "	с р	+ 5 V	open "	automatol
Single jumper	Lo on M	Warden -	arear a	Ground
Single jumper "	S1 S1	Board- Select	open closed	version A version B
Strapping socket S3 S3 S3 S3 S3 S3 S3 S3 S3 S3 S3 S3 S3	1-16 2-15 3-14 4-13 5-12 6-11 7-10 8-9	Mand Mand	open closed " " open " " closed	Address

Туре	Designation	Signal	Closed/Open	Remarks
Strapping socket	1–16 2–15		open "	address
54	3-14		11	and the set
	4-13 5-12	16.9	closed	He g
	6-11	S. F.	open	an ^{ab}
- <u>6</u> -0-	7-10		н	Lough Control
N. Cr	8-9		in States and the second se	N.S.

## Variable Setting for Probe

Туре	Designation	Signal	Standard	and the second
Soldering pins	A-B R46 🔊		N.C.	Other settings,
H NIGOOC	A-B R47		open	dependent on probe output
Wire	X347.1F		closed	See section
	X347.5OV			"Interface"

# 9.14 Position Control 03350

Replaced 03310/03323

## 9.14.1 Fixed Strappings

Туре	Designation	Signal	Closed/Open	Remarks
Single jumper "	S1 S2	-stan	open "	ANO.I
Single jumper "	S3 S4	SPBC1 SPCB2	open "	address "
Single jumper " "	L-K R-S V-W T-U	M-EXT M-EXT M-EXT O V	open n * open closed *	sent si
Single jumper " "	W7 W8 W9 W 10		open " closed * open	Wait

* Connection W9...is on trace

Option with Option W/O Signal Designation Туре Integrated EXE Integrated EXE 1-16 closed open Strapping 11 11 2-15 socket *SDMB .. 3-14 11 X1:Z-axis 11 4-13 SDMB 11 X3:X-axis 11 w/o 5-12 *SDMA 11 6-11 SDMA 11 op-11 *SDMZ 7-10 tion SDMZ 11 8-9 with option

9.14.2 Fixed Strappings for EXE

Strapping socket X2 for spindle, always without integrated EXE.

9.14.3 Variable Settings for Probes (see section 9.9.3)

9.14.4 Variable Strapping for Servo-Ready-Simulation (see section 9.12.3)

9.14.5 Variable Strappings - Command Value Output

Туре	Designation	Signal	Command Value Meaning	Standard	Special
	ພາ	CVG1	1st. axis	closed 🔬	see the
11 JONG	ຟ2 🧬	O V	1st. axis	II JONCO	"Inter-
H (202)	₩3 🔊	CVG2	2nd. axis	H Contraction	face"
II wash	W4	ον	2nd. axis	11	section
11	-₩5	CVG3	3rd. axis	11	
II	W6	οv	3rd. axis	11	2

Note: There are no soldering pins for these jumpers; also, they are not soldered when delivered by the manufacturer.

9.14.6 Settings for Command Value Output

Туре	Designation	Command Value Meaning	Standard	Special
Soldering pins	A-B/C18	Comm. value 1	open	see sec-
" tome	А-В/С19 🔬	Comm. value 2	п	tion
H KOOL	A-B/C35	Comm. value 3	"	"Interface"
Soldering pins	A-B/R21	Adal.	the second s	see sec-
н	A-B/R22	~	~	tion
n	A-B/R23	2 ^{40^{-X}}	No.X	"Interface"

### 9.15 Integrated EXE, 03390, 03395

These boards are assembled on the position control boards 03315 and 03350. They are optional boards.

No changes may be executed on these boards.

### 9.16 I/O Board 03400

Fixed strappings for basic version D: Strapping socket X1 without jumper.

Versions 2 and 3

Туре	Designation	Signal/ Meaning	Fixed strappings	Strappings acc. to PC Program
Strapping	1-16	Address		×
socket	2-15	11	-	×
X1	3-14	yer	<u>- 10</u> 2	×
	4-13	ρ. Π	open	- 100
-00#	5-12	free	11	Jahr
	6-11	II S	in and the second	<u> </u>
	7-10	H 4	11	- 47
2	8-9	11	"	

* see section 9.30

βB

9.16a

I/O Board 03 401 Address coding socket S1:

Link designation		Address Byte value
WA 4	100	8
WA 5		16
WA 6	i	32
WA 7 🔊	1	64

# 9.17 Input Board 03410

(N)			(V)	A contract of the second se
Туре	Designation	Signal/ Meaning	Fixed strappings	Strappings acc. to PC Program *
Strapping	à	w/o effect		
socket X1		6	8	(C)
Strapping	1-16	Address	- utomo	×
X2	2–15	11	<u>. 2007 </u>	x
to and the second	3-14	H 54	open	- 44
-00# -00#	4-13	"	11	-
	5-12	1200	n Kard	
•0 0=	6–11	(n	H 600	- 600
-6 <u>0</u> -	7–10	11	Harris	50 ¹⁵⁰
C. S.	8-9	free	NI	> -
4	22	20	4	14
Single jumper	A – B	MEXT	closed	-
11	C - D	MEXT	н ""А́?"	

## 9.18 Output Board 03421

Туре	Designation	Signal/ Meaning	Fixed strappings	Strappings acc. to PC Program *
Strapping	1-16	Address		×
socket X1	2 <b>-</b> 15	11	- iton	×
500	3–14	. 11	Č ²⁰	x
	4-13	n 4 ³⁵	open	- 375
	5-12	u _	n	-
• 0 0 i	6-11	y a S	н "үс ²	
• <u>6</u> •	7-10	с. П	H Star	
10 ²⁰¹⁰	8-9	11	Harris	-O ^{life}
Single jumper	S1	14	open	
11	52	22	H	14
	B1	2	closed 🔊	- 2
11	B2	S.		
11	вз 🔊		H JON	- uton
11 110	А-В	MEXT	open	<u>8</u>
H SAME	C-D	моит	H 45.45	

## 9.19 <u>I/O Board 03450</u>

Туре	Designation	Signal/ Meaning	Fixed strappings	Strappings acc. to PC Program *	
Strapping socket X1	. Staufor	w∕o effect	- Seattornal's	- Dalloman	
Strapping	1-16	Address	closed	- 54	
socket	2 <del>-</del> 15	n	11	-	
X2	3-14	n S	open/closed **		
- <b>6-9</b> =	4-13	Øn.	open		
	5-12	11	20 ⁰¹⁰⁰	x	
	6-11	н "А	5 	×	
	7-10	11		x	
м3	8-9	n de	- :?	× x	
Single jumper	A-B	MEXT	open	- mas	
H JOSTIC	D-C	Address	n Shi	50 ⁰⁰⁰	
" AND	D-E		1	- 3	

- ** Closed on versions AA and AB Open on version AC
- * see section 9.31

#### 9.20 Output Board 03461

<u>Output Board 034</u>	<u>61</u>			
Туре	Designation	Signal/ Meaning	Fixed strapping	Strappings acc. to PC Program *
Strapping socket S1	MANNIGDOL	w∕o effect	<u>o</u> or waan	<u>C</u> o stadi
Strapping	S2	Address		×
socket SO	S3	d.		×
J.C.C.	S4 🛒 🔬	11	- 4000	×
<u>. 800</u>	S5 🔊	H S	<u>9</u> 00	x
	S6	H 3 ⁴⁵⁴		x
0 0 - **	S7	n	open	-
00	S8	Hog.	closed/open **	
<b>6-0</b>	S9 🥳	11	closed	- KOMAN
Single jumper	S18	w/o effect	<u>-</u>	8
H and a	S19	11 11	- 4944	

- ** Closed on versions AA and AB Open on version AC
- see section 9.31

9.21 <u>Power Supply 03500</u> (Type designation 6EV3 054...) Fixed strapping for 03502

Туре	Designation	Closed/Open	Remarks
Strapping	1–16	closed	Clear
socket	2-15	"	Under voltage +24 V
X1	3-14	11	Under voltage + 5 V
- 6-0#	4–13	11	Under voltage + 15 V
-0-0#	5-12	"	Under voltage + 15 V
• 0-0 ±	6–11	11	Under voltage - 15 V
•0-0•	7-10	11 .	Voltage monitor
ALCON .	8-9	<b>II</b>	Collective signals
Single jumper	S-T	closed	Thyristor (over voltage 5 V)
. 11	Z-Y 1)	<b>11</b>	Battery voltage
"	ZA-ZB	H	+ 15 V monitor
n	ZC-ZD	11	- 15 V monitor
н "С ⁵⁶	ZY-ZZ	и 🔬	VCC RAM
II Salar	X6 2)	H 4 ²⁴⁵⁴	Monitor
11	ZG-ZH	"	Current limit

1) In power supply 501, designation U-V

2) This jumper is not present in power supply 501

X2 and X3 are sockets for factory tests.

9.22 Power Supply 03510

No strappings

9.23 Operator Panel 03700 (only basic versions 0 and 2)

Fixed strappings

Туре	Designation	Closed/Open	Remarks
Pin	DA-DB		Test jumper ca. 10 m Ohm
<b>n</b> 200	EA-EB		0.1 µF
11 5	LA-LB	closed	5 V load separation

- 9.24 <u>Operator Panel 03710</u> (Versions 0 and 2) No jumpers
- 9.25 <u>Machine Control Panel 03720</u> No jumpers
- 9.26 Operator Panel 03770 (only Version 3): Keyboard PCB
- 9.27 <u>Operator Panel 03780</u> (Version 3): Interface PCB Fixed strapping Single jumper A-B,closed
- 9.28 Couple Board 03800 A

Fixed strappings

Туре	Designation	Signal	Closed/Open	Remarks
Strapping	1-16		open	Test point
socket P1	2-15	44	e' H - 4	🗧 Test jumper 🔬
	3-14	~	closed	for extension
	4-13	X2.9	open	unit
-0-0r +0 0s	5-12		" ^{(0'''}	. officer
	6-11		Barrie	Cont.
	7-10		A H	and and
14	8-9	14	"	19

Variable strappings

Туре	Designation	Signal	Standard,no output block at NC faults	Special, PC output block
Single jumper	А-В		closed	open
"	С-В	143.9	open	closed

### 9-21

## 9.29 <u>Video Interface 03810</u> (only Version 3)

Fixed strappings

Туре	Designation	Signal	Closed/Open	Remarks
Strapping	19	2	open	address
socket S1	18	AND STORES	closed	н
autor.	17 🔊	8	II JION	H JO
ALCON.	16		open	, m
34	1-5	4	closed 🗠	⁶ н 3 ³
	14	~	"	
	13		11 342.7	H ARX
all officer	12		H JONOT	" utomos
Single jumper	W1		open	~ Screen Type
H 44	W2		closed 🔹	²⁶ н 4 ²
n	W3		"	"
<b>n</b> 2	А-В		open	AND.Y
11 Northard	C-D	Star,	H MORIOL	Jonac,

## 9.30 <u>Video Interface 03811</u> (Version 3)

Replacing type 03810

Fixed strappings

Туре	Designation	Signal	Closed/Open	Remarks
Strapping	12	4	open	address 12
socket S2	13		n	" 13
2	14	NO.P	п "К	" 14
offer?	15	Claro .	H SUSS	" 15
. doalle	16		closed	" 16
and it	17		open	" 17
2	18		11	" 18
2	19		closed	" 19
Single jumper	АВ	man	open	Hardware
и озле	C D		T DALL	designation
n and see	EF		a n	Ground shield
н 🔏	KL		closed	Screen blanking
)п	MN	à	open	11 11
<b>0</b>	S1	VAB9	•	Operating mode

9.31 List of Strapping Possibilities (Addressing) of the <u>I/O Boards according to the PC Program</u>

The boards can be plugged randomly into the frame. Each board is strapped for an address range.



* On boards with edition AC, jumper 3-14 is omitted ** On boards with edition AC, jumper 8-11 is omitted

9 - 23 03400 03410 03421 03421 03450 + 03460 + Byte 48 I 24 Ū 96 I 32 0 **48** 0 **48** D **32** I **16** D 2x6 Bytes) (6 Bytes) (3 Bytes) (2x3 Bytes) (2x3Bytes) (4Bytes) (4 Bytes) (2 Bytes) ....... 45.1991.04 .......... 0000000 ..... **X1** X2 X2 **X2** 5**9** X1 • • • • • • • • • Þ..... > ] > ..... <u>-</u> > III..... Þ..... Þ::::: 



Addresses always locked

Adr

32

33 34

35 36

37 38

39 40

41

42

43

Addresses may be used on other boards

On boards with edition AC, jumper 3-14 is omitted
 On boards with edition AC, jumper 8-11 is omitted

10 Engineering Panel and Test Board

CONTENT

- 10.1 Prerequisites
- 10.2 Test Board
  - 2.1 Construction
  - 2.2 Front Plate
  - 2.3 Designation of Switches
- 10.3 States
  - 3.1 Normal NC Operation
  - 3.2 Deactivated Engineering Panel Program
  - 3.3 Activated Engineering Panel Program
  - 3.4 Engineering Panel Program Hold-state
- 10.4 Functions
- 10.5 Sequence
  - 5.1 Engineering Panel Program Activation
  - 5.2 Display
  - 5.3 Break Points
- 10.6 Engineering Panel Instructions

6.1 Display or Change Memory Content

6.2 Display or Change Register Content

- 6.3 Set Break Point
- 6.4 GO Instruction
- 6.5 Single Step
- 10.7 Input List
- 10.8 NC Address Lists
  - 8.1 Overview
  - 8.2 EPROM Memory
- 10.9 Access to PC with the NC-Engineering Panel

### 10.1 Prerequisites

### - Hardware:

The CPU hardware of the NC, as well as the operator panel with its key board and display unit, must be functional. Test board 03220 must also be connected.

#### - Software:

The engineering panel software must be plugged into the test board. From the system program, EPROMs 31 and 32 at least, must be present on the O3200 EPROM board.

- Versions 0 and 2, respectively version 3, have different EPROMs for the engineering panel.

- The PC diagnosis program for version 3 also contains the normal engineering panel program.

### 10.2 Test Board 03220

#### 10.2.1 Construction

The board contains the system and working memory for the engineering panel.The EPROM memory range for test engineering panel programs contains 20 k words.The engineering panel program program,however,only uses 2 EPROM for PROM locations 71 and 72.The RAM range is 2 k words. The board also contains the wait-state-generator,break point register,comparator circuit,address decoder,one LED,three push-buttons,and two switches.

### 10.2.2 Test board front plate:



### Explanations on next page

### 10.2.3 <u>Designation of Switches and LED</u>

LED: The LED lights up when the circuit of the test board is waiting for a break point.

Break push-button active: An already set break point is reactivated. NMI push-button: Direct release of the break point; jump into the engineering panel program (NMI = NON MASKABLE INTERRUPT , high priority interrupt).

Push-button Rdy: Ready Simulation. If the CPU stops because false addresses have been entered, it can be started again by pressing "N" and the Rdy push-button.

Debug switch: Switch in lower position, jump into the engineering panel program.

Diagnostics switch: Switch in the lower position, jump into the PC diagnosis program (only for version 3).

### 10.3 <u>States</u>

### 10.3.1 Normal NC Operation

Switch S3 on the CPU is in upper position, the test board is not scanned. This operating mode <u>must</u> be used in normal operation.

### 10.3.2 Deactivated Engineering Panel Program

Switch S3 on CPU 03100 in lower position, debug switch in upper position. The engineering panel program does not scan the operator panel under these conditions.The control jumps into this state after PORESET (power clear). When a break point is reached or if the NMI button is activated,the control jumps into engineering panel program hold-state.One can return from holdstate into the system program with G LF.If switch S3 is in lower position but the test board is not connected,the CPU goes into a stop loop and the red LED on the CPU lights up. Remark: In this state, in the data input operating mode, after the startkey "Control in Action" has been activated, read data type such as TE,%, etc, are not displayed.

### 10.3.3 Activated Engineering Panel Program

Switches S3 and Debug in lower position. The jump into this program can be made by pushing the debug switch into lower position after PORESET, or by activating G LF during engineering panel program hold-state. In this state, both system and engineering panel programs are running. The operator panel has available only keys for the engineering panel program. The remaining 18 characters of the lowest line are scanned by the engineering panel program in the display. STEP-instructions, all registers display, and register changes are not possible.

#### 10.3.4 Engineering Panel Program - Hold-state

Switches S3 and Debug in lower position. The jump into this hold-state can be made by pushing the debug switch into lower position before PORESET or during the triggering of a break point. A jump to hold-state occurs during the triggering of a break point, even if the debug switch is in upper position.

The system program is inactive during hold-state; only the engineering panel program runs. This is indicated on the operator panel through the "Program runs" LED. The complete display and operator panel are available for the engineering panel program.

EPROMs 31 and 32 from the system software must be present on the 03200 PC board.

In this operating mode, signal "NC Rdy", and thus also ready 1 and 2, are revoked.

#### 10.4 <u>Functions</u>

- Display and change memory contents (changes in RAM range only)
- Display and change register contents
- Set break points (program addresses, as well as addresses in data range)

- Start and stop program sequence
- Single-step operation with display

#### 10.5 Sequence

#### 10.5.1 Engineering Panel Activation

The test board is scanned when the S3 switch on the CPU is in lower position. The engineering panel program is activated with the debug switch of the test board.Data (1 word) can be displayed and changed, or break points can be set and activated in engineering panel mode, while the system program is running. In engineering panel program hold-state, the CPU runs in a loop in the engineering panel program and waits for inputs from the operator, whereby all interrupt levels are locked out. The hold-state can be (Stop via NMI interrupt, triggered by:

- Arrival at a break point

at random positions in the program)

- NMI key activation

The hold-state is indicated by the "Program runs" LED.

When stopping via an interrupt (NMI or single-step), all registers are stored, so that the program may be restarted from the same position.After stopping through an interrupt, all CPU registers can be displayed or changed.

The program can be restarted with a GO or STEP instruction.

Hold-state "H" :

- The CPU waits for input instructions
- All interrupt flags in the CPU are cleared (interrupts locked)
- in the lowest level (B) - The CPU can be within the NMI level , or
- Display in the register page:

CPU runs on NMI or Single-step level, IM = 80 XX --> the page shows the actual register state or

IM = OO XX -----> CPU runs in the lowest level.The page does not show the actual register state, but that of the last break point or single-step. Single-step operation is not possible.

### 10.5.2 Display

If the engineering panel is active ("Activated Engineering Panel" state), it can control the display like in normal operating mode.Only the last 18 characters of the lowest line are controlled by the engineering panel program, and are used as input-feedback and output line.In the hold-state the entire display is available for showing the contents of the registers.

### 10.5.3 Break Points

If the processor should be stopped at a particular address, when the program is in ROM range, this must be accomplished through a hardware comparison (i.e. the CPU is stopped via an interrupt when the desired address appears on the address bus). Since CPU 8086, for higher speed, has a 6 byte buffer memory used as "instructions queue", the reading of an instruction is not identical to its execution. Depending on the length of an instruction, the stop 1-5 instruction can thus be issued too soon. If the instruction queue is empty, i.e. if several instructions with very short execution times were issued just before, the CPU will only be able to stop on the immediately following address. For this reason, it cannot be established after an NMI, whether the processor stopped before or after the desired address, i.e. it is not possible to make an exact stop of the execution of an instruction by subsequent processing of a queue. Example of error possibility :

Program sequence

XOR AX, AX JMP MARK2

MARK1:

Break point ----> MOV AX, VAR1

MARK2

The processor fetches the MOV instruction before it executes the JMP instruction --> the processor stops, even though the MOV instruction is not executed in the current program sequence.

From the stop address displayed, the operator can determine whether or not the processor has stopped on the desired address.

If the break point was set to an address within data range, the processor will always stop after the execution of the instruction.

Hereby, it is not only possible to stop on certain data addresses, but also certain data contents.

The break point register can be set while the program is running. The NMI enable bit can be set directly, with the set or break active key. This activation via the break active key is also scanned continuously during normal program sequence. A break point can thus be activated, even if the debug switch is not in lower position. The resulting interrupt stops the running program and starts the engineering panel routine. Engineering panel instructions can now be entered (even without resetting the debug switch).

10-8

To prevent the break points from being masked, the NMI (non maskable interrupt) is used. This NMI can also be triggered by hand, via the NMI key on the test board, independently of the setting of a break point. The NMI is stored with a flip-flop, which is cleared after the execution of the interrupt service routine. After the stop of an NMI interrupt, the lowest line of the display shows the program address (CS and IP register) of the next instruction to be executed.

#### ...6 Engineering Panel Instructions

Each instruction must be terminated with the LF key. Bracket [...]-----> instructions may be omitted.

10.6.1 Display and Change of Memory Content



- After M is entered, the address (segment and offset) of the last M-instruction is outputted automatically, and the contents of addressed data word are displayed.
- A segment address can be changed by entering **S** and typing in the new address.
- An offset address can be changed by entering **L** and typing in the new address.
- The data word can be changed with X and the input of new data. (Input as word: high byte - low byte)
- Incrementing with key by 2, and
   Decrementing with key by 2,
   are followed by the display of the next or preceeding word;
   changes can then be executed, as described above.
- The M-instruction is terminated automatically, if any other than the function keys mentioned above is activated, or through the input of the T LF end-instruction.

When entering new offset and segment addresses, care should be taken, that no address which the processor is unable to access due to its hardware, be created (the hardware does not give the "Ready" signal to the CPU). In such cases, the CPU stops and must be restarted by pressing the N and Ready keys.

### 10.6.2 Display and Change of Register Contents

- Display of all registers



- Register display and change



Register number

e.g. X C LF old value is displayed input of new value LF.

After the X-instruction is entered, the register designation appears, and the contents of the register are in the input line. After this, the contents of the register can be changed and/or it can be incremented to the next register. Incrementing: LF.

Register number:

AX	:	0 20	Accumulator		
BX	:	1	Base register		
CX	:	2	Count register		
DX	1 Park	3	Data register		
SI	e 1935	4	Source index		
DI	:	5 ₁₀ 10	Destination index		
DS	:	6	Data segment		
ES	:	7	Extra segment		
SS	. 🗧	8	Stack segment		
SP	all and	9	Stack pointer		

BP	:	.0	BASE pointer
FL	:	• 1 · ·	Status flag
IM	:	.2	(Interrupt mask register)
CS	1.2	.3	Code segment
IP	and the	•4	Instruction pointer

## 10.6.3 Break Point Setting

The stop address is composed of segment- (code or data segment) and offset address. When setting the break point to a data cell, it must be stated whether the stop should be on a byte, and whether on -WRITE- or -READ & WRITE-. The break point can be activated with the SET instruction or by pressing the "Break active" key.



After the break point is reached, the comparator circuit becomes <u>inactive</u> automatically.

F	A	LF
F	LF	

Display of break points
and control bits

Beactivation



(A ≘



10-11

Meaning of B, W, A, D

- B Must be entered if the break point should be activated on a byte address.
- W The break point is activated only when writing; when W is omitted, it stops at read and write.
- A Break point activation
- D When D is entered, the data content at which the break point should be activated, must be entered under K.

The sequence order B, W, A, D must be respected during input.

Input under K :

Byte	K	
Word	K	
Word low byte	K	
Word high byte	K	

10.6.4 GO Instruction -Start Instruction for Simulation

G LF

#### Loop counter

The engineering panel routine is exited, and the CPU continues from the position where the NMI interrupt occurred or, if the instruction pointer (and perhaps the code segment register) was changed, from the corresponding program address. The CPU will hereby start, and will continue running until the break point circuit generates another NMI.

With the aid of a loop counter, the number of break point loops which should be executed in sequence automatically, can be selected. ( $\emptyset$  = 1 loop). The CPU makes the final stop and the stop address is displayed only after the count-down of this loop conter. If no break points are set, the CPU runs until the NMI key is activated.

If, after **G LF** instruction, the debug switch is pushed from lower to upper position, the system will again be in "Engineering panel program deactivated" status.

10.6.4 Single-step Instruction - CPU runs in single-step operating mode





Step Counter

Within the program, the CPU starts at the point where a NMI interrupt occurred, or at an address newly written into the IP (instruction pointer). The CPU stops after each instruction execution, and can be sequenced manually with the LF key.

The number of steps (0-255) can be selected with the step counter; these steps should be executed automatically. (0 = 1 STEP)

After each step, respectively after the execution of the preselected number of steps, the current program address is displayed automatically (= the address of the next instruction to be executed).

The display shows:

CS = ---- IP = ----

If an R-instruction was issued before the step instruction, then the register page is up-dated after each step, automatically.

Error Possibilities in Single-step Operation :

In single-step operation, if a high-priority interrupt appears before the execution of an instruction, the CPU goes onto the higher level, and before executing the 1st. instruction, it executes the single-step interrupt routine.

Interrupt

higher level CPU stays on higher level

Test Program in Single-step Operating Mode

If an "IRET" instruction should be executed in single-step mode, and should an interrupt arrive at this point with higher priority than the return level, the processor will jump into the new interrupt routine, and the single-step interrupt routine is <u>not</u> executed.

When testing with single-step, if at all possible, the other levels should be masked !

#### Example:

Interrupt locked by setting a mask with:



IM (=interrupt register,see 12.7) Old content is displayed New value input

www.dbautond

0.7 <u>INPUT LIST</u>	anatyles				
CIFAD KEY (CIFADS LAST CHADACTED)	3	 [2]			
TNSTRUCTION TRANSFER		CA (FCH)	n /*/)		
HEX NUMBERS (=NR_ FOR	REGISTERS)		<u></u>		Jan
AX 0	Strain I	ם	59°	S.S.S.	8
BX	Souther D		LEADING TED WHEN	ZEROES MAY B N ENTERING NU	E OMIT- Imbers
2 CA 3 DX	ALOT I	2	(EXCEPT	FOR "K" INST	RUC-
4 SI		5	TIONS)	el ^{er} .	. 3
5 DI 6 DS	8	5			
7 ES		Ž			
8 SS	Sec. 1	8			
A BP		ופת רי			
B FL	M ¹⁰				
D CS		<u>ו</u> 1 ו	•		
E IP	. 6				
- 10 F	and the second sec	្រា			
INSTRUCTIONS:	10	105		205	
* GO (LOOP COUNTER $\phi = FF$ ) * STEP (LOOP COUNTER $\phi = FF$ )	S. I				
NEXT STEP		ין (אַאַן ו ניין ניין ו			
DISPLAY MEMORY			· · · · · · · ·		
CHANGE SEGMENT	100				
CHANGE DATA WORD	and the		XXX 🖸	, de	
INCREMENT OFFSET (+2)	25 ⁰⁰				
CLEAR MIM DISPLAY ADDRESS	S	<u> </u>	N & SIMI	ULTANEOUS RDY	SIMULA-
	C		TION AF	TER ERRONEOUS	ACCESS
* DISPLAY ALL REGISTERS					
* DISPLAY/CHANGE REGISTER	No.S		E (KI	EGISTER NR. (	ABOVE)
NEXT REGISTER	Caro .	e G			S
ACTIVATE BREAK POINT	Sauto		(A)		
SET BREAK POINT, ADDRESS + CONTR	OL		(L) XXX		
SET BREAK POINT, DATA COMPARATO	R		_	s ^a L	- BREAK P
BYTE	2	K ××			- WRITE PI
WORD MASK HTCH BYTE	No.N		<u> </u>	ġ	DATA BY
WORD MASK LOW BYTE	- officers				
END OF "DISPLAY MEMORY", "CHANGE	REGISTER		(U)		
MUUL" ERROR CHARACTER ON DISPLAY	A ^{1,0}	T.			
Auren Au	44				
	8				
	NO ^N				

Character	ISO-Code (8 Bit)	Character	ISO-Code (8 Bit)
7	A5	e	CØ
LF 👌	ØA 🔏	A	41 8
CR	8D	B	42
•5 ⁵⁶	3A 🚽	C	C3
*	<b>A</b> A	D	44
ONIL NUL	୍ ବୁଷ୍	E	C5
Space	AØ	F	C6
!	21	G	47
**	22	н	48
2	× A3 🔬	I	C9
\$ 20	24	J	CA CA
8	A6	K	48
•	27	L	
<u> </u>	28	M	4D
>>	A9	N.	41
+	2B	0	Cr Sd
;	AC C	P	50
	2D	N Y	
	ZE C	ĸ	52
1.55	AF	S	33
0	30 ~		55 D4
	BI		55
<u>2</u>	B2		D7
3	33	S. W	DP
4	B4	A V	50
5	35		54
<u> </u>	30	<b>1 4</b>	
	B/	1 <i>1</i> , <i>s</i>	50
8	B0	7 8	
9	39	-'	60
à 🦻	BD 3C	<u>\$</u>	d'
	SC BD	Nº.	No.
=	BE		R.
2	JE V	il i	NO.
÷		11	-75

ISO Code with Parity Bit (HEXA Code)

The information for user programs (PP and SP) in RAM memory 03210 or 03260 is stored in ISO code with parity bit. The ISO code is a subset of the ASCII code, but with parity bit, e.g. letter R in ISO code = D2; R in ASCII = 62.

### 10.8 NC Address Lists

## 10.8.1 Overview for Versions 0 and 2 :

	ADDF	RESS	na toma	
	Start	End	Ward Contraction of the second second	Board
	00000 01800 0E000	017FF 057FF 0FFFF	NMOS RAM CMOS RAM PERIPHERY	03100 03210 03310
		ANNI-BOUL	free	03320 03400
	20000	2FFFF	PC	03800
	Jone	3FFFF	PC	doautorna
A.M.M.	40000	4FFFF	TEST	03220
	tomatkart		free	
4 and I.	60000	6FFFF	EPROM	anna. Char
	70000	7FFFF	EPROM	03200

# 10.8.2 Overview for Version 3

ADDRESS			
Start	End		Board
00000	017FF	NMOS RAM	03100
01800	057FF	CMOS RAM	03210
0E000	OFFFF	PERIPHERY	03320
10000	10FFF	VIDEO INTERFACE	03810
11000	1FFFF	free	
20000	2FFFF	PC	03800
ar ward in the	3FFFF	PC	4
40000	2	and a second sec	2
101	4FFFF	TEST	03220
-MANIGOR		free	
60000	6FFFF	EPROM	03200
70000	7FFFF	EPROM	

## 10.8.3 Board 03 200 (Operating System)

Address subordination, EPROM designation, locations

Start Address		N.	К-	Plug location	
		End Addre	ss Words	H	I L
	60000	61 FFF	0-3	D02	I D01
	62000	63 FFF	4-7	D04	D03
	64000	65 FFF	8-11	D06	D05
	66000	67 FFF	2 12-15	D08	D07
	68000	69 FFF	16-19	D10	D09
	6A000	6B FFF	20-23	D12	I D11
	6000	6D FFF	24-27	D14	D13
	6E000	6F FFF	28-31	D16	D15
	70000	71 FFF	32-35	D18	D17
	72000	73 FFF	36-39	D20	D19
	74000	75 FFF	40-43	D22	I D21
	76000	77 FFF	44-47	D24	D23
	[∞] . 78000	79 FFF	48-51	D26	l D25
	7A000	7B FFF	52-55	D28	I D27
	7C000	7D FFF		D30	D29
	7E000	7F FFF	60-63	D32	D31

#### 10.9 Access to PC with the NC-Engineering Panel

Note: all addresses are in hexadecimal system,without designation of index "H".

```
10.9.1 Converting PC to NC Address
```

PC address times 2

000ر20 +

e.g. PC address EAOF

= system data word 7, 2nd. byte

EADF • 2 = 1D41E+  $\frac{20000}{3D41E}$ 

### 10.9.2 Reading and Setting of Inputs and Outputs of the I/O Boards

- 1. Test board 03220 must be connected.
- 2. Activate test board 03220 with switch S3 (int) on CPU 03100
- 3. NC ON (PORESET)
- 4. Debug-switch on the test board in lower position
- 5. Set break point Edit key

F 3000-D418-A-LF

6. Input for display

M LF

7. Read address

S 3E00 LF LOO10LF

8. Display shows:

M*3E00, - 0010, - FF__

Display of input information

e.g. Machine control panel on input byte 8 through 13 Operating mode selector switch to reference point FFE_ Feed rate override switch to 120% FFE8

9. Input of next input byte

Activate key 🖉 🛉

Display shows:

M*3E00_ - 0012_ - FF

Display of input information ______, byte 9

* See remarks under 10.4.3
10. Display of further input bytes

with key 🗼 with key 🚺

increment decrement

11. Outputs setting

After an address is read, the corresponding outputs can be set.

Input X _ _ LF

e.g. XFFLF, i.e.

Output is only possible if jumper A-B (standard strapping) is present on coupling board 03800.

- 10.9.3 Direct Access of the I/O Boards through PC Interface without PC, with the Aid of the Engineering Panel
  - 1. Board CPU PC is disconnected
  - 2. PC coupling 03800 strapping socket P1 ,change from jumper 3-14 to 2-15.
  - 3. The test board must be connected, and the debug switch must be in lower position.
  - 4. CPU 03100 switch S3 should be on Test
  - 5. Hardware reset (only engineering panel program possible)
  - 6. Input for display

MLF

7. Read address

S 3E00, LF L 0010, LF

see the remark

*Remark to the address shown in the example: Address 3E00 - 0010 corresponds to input word 8,i.e. machine control panel is on I/O board nr. 2, strapping socket X1 jumper 1-16. For other addressings,see "address range I/O",section 10.8.5. 8. Display shows:

M_3E00, - 0010, - FF ___

Display according to input (see 10.4.2.8)

9. Outputs setting

The corresponding outputs can be set after the address has been read Input  $X _ LF$ 

e.g. FF ,all outputs of the output byte read.

# 10.9.4 Display ISTACK with the Engineering Panel

With PC 130 W, the display of ISTACK with the aid of the engineering panel program is only possible with the new system program.

Old: Edition 4/81 3WA12

Display not possible

- New: Edition 10/81 3WA13 Display possible
- 1. PC-switch to Stop, to prevent clearing the ISTACK
- 2. NC OFF

3. Test board 03220 must be connected

4. Activate the test board with switch S3 on CPU 03100

5. Test board debug switch in lower position

6. NC ON (PORESET)

7. Set the break point

F 3000-D418-A-LF

8. G LF

9. Input for display

M LF

10. Read address

S LF L LF

e.g. SD214 S3D75 LF L0008 LF

11. For other addresses, see lists in section 10.8

# Notes to the PC

- 11.1 Prerequisites, Settings 130 WA
- 11.2 Prerequisites, Settings 130 WB
- 11.3 Function Blocks
- 11.4 Program Examples
- 11.5 Testing Aids
- 11.6 Trouble-shooting in the PC
- 11.6.1 ISTACK
- 11.6.2 BSTACK
- 11.7
- 11.8 PC Lists
- 11.8.1 Memory Map 130 WA
- 11.8.2 Address List 130 WA
- 11.8.3 Memory Map Internal RAM 130 WA/130 WB
- 11.8.4 SD Range 130 WA/130 WB
- 11.8.5 Memory Map 130 WB
- 11.8.6 Address List 130 WB
- 11.8.7 Instructions Sets

# 11.1 Prerequisites, Settings for 130 WA

# 11.1.1 Assembly

11.1.1.1 PC 130 WA, central processing unit 6ES5-921-3WA With receptacle for one EPROM module,for a maximum of 8k instructions for the user program.

This board also contains a RAM memory for 0.5k instructions for the user program.

EPROM module with memory for 4k instructions Type 370 with 4 EPROM 2716 Type 820 with 2 EPROM 2532

EPROM module with memory for 8k instructions with option N2O Type 371 with 4 EPROM 2532 Type 820 with 4 EPROM 2532

The control is shipped from the factory with an EPROM containing a program, see section 11.4 .

#### Generalities concerning the memory modules

Because the EPROMs on memory modules 370 and 371 are soldered, the covers for the retraction of the modules can no longer be used. For this reason, we have available a different cover with slide :

Order number for the cover: C79451-A3079-C258 Order number for the slide: C79451-A3079-C259 Ordering address: GWK

The covers are also used with type 820. The assembly of the cover and slide to the memory module is explained in a diagram.



11.1.1.2 PC coupling board 03800 A

11.1.1.3 PC memory board 6ES5-350

For option N22 w/o EPROM memory For option N23 with EPROM memory, 4k instructions For option N24 with EPROM memory, 8k instructions For option N25 with EPROM memory, 8k + 4k instructions

# 11.1.2 Settings

11.1.2.1 CPU 6ES5-921

No settings are necessary on the board. The following jumpers are present: Basic board U1: jumper A-B, C-D, K-L, P-Q Complementary board U2: jumper A-B, D-E EPROM module type 370/371: no jumpers

Туре 820	Wire Jumper				
	W1	W2			
2k 4k instructions <b>6k</b> 8k instructions	open closed open closed	open open closed closed			

- 11.1.2.2 I/O boards 03400, 03410, 03421, 03450, 03460 Strapping according to the user program, see section 9.
- 11.1.2.3 S5 I/O boards 402-3, 445-3, 444-3, 432-3 Strappings according to user program, see section 9.

#### 11.1.2.4 Coupling board 03800A

No setting is necessary under normal circumstances. The PC outputs are not locked when there is a fault in the NC.Should the **customer** want the PC outputs locked,jumper A-B must be removed,and jumper B-C must be inserted.

The controls are shipped with strappings 14-3 (command output lock for EXTENSION UNIT) on socket P1:

Socket P1

9	С	0	8	2 ²
10	0	•	7	St St
11	0	0	6	.offic .offic
12	0	•	5	and the second sec
13	0	0	4	
14	0	-0	3	BASP Extension Unit
15	0	0	2	for test (see section 12)
16	0	0	a.	

11.1.2.5 Memory Board 6ES5-350

The memory board provides for a maximum of 12k words for the EPROM Range and 4k words for the RAM range.

OPTION N22: RAM 4.5 k EPROM 8 k

Strapping socket Nr. 4 Nr. 19 Nr. 26 95402300 991111100 . . . . . . . . . 0000 000 EPROM EPRON RAM 000000000 . . . . . . . . . ...... OPTION N23: RAM 4.5 k EPROM 12 k Strapping socket Nr. 4 Nr. 19 Nr. 26 2122200 827<u>7</u>2220 12220 . . . 000 . . . . . . . . . .... ...... OPTION N24: RAM 4.5 k EPROM 16 k Strapping socket Nr. 4 Nr. 19 Nr. 26 9575550 0 T 0 0 T 0 000 0000 0 0 ...... . . . . . . . . . . . . . . . .

OPTION N25: RAM 4.5 k EPROM 20 k

socket Nr. 4	Nr. 19	Nr. 26
0 95723290 VO	2222229	*******
••••••••••••••••••••••••••••••••••••••		
<b></b>		
- C - C - C - C - C - C - C - C - C - C	~~~~~	

# 11.1.3 Function Test of the PC

#### 11.1.3.1 PC-CPU :

After the NC and PC are switched on, the green LED of the PC-CPU must light up.If the red LED is lit instead, it could be due to the following causes:

- The switch of the PC-CPU is on Stop
- The PC RAM is in undefined state

Remedy: execute cancel 0, see section 4.3

- Memory addressed incorrectly (Address coding)
- Hardware fault of the PC-CPU
- No battery voltage
- Cycle time exceeded (watch-dog timer)

Check of the battery in the power supply Further information in the system data.

These can be read out with PG 670, PC program correction (diagnostic program only for 3T/3M3),or the NC engineering panel.If at power-up,the PC is in Stop state,the NC-CPU 03100 will also go to Stop state. See section 4.2.

If the PC goes to Stop state during operation, the NC displays "PC fault".

LED significance:

If the red LED lights up,the PC is running in a stop loop.Access is possible with PG 670,PC program correction,or NC engineering panel. If the green LED is lit,the PC is running in cyclic operation. If both red and green LEDs are lit,the PC is running in the restart branch.

If no LED is lit, the voltage to the PC is lacking or there is a hardware fault in the PC.

#### 11.1.3.2 Coupling board 03800 A:

Red LED "NC Fault": this LED indicates NC faults, but only if the A-B jumper is removed (outputs locked).

Red LED "Periphery Fault": this LED is lit continuously if no extension unit is connected. If there is a connected extension unit, it only lights up during malfunctions.

# 11.1.3.3 User Program

If no user program is yet available, or if the one available presents problems, the NC can be started with the program shipped with the control.

The RAM memory board 6ES5-340 can also be used for the testing of the user program.

The start address must be set properly:

Location nr. 51

All					All	•		
16	0	0	0	0	0	0	0	0
					16	8	4	2
1	0	0	0	0	0	o	0	0
	3.				f	ъ.	×.	

Jumper for the 16k memory board Valuation in k-words

Jumper for the 8k memory board

Location nr. 71 (masking) without jumpers.

# 11.2 Prerequisites, Settings for 130 WB

The 130 WB has been used instead of the 130 WA, since the middle of 1983.

# 11.2.1 Assembly

#### 11.2.1.1 PC 130 WB, CPU 6ES5 921-3WB

With receptacle for 2 EPROM modules, respectively for a maximum of 16k instructions for the user program. The board also contains a RAM memory for 2.9k instructions for the user program.

#### 11.2.1.2 EPROM modules

Basic unit 1 module for 4k instructions Option N32 1 module for 8k instructions Option N34 2 modules for 2 x 8k instructions

Type MS820 is used: with 2 EPROMs 2532 for 4k, with 4 EPROMs 2532 for 8k.

S5 modules 371 and 373 can be used for type 820. Module 370 may not be used.

The 1st. module must be plugged into the upper receptacle.

11.2.1.3 PC coupling board O380CA, as for the 130 WA

# 11.2.2 Settings

# 11.2.2.1 CPU 130 WB

No settings are necessary on the board. The following fixed strappings may not be changed. Basic board U1 Type O1101 (CPU) Single jumper G-F closed Single jumper K-L closed

Compler	nentary	board U2	Type 01201 (memory)
Single	jumper	W1	open
п	J.	₩2	open
" S	11	W3	closed
H. S.	11	W4	open
ŦŤ	11	W5	closed

#### 11.2.2.2 EPROM module

Fixed strapping type 820, see 11.1.2.1 Modules 371, 373 have no strapping.

11.2.2.3 I/O boards, O3800 A, as in section 11.1 Remark: memory board 6ES5-350 is not included in the shipment.

11.2.3 Function test for the PC

# 11.2.3.1 PC-CPU

see section 11.1.3.1

The following distinguishes the 130WB from the 130 WA :

an additional "Restart" push-button.

The entire RAM memory can be cleared with this key (as with "Cancel O") The following sequence must be respected :

- 1. PC switch on Stop
- 2. Press the restart key, and simultaneously the hardware reset on the power supply
- 3. PC switch to Operation
- 4. PC switch on Stop
- 5. PC switch to Operation, afterwards the green LED must light up
- 6. Hardware reset, so that the NC is restarted

11.2.3.2 Coupling Board, see section 11.1.3.2

#### 11.2.3.3 User Program

If no user program is available yet, or if there are problems with the existing user program, the NC can be started with the program shipped with the control.

The RAM memory board 6ES5-340 may also be used to test the user program.

The start address must be set correctly : Location nr. 51

		12						
16	00	ିତ	0	0	0	0	0	0
	30				16	8	4	2
<u>_</u> 1	0	ο	0	0	0	0	σ	0
				274	ŧ			

Jumper for 16k meory board Valuation in k-words

Jumper for 8k memory board

Location nr. 71 (masking) without jumper.

Board 5-350 for 4k RAM :

RAM		1	2	3	4	5	6	7	8	and the second
Range 1	16	0	0	0	٩	०	q	0	0	Valuation in
Location	à.	ĺ			16	्रे8	4	2	1	k-words
Dr. 4	1	0	· 0	0	ď	<u>`</u> 0	لم	0	0	and the second
Callon		16	15	14	13	12	11	10	9	"autor"

# 11.3 <u>Function Blocks</u>

# 11.3.1 Overview

Block Brief Description Name

0	ALARMST	Alarm controlled program
11	EINR-DB	Set up data block
12	HILFSFKT	Auxiliary function
15	AK3:AUT	Sequence control 3, automatic
16	BLOCK-TR	Block transfer
20	M-DECOD	M-decoder
21	S-UEBERG	S-transfer
22	DATANNC	Data transfer to NC
24	S-DECOD	S-decoder
35	SER 130W	Service unit 333-OBA
36	VERZOEG	Delay
37	STATUS 1	Status display 1
40	RI-AUSW 1	Direction selection 1

#### Remark:

The function blocks for other systems have the same numbers and designation.For this reason, it is important to use only the right function blocks for System 3 (they are stored on a special diskette for System 3).The differentiation is done through the library number, respectively 130 WA/130 WB.

The same function blocks are used for versions 2 and 3.

# 11.3.2 Function Blocks Designation

		P 7 1 2 0 0		$0 - \mathbf{A} = 0$
For SINUMERIK	. tor	30 ⁰	30	
Order nr. for SINUMERIK (always B) -	>>	WWW.ODOL		44
Dept. E3	<u>~</u>			8
SINUMERIK system 1 for System 3 Versions 2+3	utorensko.			en de la companya de
Function block nr				J [ ]
Version A for 130 W	mashad	and and a second		Chard Hoo
		walter .		
Edition		and the second s	and Co	e de la companya de la
anathe a				
	P P	7 1 2 0 0 -	B 3 1 1 2	- A - 5
These designations are fixed for Sys- tem 3,version 2+3	Ľ	A COLOR		and the second
Function block 12	- Nor		×	stor
Edition 5	MORTIO	- and Chico		N ^o

Note: The function blocks for version 4 have different designations, and are not functional in versions 2 and 3. 11.3.3 Function Blocks Description

FB12: HILFSFKT Auxiliary functions

The task of the "Auxiliary functions" function block consists of: bringing the PC in a defined reset state,

supplying signals used in other function blocks,

11-13

exchanging the interface signals for the signal exchange NC/PC between flags and data block 9, and

setting up data blocks 2 and 9 in the RAM range of the PC,at Start.

When the system is started (power-up or switching from "Stop" to "Operation" on the front plate of the CPU), the interface between NC and PC is brought in a defined start state. In addition, depending on the type of start, signal "Reset state" (at start) or "New start" (at start with reset) is issued.

If not already present, data blocks 2 and 9 are also set up. Signals "Reset state" and "New start" are issued as pulses.

During <u>normal program execution</u>, the signals corresponding to the NC are exchanged by the FB12 between the flags (FBO-17) and DB9, because the signal transfer between NC and PC occurs over DB9 (see diagram).

Detailed descriptions of the function blocks can be found in the publication "Function Blocks for S5-130 W, SINUMERIK System 3".

# 1. Data blocks

DB	0	Address lists		
DB	1	Variables for service unit 333		
DB	2	Variables for function blocks		
DB	9	For data exchange PC/NC		
fro	m DB 10	*For user variables,e.g. states of a	control sequences,et	с.

# 2. Flags

Flags O through 17 are reserved for the PC/NC interface. Flags 188 through 255 are reserved for function blocks.

# 3. Counters

Counter O is used for the code conversion BCD/binary.

4. Timers

Timers D and 1 are used for function block "Auxiliary functions" data to NC.

# 5. Function Blocks

Function blocks 0 through 99 are used as standard FBs.

*DW O through 9 of the data block must be reserved for parametering of sequences.

# 11.3.5 Flag List

Flag Byte       Function         0       Signals PC to NC         9       Unassigned         11       Unassigned         12       Signals NC to PC         17       Signals NC to PC         18       Individual signals         220       Individual signals         221       Turret TC actual position         222       Turret TC command position         223       Transfer flags for         224       function block DATANNC         225       Output signals for         236       Auxiliary signals         240       Reserved for PSP inputs         243       Status flags         244       Status flags         251       Input flags for scratch pad results	~ ×			
0       Signals PC to NC         8	KORNACHE F	lag Byte	Function	rathe on and the
9       Unassigned         12       Signals NC to PC         17       Signals NC to PC         18       Image: Signals NC to PC         187       Image: Signals NC to PC         188       Reserved for individual signals         220       Turnet TC actual position         221       Turnet TC command position         223       Transfer flags for         224       function block DATANNC         225       Output signals for         M-function       See FB20 M-DECOD         238       Auxiliary signals         240       Reserved for PSP inputs         241       Status flags         242       Input flags         243       Input flags for scratch pad results	www.gaue	0   8***********************************	Signals PC to NC	For more information consult the interface description for
12       Signals NC to PC         17       18         18       187         188       Reserved for individual signals         219       220         221       Turret TC actual position         222       Turret TC command position         223       Transfer flags for function block DATANNC         224       function block DATANNC         225       Dutput signals for M-function         238       Auxiliary signals         240       Reserved for PSP inputs         240       Reserved for PSP inputs         241       Status flags         242       Input flags         251       Auxiliary flags for scratch pad results	and	9   11	Unassigned	SINUMERIK System 3
18       187         187       individual signals         219       individual signals         220       221         221       Turret TC actual position         222       Turret TC command position         223       Transfer flags for         224       function block DATANNC         225       Output signals for         M-function       See FB20 M-DECOD         238       Auxiliary signals         240       Reserved for PSP inputs         244       Status flags         244       Status flags         247       Input flags         251       Auxiliary flags for scratch pad results	A CONTRACT	12     17	Signals NC to PC	e and a state of the state of t
187         168       Reserved for individual signals         219         220         221       Turret TC actual position         222       Turret TC command position         223       Transfer flags for function block DATANNC         225       Output signals for M-function         238       Auxiliary signals         240       Reserved for PSP inputs         244       Status flags         244       Status flags         251       Input flags         252       Auxiliary flags for scratch pad results		18		
188Reserved for individual signals219individual signals220221221Turret TC actual position222Turret TC command position223Transfer flags for function block DATANNC224function block DATANNC225Output signals for M-function237decoding238Auxiliary signals240Reserved for PSP inputs241Status flags243Status flags244Status flags245Input flags252Auxiliary flags for scratch pad results	Cardin -	187	<u></u>	and the second second
220221Turret TC actual position222Turret TC command position223Transfer flags for function block DATANNC224function block DATANNC225Output signals for M-function decoding238Auxiliary signals240Reserved for PSP inputs243Status flags244Status flags245Input flags246See corresponding function blocks247Auxiliary flags for scratch pad results	Mark Charles	219	Reserved for individual signals	
221Turret TC actual positionSee FB40 RI-Ausu (Direction selection)222Turret TC command positionSee FB22 DATANNC (Data to NC)223Transfer flags for function block DATANNCSee FB22 DATANNC (Data to NC)225Output signals for M-functionSee FB20 M-DECOD237decodingSee FB12 HILFSFKT (Auxiliary function)240 243Reserved for PSP inputsSee corresponding function blocks244 247Status flagsSee corresponding function blocks248 251Input flagsSee corresponding function blocks	3	220		24° 24°
222Turret TC command position(Direction selection223Transfer flags for function block DATANNCSee FB22 DATANNC (Data to NC)225Output signals for M-function decodingSee FB20 M-DECOD237decodingSee FB12 HILFSFKT (Auxiliary function240 243Reserved for PSP inputsSee corresponding function blocks244 243Status flagsSee corresponding function blocks243Input flagsSee corresponding function blocks	2	221	Turret TC actual position	See FB40 RI-Ausw
223Transfer flags for function block DATANNCSee FB22 DATANNC (Data to NC)225Output signals for M-function decodingSee FB20 M-DECOD237decodingSee FB12 HILFSFKT (Auxiliary function)238Auxiliary signalsSee FB12 HILFSFKT (Auxiliary function)240 243Reserved for PSP inputsSee corresponding function blocks244 247Status flagsSee corresponding function blocks248 251Input flagsSee corresponding function blocks255Auxiliary flags for scratch pad resultsSee corresponding function	20 ⁵¹	22	Turret TC command position	<ul> <li>(Direction selection)</li> </ul>
225Output signals for M-function decodingSee FB20 M-DECOD237decodingSee FB12 HILFSFKT (Auxiliary function)238 239Auxiliary signalsSee FB12 HILFSFKT (Auxiliary function)240 243Reserved for PSP inputsSee corresponding function244 244 244 245Status flagsSee corresponding function blocks248 251Input flagsSee corresponding function blocks	diation	23	Transfer flags for function block DATANNC	See FB22 DATANNC (Data to NC)
238 239Auxiliary signalsSee FB12 HILFSFKT (Auxiliary function)240 243Reserved for PSP inputs244 244 247Status flags247 248 		25	Output signals for M-function decoding	See FB20 M-DECOD
240       Reserved for PSP inputs         243       244         244       Status flags         247       See corresponding function blocks         248       Input flags         251       Auxiliary flags for scratch pad results	10 ¹⁰ 2	38 39	Auxiliary signals	See FB12 HILFSFKT (Auxiliary functions)
244     Status flags       247     Status flags       248     Input flags       251     Auxiliary flags for scratch pad results	2	40	Reserved for PSP inputs	www.chav
248     Input flags     function blocks       251     Auxiliary flags for scratch pad results     .	2	44   47	Status flags	See corresponding
252 Auxiliary flags for scratch pad results 255	2	48 51	Input flags	function blocks
	2	52	Auxiliary flags for scratch pad results	and and a
	20012	>>	and the second s	
Contraction and charlon and charlon and charlon and charlon	- Matanton			

#### 11.4 Program Examples

#### 11.4.1 Factory Program

The PC is shipped from the factory with a PC program on the first EPROM on the EPROM memory module of the PC-CPU.

Use: to test the NC functions without a customer PC program.

Setting: The machine control panel cable must be connected to an input board. The strapping must be for the 0-5 input byte, i.e. X1 without jumpers.

Blocks used: the standard function blocks FB11 and FB12, the test block FB200, and one OB1.

EPROM Modules: The modules have the factory designation:

548	811	0061	03	² 4k	memory 🔊	
548	811	0062	03	8k	memory 🔊	
			N		- Alba 🖉	nn

When clearing the factory program to write the user program, a spare module should be used.Such a module may be found in the spare parts case.

Program Construction: 3 different programs can be jumped into by interrogating during PORESET:

<u>1. Program</u> with Axis Lock: At PORESET, the operating mode selector switch should be in "Data Output" position, and the SNS key should be pressed. All keys and switches of the machine control panel (except E-Stop) may be used.

Test of operating modes and programs without axis movement.

2. Program without Axis Lock: At PORESET, set the operating mode selector switch to "Reference Point Approach" and press the Sequence Number Search (SNS) key.All switches and keys (except E-Stop) may be used. Test of operating modes and programs with axis movement. Caution must be used with this program, because the signals E-Stop, Feed, and Control Enable are fixed by the PC program.

<u>3. Program</u> I/O-Simulator. This program is executed when not jumping in either program with or without axis lock. All keys and switches of the machine control panel, except Feed Hold/Start, Spindle ON/OFF and E-Stop, can be used.

Test of operating modes. Axis movement is possible only if the axis enables are connected.

Remark: only the I/O Simulator Program is available with EPROM modules in editions 01 and 02.

11.4.2 Structure of the Factory Program



11 - 17

11.4.3 Shipping Program Printout

BLOCK LIST

BLOCK TYPE	NUMBER	SYMBOL	LENGTH	LIBRARY NUMBER	
FB	11		116	P71200-B 3111-A-	1
FB	12		206	P71200-B 3112-A-	5
FB	200		85	382	
FV S	200		16		
OB CONTRACTOR	100		13		
711 FREE BLOCK	S				

0B1

SEGMENT 1		AG150A	
0000	:A 👌	F239.6	
0002	;÷	F251.0	
0004	:=	F251.1	
0006	ະວນ	FB12	
0008 NAME	:AUXI	LIARY FUNCTION	)N
000A	: 30 🔪	FB200	
DOOC NAME	:SERV	ICE -TE	
noof a	:BE		

11 - 19

FB 200

www.chautomati

					MB150		MB151 👌	
	SEGMENT 1					P.	No.	0 +B.t
	NAME :SER	۲ <b>۲۰</b> -۲ε			Sel Switch	Witth		
	DDDA	: UN	M 238.1	GST (Reset State)	DCBA		S	
	000C	: SPB	* PROG	S.			ND 15 2	
	DODE	:L	EWD		MB152	1.8-3		0+8-1
	0010	:T	NW15D		00.1	lode Se		FTAFT
	0012	:L	NB150		0.0.0.01	itch		91 W -
	001.4	:SRW	4.2					
	0016	:T	NB152					
	0018	:L	HB152		A = Axis lock			5
$\sim$ 3	OD1A	∃E ⊗	KHODD1	Position				
	001E	: IEF						
	0020		H 153.0	AF with axis lock				
. E	0022		MR152	. all all a				
		•1	KHODOF	Position [1]				
	0014	• 1 • 6	KNUUUL					
<u> </u>	0020	• 2 • 6	M 151 1	At who aris lock				
	0020 0005	•11	M 151 7	CNC	0044 H-A	THN M	5.3 8	<b>`</b>
	0025	•11	H 157.0	AL with avia look	0040	:5 M	5.3	<u>}</u>
	0010		-M-A	AF WILL BXIS TOCK	DDAE D-A	ti FN		
	0030	1350	= [] - M M 4 E 4 7	CNC 8 242	0002 0 4	T NH	n S	8
	0037	20	<b>H</b> 101.3	AF w/o avis lock	0070	•11 E	2 2	Program
	0034		-0.4	AL W/U GAIS TOOK	0072		2 2 Deadron	with
	0030	1348			0074		2 2 W/o	<b>&gt;</b> axis
	0038	:L	FAN		0078	10 E		lock
	0036		MWU .		0078	ALL E	2.3 lock	1
	0030	:L	EWZ		0074			1
	UUSE		MWZ /		0076		5 6	1
	OUAU	್ಟನ್	EW4				2.5	
	0042	<u>_</u>	MW4		0000			S.
	0044	SIL .	EBB		0087		2.4	250
	0046	:1	MB6		0084		2.0	24
	0048	:L	EB9		0088	.U E	2 7	
~	DD4A	:1	MBB		0080		2 2	
$\sim$	004C	:L	EBIO	Process with	0080		3.2	Ì
	OD4E	:1	MB7	Frogram with	008E	:U E	3.4	1
	0050	:L 🔬	EB11	1/0 21mulator	0090	:5 N	4.2	:
	0052	ः	MB9		0092	UN E	3.5	1 20
A. 19	0054	3E	MW1 2		0094	R H	4.2	
and the second second	0056	<b>11</b>	ANO		0096	:U E	3.6	
	0058	:L	MB15		0098	:5 N	4.3	
	005A	:T	ABZ		009A	UN E	3.7	
	0050	:L	MB16		0090	R H	4.3	
	DOSE	<b>:T</b>	AB8		DOPE ENDE	BE	100	
	0060	:L :	MB17				affic	
	0062	ः म् ्र	AB7					
	0064	:10	MB14 /					
	0066	зT	ABID					
	0068	:SPA	=ENDE					



This is an example of a program structure in which only the function block "Auxiliary functions" is used. In the cyclic program (OB1), this function block is called in a program block which follows the PB for axis-specific functions. This sequence is necessary, to ensure the shortest possible response time for the axis-specific signals.

The main task of the "Auxiliary functions" FB, in addition to the start routine, is to interchange the signals to be transferred between NC and PC, between the flag ranges O-17 and DB9.This transfer achieves the following:

- a) intermediate results are not transferred to the NC, and
- b) The logic processing of NC signals is independent of its position within the PC program.

DB9, necessary for the signal exchange, is set up automatically during system start.



11.4.5 Example of NC Oriented Function Blocks

NC oriented function blocks serve to simplify the transfer of data between the NC and the PC.

It is most convenient to combine all the NC oriented function blocks into one program block.

# 11.5 Test Aids

# 11.5.1 Input/Output Signal Image on NC CRT

The image of the interface between the PC and machine tool can be displayed any time, under ident. nr. 35 - 54.

Refers only to version 3, starting with software edition 02.

Input is not possible.

MDI SE-TE				Ś
PC INPUT SI	GNALS			
35		36		
B00 1111	1111	B10	00000011	
B01 1111	1111	B11	00000000	
B02 1111	1111	B12	00000000	6
B03 1111	1111	в13	00000000	P
B04 1111	1111 👘	B14	00000000	
B05 1111	1113	B15	00000000	
B06 1111	1111	B16	00000000	
B07 1111	1111	B17	00000000	
B08 0000	0011	B18	00000000	đ
B09 1000	0000	в19 в	00000000	ľ

I bytes 0-7 are not used; machine control panel connected to I bytes 8-10

e.g.

Ident.	пг.	37	and	38	B20	through	B39
11	S. H	39	and	40	B40	71	B59
H 3	11	41			B60	11	B63

All 64 input bytes are always displayed.

BOO corresponds to input byte O, etc. Input signal 1 (+24 V): Display 1 Input signal O : Display O Input board not installed: Display 1

MDI	SE-TE	autorne	auto
PC OL	TPUT SIGNALS		
48		49	
B00	0000000	B10	00000000
B01	0000000	B11	00000000
B02	0000000	B12	0000000 🧹
B03	0000000	🖉 В1З	00000000
B04	0000000	B14	00000000
B05	0000000	B15	00000000
B06	0000000	B16	00000000
B07	0000000	B17	00000000
B08	0000000	<b>B18</b>	00000000
B09	0000000	E19	00000000

Ident.	nr.	50	and	51	B20	through	B39
11	. <mark>н</mark> с [©]	52	and	53	B40		B59
" "	11	54			860	11	B63

All 64 output bytes are always displayed.

BOO corresponds to output byte O, etc. Output signal 1: Display 1 Output signal O: Display O Output board not installed: Display O

11.5.2 <u>PC Program Correction via NC</u> (see section 12)

11.5.3 Access to PC via NC Engineering Panel (see section 10)

# 11.5.4 Service Unit 333-OBA

# 11.5.4.1 Application area

Service unit 333 is a testing aid for the PC SIMATIC S5-130 W.

All the signal states of the device and program can be interrogated and changed rapidly with the help of this unit.

The service unit can furthermore, remain permanently connected to the PC, and be used for fast location of problems.

The following functions can be executed:

· Data output, time and counter values

- · Data input, time and counter values through data words
- Signal state display of inputs, outputs, flags (bit, byte, wordwise)

The service unit is connected to the PC **through** digital inputs and outputs.

The user program of the PC <u>cannot</u> be changed through the service unit, even if improperly operated.

# 11.5.4.2 Construction

The service unit consists of a plastic housing with operating and display elements on the front side. The two connecting cables (approximately 3 m long) exit at the bottom, and have a 48 pole connector, construction form F.

The service unit is provided with magnets, which allow it to be mounted on metal cabinets.

# 11.5.4.3 I/O Board

An S5 I/O board,e.g. 6ES5-482,with 16 inputs and 16 outputs must be used.This board is plugged into the location of the PG interface or that of an I/O board, and can be removed again during normal operation. The address selected in the program must be strapped properly on the addressing socket of the board. Program and Data Structure with Service Unit 333-OBA



1) FB 36 should be called if no user program is available.

FB 35 has to be parametrized by a program block, and must be called cyclically.

# 11.5.4.5 Program Example

for use with FB200, according to section 10.3.3

Address 32 of the I/O board is called in PB2.On board 482,pins 3-14,on strapping socket X1,must be jumpered.

# BLOCK LIST

BLOCK T	YPE NUMBER	SYMBOL	LENGTH	LIBRARY NUMBER
DB	35		64	3500
DB 🔊	2		10	
DV	35		14	
FB	11		116	P71200-B 3111-A-1
FB	12		206	P71200-B 3112-A-4
FB	16		78	P71200-B 3116-A-D
FB	35		464	P71200-B 3135-A-1
FB	36		19	P71200-B 3136-A-O
FB	200		85	382
FV	> 200		16	
OB	1		16	

697 FREE BLOCKS

0012 NAME :DELAY

:BE

0014

0B1	AG15DA	PB2	AG150A
LEN =16	ABS	LEN =10	ABS
SEGMENT 1		SEGMENT 1	
0000	:A F 239.6	0000	:L KB32 💰
0002	:= F 251.0	0002	:T F 8255
0004	:= F 251.1	0004	:JU FB35
0006	: JU FB12	0006 NAME	:SER 130W
0008 NAME	:AUX. FUNCTION	0008	:BE
000A 🔊	: JU FB200		
DOOC NAME	:SERV. UNITS		
000E	: JU PB2		
0010	: JUN FB36		

# 11.5.5 Service Unit 333 C

Service unit 333 C replaces the previously used unit 333-OBA.Certain functions have been extended in the new unit.

#### Functions

- Data output, time and counter values
- · Data input, time and counter values through data words
- Signal state display of intputs, outputs, flags (bit,byte or wordwise)
- Correction of program and step blocks

Construction .:



<u>Remark</u> : Currently, service unit 333 on the S5-130W does not function with FB 25.

# 11.5.6 <u>PG 670/PG 675</u>

For functions and handling, see the training manual for S5-130W and the operating instructions PG 670/675.

## 11.6 Troubleshooting in the PC

# 11.6.1 Interrupt Stack (I-STACK)

1	CON	ITROL Passch	BITS BSTSCH		ADRBAU	SPABBR	NAUAS	OUITT	Control bits
	NB.	N3	HB	NB	48	64		<b>**</b>	SD5
2	STOZUS X NB	STOANZ X UAFEHL	NEUSTA X MAFEHL X	VIEDAS *	BAT PUF X NB	18 18	BARB NB	BARBEND	SD6
3	ASPNPR KEINAS	ASPNRA X Synfeh X	KOPFN1 NINEU X	PROEND HIVIED "	MB MB	PADRFE BAPERL	ASPLUE SUNF	RAMADFE URLAD X	SD7
2 - 2 - 2 - 2 - 2 - 2 - 2 - 2 - 2 - 2 -	IN.	TERRUP	T STA	СК	h.	A.C.		And M.C.	I-STACK read
•	BEF-REG BEF-REG BET-ST	: 7000 : E803	SAZ Rel	: D1 NR.: -SAZ:	SDE	88-ADR : 88-MR. :	8000		mathad
	AKKU1 :	£388	AKK	2: E398		doaute		, S ⁱ	
5	Result	t flag:	ANZ	ANZO OV	FL CARRY	ODER ST	TATUS VIC	e ekāi	J
6	Cause	of fau	lt: 570	PS STUER	NAU OVZ	ZYK BAL	T NUM ST	8	ate of
	(spautone	Displ	ay wit	th PG	670/6	75	9	Fo ram corr	r PC pro- ection rmation
*** 5	only 13	30 WB					l u f	unctions	rmation

The INTERRUPT STACK is a stack into which the system program enters the information required by the PC on COLD START (new start) or WARM START (restart).The contents of the interrupt stack can be read with the PG 670 /675 ("OUTPUT STACK"),when the PC is in STOP STATE.The interrupt stack contains important information concerning the CAUSE for the STOP STATE: with the OUTPUT of the ISTACK of the PG , the contents of system data words BS5/BS6/BS7 are also displayed:

 CONTROL BITS IN SYSTEM DATA WORD BS5/BS6 (SD5, SD6):
 Control bits are internal flags set by the operating system and evaluated at each change from the STOP STATE to cyclic OPERA-TING STATE. CONTROL BITS IN SYSTEM DATA WORD BS7: (SD7)

These displays give additional information about interrupt cause and the procedure for restart.

INFORMATION ABOUT INTERRUPT POSITION (ERROR LOCATION); CPU STATE:

Includes contents of accumulators 1 and 2,

and the result displays for binary and digital operations whose execution was interrupted with the STOP-STATE.

INTERRUPT CAUSE (IN THE ISTACK):

This line shows the user the first information concerning the cause of an interrupt of cyclic processing.The cause displayed is the determining factor for the further course of the procedure of subsequent diagnosis.

The control bits in the system words have the following meanings :

CONTROL BITS IN SYSTEM DATA WORD BS5 (SD5)

	PBSSCH	not	used
--	--------	-----	------

BSTSCH The function "Compress memory content" (COMP:PC) has not been finished.After the cyclic operation is resumed (green LED lit), reactivate function "Compress memory content" at the PC. SCHTAE Block gap in the user program.Restart is possible only after

PC cleared with restart.

ADRBAU Block address is not yet generated/activated.

SPABBR Function "Compress memory content" has been interrupted through a power failure or PC stop.

NAUAS Designation for "Power failure" for the programing unit interface;

QUITT Not used 2 CONTROL BITS IN SYSTEM DATA WORD BS6: (SD6)

"Processing control" with PG.

STOZUS STOP designations.STOZUS indicates that the PC is in stop state
STOANZ STOANZ indicates that the PC is in stop state.
NEUSTA NEW START: cyclic operation is possible only with restart.
WIEDAN* Restart interrupted.
BATPUF CPU contains a buffer battery for RAM memory.
BARB State display for operation modes.

4 5

3

6

BARBEND

UAFEHL Interrupt stack is being processed without prior input. MAFEHL Collective signal for displays in system data word BS7. FOVH PC contains input byte O (alarm processing). CONTROL BITS IN SYSTEM DATA WORD BS7: (SD7) 3 ASPNPR Indicates that the user memory connected additionally consists only of EPROMs. ASPNRA Indicates that the user memory only consists of RAM.In principle, the user has available a RAM capacity of about 5.8 kbytes** in the CPU.The PC is hence operational even without any additional user memory. KOPFNI Indicates that the block type was not recognized during address list generation. The PC is not operational when this flag is set. The program sequence hence branches into the stop-loop. Remedy: initial clear of the PC. PROEND not used PADRFE Indicates that the user PROM memory is incorrectly addressed. The PC is not functional when this flag is set. The program branches into the micro programed stop-loop. Remedy: New addressing of the EPROM modules following the proper quidelines. ASPLUE Indicates that the user memory is being addressed with gaps. The flag is set in combination with flag "PADRFE" CL "RADRFE" .The PC is not operational when the flag is set; the program branches into the stop-loop. Remedy: Readdressing the user memory. RAMADEE Indicates that the user RAM memory is incorrectly addressed. The PC is not operational when the flag is set; the program sequence branches into the micro programed stop-loop. Remedy: Readdressing the user RAM memory following guidelines. KEINAS Indicates that, up to 48 K, no additional user memory is connected, respectively addressed. This means, that the PC is assembled with only the user memory on the CPU. SYNFEH' Indicates that there is no synchronization pattern (inadmissable code) on certain locations in the user memory. Blocks cannot be found when the memory content is undefined; the program sequence branches into the stop-loop. Remedy: Initial clear. only 1 kbyte in the case of 130 WA

11-31

NINEU

a) Indicates that a new start cannot be executed. It is always set in conjunction with a more specific error indication. The exact reason for the restart interrupt can be gathered from the additional error indications. Remedy: Initial clear.

b) Restart could not be executed; the cause was eliminated in the mean time.

Restart no longer possible.Execute newstart. NIWIED*

I/O board fault or EU couple defect or peripheral configuration FAFEHL** changed.

Remedy: Exchange defective board and/or initial clear/load.

A sum error has been recognized within the system program memory or the user program. If the sum fault is still recognized after initial clear and newstart, exchange the system program memory, reload the user program.

Indicates that cyclic operation is possible only after initial clear with initial load, and finally newstart. The initial loading process is executed by the programing unit interface in the range of 0 through 64 kbytes.Afterwards,all RAM memory cells contain OOOOH.

#### INFORMATION ABOUT INTERRUPT LOCATION (ERROR LOCATION) 4

- TIEFE no meaning
- MC-5-code of the last processed instruction. In most cases, this BEF-REG is the false instruction in programing errors.
- This is the memory cell in which the last entry was made, in the BST-STP block stack (BSTACK). The display is without meaning. Display the BSTACK if necessary.
- Address of the memory cell which contained the next instruction SAZ to be processed, at the time when the stop-state occurred. In case of "NNN" error, SAZ contains the address of the instruction where the error occurred. The contents of the memory cell can be read with "Output addr:PC,"SAZ"!" in MC-5-code The error location can be found easier with "Block nr." and "REL-SAZ".
- "BST"-NR Display of block OB-PB-FB, processed before the stop-state. In programing errors, the location of the fault must be found in this block with "DUTPUT PC,"BST-NR."".
- Relative address in given block. When the key-switch "Input lock" REL-SAZ is turned to the right, the relative block addresses can also be .The relative address corresponds displayed on the CRT of PG to the absolute "SAZ" address. The faulty address is directly in front of the relative address
- DB-ADR Start address and number of the data block last called in the DB-NR. program.

SUMF

URLAD

#### 5 STATE OF THE CPU

AKKU 1 Content of AKKU 2 both accumulators

#### Result Display Bits:

ANZO	Display bits 1 and 0 with 2-3 meanings, depending on the type of logic operations (e.g. logic result, comparison result, bit-test result for shift operations).
OVFL	Overflow;for cases where the numerical range was exceeded in a just executed arithmetic operation.
CARRY	Carry-over between the two bytes of the CPU.
ODER	OR-memory.When in a previous OR operation,the RLO was =1.
STATUS	Signal state of the last processed operands.
NLO.	

- VKE (RLO) Logic result of the instruction last processed.
- ERAB The instruction last processed was a first instruction (= beginning of new logic).

#### 6 INTERRUPT CAUSE (IN ISTACK):

The causes of the faults have the following meanings:

STOPS The stop-switch is on NNN <u>Syntax error</u> The user programed inadmissable operations (e.g. access to data blocks with instruction parameter data block length), or operations which exceed the instruction set of the PC S5-130W. STS a ) Stop-request from the user Through operation STP, the user has the option of requesting from the system software, that the PC branch into stop-loop at the end of the running cycle. The branching itself occurs through the stop-instruction of the system software STS. b) STUEB Block stack overflow The continuation address of the block called is stored in the block stack, each time a block is called. When the block stack overflows,PC goes into the stop-loop. NAU Power failure If, when the power returns, the operating mode selector switch is in the OPERATION position, there is an automatic newstart of the PC, respectively a restart, if the OB22 is present. **b**) General PC system error: Further info can be gathered from the control bits of system data BS5/BS6, e. g. addressing of boards wrong etc.

#### Acknowledge Delay

If an addressable range does not **a**cknowledge its addressing within a monitored time interval, an **a**cknowledge delay is issued. Two possibilities for **a**cknowledge delay exist, depending on the addressed range:

• Acknowledge delay at memory access • Acknowledge delay at periphery access

Cycle Time Exceeded

If the cycle time is exceeded, the STEP-5 program is interrupted. The PC goes into STOP. This can happen due to false programing (program execution time too long). Fixed cycle times: 130 WA...270 ms 130 WB...360 ms

#### BAU: Battery Voltage Failure

The signal Battery Voltage Failure is generated by a monitoring circuit in the power supply unit. This signal is recognized by the PC at newstarts; the program sequence branches into the stop-loop. If power failure occurs during battery voltage loss, the contents of the RAM memory are lost. The user must initial clear and initial load the PC. The battery can be exchanged during cyclic operations, without interruption.

Remark: The control bits and the ISTACK can be displayed on the NC CRT with the PC program structure,according to section 12. Control bits SD5,6,7,and 214 can also be read with the NC engineering panel (see section 10 and 11.8.4).

# 11.6.2 Block Stack (BSTACK)

In the BLOCK STACK of the PC S5-130W, during the execution of the program, each time a block is left, two informations are written:

- 1. The start address of the data block valid before the block was left
- 2. The number of the memory address from which the program execution must be continued, after returning form the called blocks (return address)

The information written in the block stack can be read with the PG 670/675 or with the PC program correction in the STOP-STATE of PC S5-130W (OUTPUT BSTACK!).

QVZ

ZYK


11 - 35



JI - 36





11.8.4 System Range for Error Localization for 130 WA and 130 WB

only130 WB

X: No meaning (not used currently)

#: SD is buffered; all others are cleared with each newstart.



**11.6.6** <u>Address List PC 130WB</u>

7/	27   20   25   20   23   27   21   20		Address List Format	
/64	Synchronization Pattern 7070		Absolute Start Address of	
	Block Designation A100 -	DIOCK Header	ADDRESS LIST BLOCKS	
	PU Designation	List_DB_0	DB's D800	
5~ 1	Library Number 2000	TO Bytes	PB'S DAUU	
	-Library Number 0000 -	all o		
		, B	SB's DEOO	
_	-Block Length (words) 0300 -		Relative Start Address of	
	Spare Word <b>80000</b>	2 Bytes	Address List Blocks	
13.2	Relative Start Address OONE	1. 1 ⁰	(Distance in words from the header	-
SE	OT THE SE'S	25	address of the address list)	
	-Spare Word UNL	č.	DB'S UU4E	
	Relative Start Address	16 Bytes	PB'S UU8E	
	of the PB's	ALC Y	IB'S UULE	
	_Relative Start Address 006E			
	of the FB's		SB'S UTUE	
8	_ Keiative Start Address UNL	6	Distance Address of the Address	
No.X	Relative Start Address	Max.	List Blocks	
2	of the DB's	50	O Distance in bytes from the start	t
	Spare Word 0000	25	address of DB 0)	
	Spare word	1000	DB'S 006A	
	- Spare Word 0000	ALC'	PB'S OOAA	
80			FB'S OOEA	
~	- 06 0		0B'S 002A	
2	• &		SB's 012A	
NOX	<b>₩</b> 64 08's	128 Bytes	length of the Address list Blocks	
$\mathcal{E}_{I_{i}}$	<u></u>	50	in Bytes	
	63	10	DB's 0200	
00	AR 0		PB's 0200	
			FB's 0200	
	All All		0B's 0080	
-	256 DB1s	512 Bytes	SB's 0200	
0		à	6 6	
X°.	DB 255	X.		
00			Header Address of the Address Lis	<u>t_</u>
		-10 ¹⁰⁰	D764	
	S S	S.		
	256 PB's	512 Bytes	Absolute Start Address of the	
an or	<b>PB 255</b>		DIOL	
	- FB 0 -	31	Length of Address List	
	() ()	205		
	256 FB1s	512 Bytes	IN DYTES: U896	
	FB 255	and the second	TH MOLOSI OAAF	
00	<b>52</b> 0			
2		132		
	NL V	<b>-</b>		
	256 SB's	512 Bytes		

11 - 40

# Instruction Set 130 WA

	- Aler	Cycle	. Georation a	No.	Cenditie	All a	
Operation	Paraseter	(µs)	Byte O	Byte 1	RLO F	[8 Function	S. S.
linary lo	gic operatio	ns	Bi t øddr	Byte addr.	, S	AND logic	Partic
A .34 ³⁵	I 0.0 to 127.7•	3.75	C 0 1100 0800	0 0 0000 x0000	X	Scan imput for #1#	
A	Q 0.0 to 127.7*	3.25	C 0 1100 0800	1000 X0000	X	Scan output for #1#	
A	F 0.0 to 255.7"	3.25	8 0 1000 0000	0 0 x000x x000x	X	Scan flag for #1#	S.
AN	I 0.0 to 127.7*	3.25	E 0 1110 0000	0 0 0000x x000x	X	Scan input for "O"	. HOLL
M 8	0.0 to 127.7*	3.25	E 0 1110 0000	8 0 1300 3000X	X	Scan output for #0#	So
All	F 0.0 to 255.7	3.25	A 0 1010 0000	0 0 X000X X000X	X	Scan flag for #0#	
_	<u> </u>		Bit øddr.	Byte addr.		OR logic	
0	I 0.0 to 127.7*	3.25	C 8 1100 13000	0 00000 x0000	X	Scan input for #1#	2ª
0	0.0 to 127.7*	3.25	C 8 1100 13000	8 0 1X0X XX0X	X	Scan output for #1#	20110
0 8	F 0.0 to 255.7	3.25	8 8 1000 13000	0 0 X000X X000X	X	Scan flag for #1#	S.
ON	I 0.0 to 127.7"	3.25	E 8 1110 13000	0 0 0000x x000x	X	Scan input for #0*	
ON (	0.0 to 127.7*	3.25	E 8 1110 13000	8 0 1X00X X000X	X	Scan output for #0*	
ON I	0.0 to 255.7	3.25	A 8 1010 13000	0 0 x000x x000x	X	Scan flag for #0#	Å
	- S ^{CO}		· 30-	Word addr.		AND logic	. S. M.
<u> </u>	0 to 127	3.25	F 8 1111 1000	0 0 x000x x000x	<b>X</b> _	Scan timer for #1#	Say.
AN STATE	0 to 127	3.25	F C 1111 1100	0 0 X000X X000X	X	Scan timer for "O"	
<u> </u>	0 to 63	3.25	B B 1011 1000	0 0 X000X X000X	X	Scan counter for centents	.>0
AN C	0 to 63	3.25	B C 1011 1100	0 0 x000x x000x	X	Scan counter for contents	-0
	-Car		and a second	Word addr.	1	OR logic	
0	0 to 127	3.25	F 9 1111 1001	0 0 X000X X000X	X	Scan timer for #1#	600 C
H SAL	0 to 127	3.25	F D	0 0 XXXXX XXXX	X	Scan timer for "O"	0
) c	0 to 63	3.5	8 9 1011 1001	0 0 X000X X000X	X	Scan counter for contents	>0
DN C	0 to 63	3.5	B D 1011 1101		X	Scan counter for centents	-0

") The input and output bits (bytes, words) 64.0–127.7 (64–127, 64–126) can be used as additional flag bits (bytes, words). They are leaded (transfered) from the system program and not from (to) the peripheral modules.

						11	-42			
										. chanon
wration		Paramater	Cycle time (ps)	Operat Byte O	ion co	ie Byte i	4	Condi codes RL0	FIB	Function
		2			- 1	<u>s</u>				AND/OR logic
		5 <u>5</u>	3.25	F 1111	1011	<u> </u>	_		X	ORing of AND functions
(	ş		4.5	8 1011	8 1011	°	_		X	ORing of pereathesised expressions
( NIC			4.5	0 B 1011	A 1010	°		31.0	X	ANDing of parenthesised expressions
Page 1	4		3.75	B 1011	F 1111	0	0 3	£4	X	Right paranthesis
tting/re	i	tting or	eration	18	BLt Laddr	Byte	addr.			2
	ī	0.0 to 63.7	3.75	D 1101	0 1XXX	00000	0 XXXX	. Х		Set input to #1#
	0	0.0 to 63.7	4.5	D 1101	0 00000	8 1XXX	0 X000X	Х	305	Set output to #1#
	F	0.0 to 255.	7 3.5	9	0	0 X000X	0 X000X	X	2	Set flag to #1#
al ^{an}	T	0 0 to 63.5	3.75	F	0000	0	0	X		Set input to "O"
	•	0 0 to 63	4.5	F 1111	0000	8 11007	0	X		Set output to "O"
<u> </u>	ч г	0.0 +0 255	7 35	B	0			X		Set flag to "O"
	- 	0.0 10 2)	2 95	D		<u>~~</u>		X	.s	Set input to #1# conditionally
.8	<u>×</u>			D	8		<u> </u>	<u>x</u>	0	Set outout to "1" conditionally
- all			7 7.7	9		0		Y		Set flag to #1# conditionally
		0.010 277	./ 1.3.7			Nend			L	
Liser and			peracito.	3	4.1	O	0	X		Start timer as miles
5r		010 121	20.0	1	C	0	0			Start time as antended sules
SE (	P	0 to 17/	38.5	2	1100	x000X 0	X000X 0		\$°°°	Start timer as extended pulse
SR (S)	T	0 to 127	38.5	0010	0100 C	X000X	X00X		+	Start timer as UN delay
SS	T	0 to 127	38.5	0010	1100	X000X	X000X	X	+	Start timer as stored ON delay
SF	T	0 to 127	38.25	0001	0100	XXXX	x000X	X	<u> </u>	Start timer as OFF delay
R	T	0 to 63	6.5	3 0011	1100	x000X	x000X	X		Reset timer
S	C	0 to 63	42.75	0101	100	x000x	20000	X	100	Set counter
R N	C	0 to 63	6.25	7 0111	C 1100	0 x000x	x000x	X	-	Reset counter
CU	C	0 to 63	11.0	6 0110	C 1100	0 X000X	20000	X		Count up
۵	C	0 to 63	9.75	5 0101	4 0100	x000x	x0000	X		Count down
Ba										

www.spaulonashaf

www.cbou

Indeastornativa é

matykan

				11-45		
Operation	Parameter	Cycle time (µs)	Operation of Byte O	ode   Byte 1	Condition   codes   RLO   FIB	Function
Load and	transfer fur	nction	8	2		2
1 T	B 0 to 177*	3.25	4 A			Load input byte of process input image
1 1	0 to 126*	4.75	5 2 0101 0010			Load input word of process input image
	R 0 to 127*	3.75	4 A			Load output byte of process output
1 0	W D to 126*	A 75	5 2	8 0		Load output word of process output
	0 40 266				<u> </u>	Land flag but fat and
		4.0	1 2	0 0		Losa Tiag byte into accu i
		5.0	0001 0010 2 A	xxxx xxxx xxx		Load flag word into accu 1 Load right-hand byte of current data
	K U to 277	25.75	0010 1010 2 2	x000x x000x	100	block into accu 1 Load left-hand byte of current data
	L 0 to 255	24.75	3 2	X000X X000X	100	block into accu 1 Load word of current actual data block
	V 0 to 255	26.75	0011 0010	x000x x000x	£	into accu 1
	0 to 127	5.5	<u></u>			Load time (binary) of timer into accu 1
L C	0 to 63	5.5	0100 0010	X000X X000X		sccu 1
L P	B 0 to 127	10.0	0111 0010	x000x x000x	- d	into accu 1, bypassing the process
L P	0 to 254	12.0	7 A 0111 1010	0 0 x000x x000x	1000	outputs into accu 1, bypassing the process image
LD T	0 to 127	10.75	0 C 0000 1100	x000x x000x	S. A.	Load time (BCD) of timer into accu 1
ນ ເ	0 to 63	40.5	4 C 0100 1100	0 0 x000x x000x		Load count (BCD) of counter into accu 1
T I	B 0 to 127*	40.0	4 B 0100 1011	0000 x0000		Transfer contents of accu 1 to input byte of process input image
T T	0 to 126*	4.0	5 3 0101 0011	0 0 2000x x000x		Transfer contents of accu 1 to input word of process input image
T Q	B 0 to 127*	4.0	4 B 0100 1011	8 0 1X00X X000X	.3°	Transfer contents of accu 1 to output byte of process autput impo
T	1 0 to 126.	4.0	5 3 0101 0011	8 0 1XXX XXXX	15	Transfer contents of accu 1 to output
T S	B 0 to 255	4.0	0 8		19	Transfer contents of accu 1 to flee but
T F	0 to 254	4.0	1 3		<u> </u>	Transfer contents of accu 1 to flag upon
T D	R 1 to 255	24.5	2 8	0 0	<u> </u>	[ransfer centents of accu 1 to the word [right_hand byte] of current actual
T D	1 to 255	25	2 3	0 0		Nate block (Transfer contents of accu 1 to the word (left-hand byte) of current data block
T	1 to 255	25.25	3 3	0 0	1000	Transfer contents of accu 1 to the word
T SA D		40.5	7 3	0 0	1996 1997	of current data block Transfer contents of accu 1 directly
		10.7	0111 0011 7 B	0 0 0	<u>.</u>	to peripheral byte Transfer contents of accu 1 directly
<u> </u>	U to 254	12.25	0111 1011	X000X X000X	<u> </u>	to peripheral word

www.costonadia.d

10	<i>2</i> ¹		Cycle   time	Opera	tion c	ode .		Cendi cedes	tion	Emobia
Operation		Parazeter	((44))	Byte	0	Dy		HLU I	110	r mic cavit
[	13	0 te 255	3.25	2 0010	8	0 X000	0 X000X			Load constant number (1 byte) into accu 1
ι ι	15*	2ASCII character	5.5	3 0011	0000	1 0001	0000		6	Lond constant character into accu 1
ι .	NF*	-32768 te +32767	5.5	3 0011	0 9000	0 0000	4 0100	3	198°	Lond constant fixed-point number into accu 1
L	101.0	0 to FFFF	5.5	3 0011	0000	4 0100	0000	44		Load constant number (hexadecimal code) into accu 1
ι	Kita	000000 te 11111	5.5	3 0011	0000	8 1000	0000			Lond constant bit pattern of a word (2 bytes) into accu 1
ι	KY*	0 to 255, 0 to 255	5.5	3 6011	8000	2 0010	0000			Load constant number (2 bytes) into accu 1
ι	KT°,	0.0 to 999.3	5.5	3 0011	0000	0 0000	2 0010		6	Lond constant number (2 bytes) as time into accu 1
L .	KC*	0 to 999	5.5	3 0011	00000	0 0000	1 0001	2	. B	Lond constant number (2 bytes) as count into accu 1

*) These are 4-bytes operations in which the constants are in bytes 2 and 3.

#### Comparison functions

1 • F	all the	8.75	2 0010	1	8	0	Fixed-point comparison for accul equal to accu2. If equal, RLO-"1"; condition code < 0 and > 0
> F	30	9.0	2 0010	1 0001	2 8010	0	Fixed-point comparison for accul > accu2. If accu2 > accu1, RLO=1=; condition code < 0 er > 0
< F	Š	9.0	2 6010	1 0001	4 0100	<u> </u>	Fixed-point comparison for accul < accu2. If accu2 accu1, RLO="1"; condition code <0 or>0

#### Note : \

Note: The programmable controllor has two accumulators for comparison and arithmetic functions and for digital operations. Londing means that the contents of accu 1 are transforred to accu 2 and that accu 1 is newly londed according to the operands in the lond operation. After two lond operations, information on the contents of the accumulators can be obtained with comparison operations.

#### Example:



A transfer operation always transfers the contents of accu 1 to the operands specified in the transfer eseration.

					11-4	5		
) Opera	tion	Paraaeter	Cycie tise (µs)	Operation Byte O	code Byte 1		Condition codes RLO   FIB	Functfon
Blo Blo	ck call	•			Herd ad	dr.		No.
u	PB	0 to 255	30.75	7 5 0111 010	0 1 X000X X	0 000X		Jump unconditionally to program block
u	FB	0 to 255	30.75	3 D 0011 110	0 1 X000X X	0. 000X	130°	Jump unconditionally to function bloc
uL	08	1	30.25	6 D 0110 110	0 1 X000X X	0 000X	and	Jump unconditionally to organisation block
x	PB	0 to 255	31.75	5 5 0101 010	1 X000X X	0 000X	X	Jusp conditionally to program block
x)	FB	0 to 255	60.25	1 D 0001 1101	0 X000X X	80X	X	Jusp conditionally to function block
x	06	5 ¹⁰ 1	8.0	4 D 0100 110	0 1 X000X X	0 000X		Jump conditionally to organisation block
c	DB	1 to 241	19.25	2 0 0010 000	0 x xxxxx x	0 0000	and the second	Call data block; the data block is valid until another DB is called.
BE	and and a		23.5	6 5 0110 010	0 1 0000 0	0	and a second	Unconditional end of block
₿€€			23.15	0 5 0000 010	1 0000 0	0 000	X	Conditional end of block
Oth	er opera	ations 2			12.9		••	10 ^{.2}
NOP (	)	S. B.	3.25	0 0	0 0000 0	0		No operation (all bits deleted)
NGP 1	100		3.25	F F 1111 111	F 1 1111 1	F 111	100	No operation (all bits set)
STP	. and .		11.5	7 0 0111 0000	0 0000 0	3 1011	A A A	Programmable step operation (at the en of the cycle, the programmable
<b>ULD</b>	<i>A</i> .	0 to 255	3.25	1 0	0 X000X X	0 XXX	4	Display construction statement for the
2 <b>86</b> 5		0 to 255	7.5	1 8 0001 1010	0 X00X X	0 XXXX	▶ <b>}</b>	An operation in the system data range
STS		S. B.	38.5	7 0 0111 0000	00000 0	0000		System stop
TAK			4.25	7 0 0111 0000	00000 0	2 1010	1000	Interchange the contents of accumulator 1 and 2
Load	and to	ransfer op	eratio	ns (syst	em oper	ati	Lons)	
LIR	1.	0 to 2	13.25	4 0 0100 0000	0 x000x x0	0 XXX		Load register (indirectly) with the ce tents of the memory word addressed by accumulator 1
TIR		0 to 2	13.0	4 8 0100 100	0 XXXXXX XX	0 XXX		Transfer the register contents (indi- rectly) onto the memory word addresse by accumulator 1
TNB	à	0 to 255	842	0 3	0 X000X X0	o xox		Block transfer in the byte mode
TBS	N.GD	0 to 255	10.75	6 3 0110 0011	0 X000X X0	0 X00X	and Co	Transfer word to the system data area
Arit	thmetic	operation	354				4	R.
4000		. 122		5 0	0	0		Add byte constant (fixed-point) to the

ADOBN	<u>+</u> 127	3.25	5	0000	0 x000x	0 X00(X	Add byte constant (fixed-point) to the contents of accusulator 1
ADOKF	-32768 +32767	5.5	5 0101	<b>8</b> 1010	0 X000X	0 X000X	Add fixed-point constant (word) to the contents of accumulator 1
				30			3 ⁰

www.costonadia.d

		Cvcle		Condition	
Ameratian	Parameter	time (us)	Operation code Byte 0   Byte 1	codes RLO   FIB	Function

#### Digital logic functions

AV	·	3.25	4 0100	1 00001	0	0	630	Digital AMDing of accul and accu2(word for word);result stored in accu1; < D or > 0
OV		3.25	4 0100	9 0001	0	0	and in	Digital ORing of accul and accu2 (word for word); result stored in accul; <d or=""> 0</d>
XOW		3.25	5 0101	1 0001	0			Digital EXORing of accu1 and accu2 (word for word);result stored in accu1; <0 or>0

## Arithmetic functions

•F	<u> </u>	5.25	7 0111	9	0_0	Add accu 1 to accu 2; result stored in accu 1; < 0, > 8 or OV
-F NIG		9.5	5 0101 1	9 1001	0_0	Subtract accu 1 from accu 2; result stored in accu 1;<0, >0 or OV

Jump fu	nctions			1	Word a	iddr.		8
- UL	"Label" (4ASCII	7.5	2 0010	D 1101	0 X000X	0 XXXXX		Jump unconditionally to label, consist- ing of 4ASCII characters. Jump dis- placement ≤ \$ 127 words.
x.	"Label" (4ASCII	8.75	1111	A 1010	0 X000X	0 X000X	X	Jump conditionally (if RLO_*1*) to label consisting of 4ASCII characters. Jump displacement # 177 words.
л.	"Label" (4ASCII	8.75	4	5 0101	0 X000X	0 XXXX	્ટે	Jump conditionally (if result-zero) to label, consisting of 4ASCII characters. Jump displacement # 177 words.
- OL	"[abel" (4ASCII characters)	8.75	0000	D 1101	0 X000X	xxxx	star.	Jump, conditionally (if condition code UW-1) to label, consisting of 4ASCII characters. Jump displacem.s±127 words.
JUR-	-32768 to +32767	9.0	7 0111	0000	00000	8 1011		Skip system software unconditionally

#### Timer and counter functions

Timer and	i counter fund	tion			Word	addr.	12 ²⁴	State -
FRT	0 to 127	5.75	0000	4 0100	0 x000x	0	X	Enable timer for celd restart (only on positive going edge of RLO)
FRC	0 to 63	5.5	4 0100	4	0 X000X	0 X000X	X	Enable counter for cold restart (only on positive going edge of RLO)

11-47

		Cycle			Cendi	ion	
•	• · · · · · · · · · · · · · · · · · · ·	tim	Operation c	ode	codes		r
Uperation	Parameter	((12))	Byte U	Byte 1	MLU	118	Function

Shift functions

ouri e runo.						rar.		
SLW	0 to 15	60.25	6 0110	1 0001	00000	<b>0</b> 0000	10	Shift contents of accul to the left. The bit positions to the right which become wacant are moded with zeroes.
SRV	0 te 15	60.25	6 0110	9 1001	0 0000	0 X00X	4	Shift contents of accul to the right. The bit positions to the left which be- come vacant are padded with zeroes.

#### Conversion functions

CFN	 3.5	00000	1 0001	0_0		Dne's complement of accu 1
CSW	 3.5	0000	9 1001	0	AN CON	Two's complement of accu 1; <0,> 0 orOV;

Decreme	inting/increment	nting			Dec./	Incr. 0 255	Ś. Ś.
D	1 to 255	3.5	1 0001	1001	0 X000X	0 X000X	Decrement only the low-order byte of accu 1 by a particular value
I	1 to 255	3.25	1 0001	1001	0 XXXX	0 X000X	Increment only the low-order byte of accu 1 by a particular value

Proce	ssing	functions	20			Word a	addr.	
DO	FN	0 to 254	15.5	4 0100	E 1110	<b>x00</b> x	0 x000x	Process flag word. The next operation specified is combined with the parame- ter in the flag word and executed.
DO	DN	0 to 255	31.75	6 0110	E 1110	xxxx	0 XXXXX	Process data word. The mext operation specified is combined with the parame- ter in the data word and executed.

#### Disable/enable command output

BAS		3.25	B 1011	E 1110	0	0 3	X	Disable command output
BAF	<del>~~</del>	3.25	F 1111	£ 1110	<u></u>		X	Enable command output

#### Disable/enable interrupts

IA SOUTH	 3.25	0000	8 1000	0000	0000		and a	Inhibit interrupt processing
RA	 3.25	0000	8 1000	0 1000	00000	19. 19. 19.		Enable interrupt processing

# Instruction Set 130 WB

-			Pin	-		<b>P</b> +			Cana		•	191	furger S
		Sec.		-		land 1							a second s
		20		1 81	82				-	-	Ø	11	102
		2000			3	ೆ				- 20	87		AND sperson with
A	۰,	0.0 to 63.7	8			] -	N	N		12	-	4.0	asserving of enjoys for "1"
A	0	0.0 10 63.7	8	80	- N	-	N	N	42	-	-	3.5	asserving of susput for "1"
A	F	0.0 to 255.7	8	1 10	-	-	N	N	-	-	-	3.5	starring of the for "1"
A 0	T	0 <b>to 12</b> 7		80	-	-	N	N	-	-	-	3.5	spanning of strain for "1"
<u>.</u>	C	0 w 127	1	1 80	-	-	N	•	-	1-	-	3.75	scarring of asumar ter = 0
AN	1	0.0 w 63.7	80	: 00	1 -		<b>N</b>	N	] -	-	1.0	3.5	scarning of input for "\$"
AN	٥	0.0 m 63.7	80		i -	130	N	N	-	ī-,	32	3.5	economy of evaport for "8"
AN	F	0.0 to 285.7	1 A0	1 80	1-0	gr.	N	N	-	Ľ	- 1	3.5	scarring of Reg for "\$"
AN	T	0 10 127	i re	1 80	$\mathbf{F}_{\mathbf{r}}$	-	N	N	- 3	1	-	3.5	accenting of similar for "\$"
AN	C	0 10 63	- <b>BC</b>	. 00	-	1 -	N	N	24	-		3.75	scanning of counter ter = 0
					-				_				Off operation with
0,8	I	0.0 m 63.7		80	]-	-	Ν.,	N	-	-	-	3.5	scenning of input for "1"
0	Q	0.0 w 63.7 🔬	a		-	-	N	N	-	-	- 3	3.5	scenning of sulput for "1"
0	F	0.0 to 255.7		80	-		<b>N</b>	N	-	]-		3.5	scenning of flag for """
0	T	0 to 127		00	! -	1	1 N	N	-	-	8°-	3.5	acoming of other ter "1"
0	C	0-10 63		60	1.59	¥	N	i N	-	Ś	-	3.75	acanning of counter for > 0
ON	, d	0.0 w 63.7	68	00	Star.	-	N	t N	- 3	19 -	: =	3.5	acanning of mout for "\$"
DN	٩	0.0 10 63.7	B	80			N	N	A.	; <del>-</del>	-	3.5	scenning of output for "#"
ON	F	0.0 to 255.7	A	60	-	-	N	N	! -	-	· -	3.5	economy of Neg for "9"
0	۲	0 to 127	PD	80	-	-	NC	NN -	-	-	-	3.5	ecerving of smer for '9'
ON	C	0 to 63	80	00	-	-	N	N	-			3.75	scanning of counter for = 0

	+ teit addre	nietove addre		 						
	Native and	-	;	i:	: Care			1,	· Function	- 12 all
	Ward 0 60 81			ेश्वी			: 07	- 1		~
<u>,</u>	00 , HE 📝	-	- N	Y	· _		-	3.5	Right paranthasis	
H S	8A 00	-	N	Y		-		4.75	ANDing of brackated expressio	2
N	88 80		9 I N	Y	-	·	39	· +.25	Offing of bracketed expressio	
	; FB , 00	- 28°	- N	, <b>Y</b>		- 8	-	3.5	ORing of AND functions	.80
444		and a second			34	24.			44	

						tomat	11-4	19				a.dl	
Betting /reartting a	perotione Perote				get Rhand 1			2		•	1	- Kell	
<u></u>		1 <b>.</b>	•			151		821	609		123		-
<u></u>		1	1	1	1	<u>.</u>	1	1	1	1.5			-
÷ · · ·	0.0 00 03.7	100		+-	100					<u>}_</u>		an andres in Pili	- 8
	0.0 m 255 7			1-2	<u>}</u>	+	$\frac{1}{\mathbf{v}}$		2	+	118	a flag	State -
			<u> </u>	10				1			1 417	Reart	<u> </u>
· R 📐 F	i 0.0 10 63.7	1 #0		i -	1 -	14	142	-	1-	-	4.0	an input (in P11)	
• # 🖉 🛛 O	0.0 w 63.7	. 10	80	! -		¥	Y	-	1_	1_	4.5	an event in PIO:	_
10 F	0.0 10 285.7	1 80	1 60	1-	1-	1.	Y	-	-	-	4.75	. e Neg	
0	10					6				1	2	Aariyn 😥	_
• 1	0.0 m 63.7	1.08	: 00	-	10	1 N	3 <b>V</b>	-	-	22	4.0	an input (in Pil)	8
- 0	0.0 10 63.7	DB	i 💼	- 3	<u> </u>	N	i y	-	al l	Ĩ	4.5	an evigut (in PID)	_d ^h
± F .	0.0 to 255 7	1 🗰	80	2		I N	14	<u> </u>	1 -	<u>  -</u>	3.75	e Neg	25
		- 100		1997-19 20 5	idrees		202						

Timer and co	wnter	eperations					5		<b>_</b>		1000		<u> </u>	5.
Operator		Parameter				<u> </u>	- 1			and?		11	E fumer	
				•r	<b>1</b> 2	80	- :1	2Ì		600	1 04	-11		
SP 💫	T	0 10 127	24	00	-	-	۲	YA	-	-	-	21.5	Start smar as pulse	
SE	T	0 to 127	10	<b>60</b>	· -	-	۷	S.	-	-		21.5	Start timer as extended pulse	_
SP	T	0 to 127	200	<b>0</b> 0	-	-	y ô	ੇ <b>∀</b>	_	-	-	21.5	Start smer as "On" deley	
55	Ŧ	0 to 127 🔬	<b>) 2</b> C	00	-	-	Ŷ	۲	-	-	- 3	21.5	Start amer as stored "On" delay	
SF	7	0 10 127	14	60	-	2	۲	¥	_		20	21.5	Start wher as "Off" delay	
R	T	0 10 127	x	00	- 3	<u>84</u>	: ¥	۲	-	100	<u> </u>	5.75	Real timer	- shi
<u>s</u>	<b>C</b> .	0 10 63	<b>6C</b>	80	150	_	¥	· •		3 ²²	-	23.5	Set counter	3200
<u> </u>	C	0 10 63	70	<b>60</b>	-	<u> </u>	1 ¥	¥		-	-	5.5	Reset aburter	
<u></u>	C	0 10 63		े क	-	-	· ¥	<u> </u>	-	-	-	6.0	* Increment counter ICOURt up?	_
CD	C	0 10 63		60	-	-	. •	Y	<u>.                                    </u>	-	. =	1.5	Decrement counter (count down)	
ć° .				18	and and	-	5	-						

www.cpaitonadka.d

						11	-50				
	- abaute			paule					doard .	2	the state
Land and search		100000 0		W1 	.1	.1	3		•		h
										1	
L 8	0 10 63		<u>n   -</u>	-	N	N		<u> -</u>		4.25	an staut avan tivern Pill
L W	0 - 62		0 -	720	11 N -	N N	-			4.75	an argut ward (train PIL)
· L 00	0 10 63			5	i N			(	<u>6</u> -	4.75	an extent word from PiO
	1 0 10 28		n - 1. 0	+	N	N	-	12	-	14.25	a Ray byte
L PA	v 1 8 to 284	12 0	- I	-	N	N	- 7	-	-	5.0	a flag ward
L DL	0 to 255	22 0	n – O	-	N	1	]-	-		13.0	a data byta (laft-hand byta)
L DA	C to 285	24 1	<b>•</b> ' -	-	N	N	-			114.25	* a date two inght-hand two-
L DV	W 0 to 286	22			100	1 1		<u>i -</u>		12.5	
<u>· L T</u>	: 0 10 127			4.50		<u>                                       </u>				1 k7	a manya
<u> </u>	0 10 0j	172 - 4		d-		1 11	1		200	10,25	a perphasi byte of the depter mouth
	128 to 755		- 13	2	1	<u> </u>	1	1	9	1	a percharai aver of the analog incuts
L	V 0 10 128		- 12	-	N	•	-	1-	-	TL D	a perphasi word of the digits! Houts a perphasi word of the shake Houts
<u>цр</u> т	0 to 127	8C (	- 0	<del>-</del>		N	 _		1 -	25.0	a ume (BCD)
<u></u>	0 m 63	્યટ			1 1	1.0	-	-	· -	25.0	a source (BCD)
	A.	2			A	-				2	to the
					<u> </u>				<u>.</u>	and the second s	
	A to 255		10 <del>-</del>	- 20	- N	. N	· <b>-</b>			i 4.0	1 ivite 20
				<u> </u>				. (	S.		. S
	the second	<del>.</del>		(1 SVB)				and i			
- Th			N.				3	2		1.6	
	derschers		••••••••••	•••							
SL I	KM Bit pattern 116 pm	°S20 (	80 08	80	- N.,	ିଳି	-	-	-	7.0	as bit pattern
L 1		30	40 50	80	<u>_</u>	N	-	-	· _	7.0	in heredecimal code
1	KF 0 10 (218.1)	30	04 00		N	· N	-	-	- 8	7.0	
L 1	KY 0 10 255	30	20 60	Som.		N		-	⁶	7.0	2 bytes
	KT 0.0 m 500.3	30	82 80		N	N		1	<u> </u>	7.0	
- 3ª	KC 0 10 100	30	01 60	1 00	· N	• N	3	÷.	· -	1 7.6	. 45 8 COUTT
			 i • •				-,				
è	<u> </u>	<del></del>				÷.					<u>, è</u>
T 🖬	6 m ft at 1	<u></u>	<b>m</b> -	-	- NY	<u></u>			· _	4.25	te an insut byte in Pil-
T N	N D to <u>12</u>	8		-	<u>En</u>					4.5	to an mout want in Pili
T 0	B D to 63	-	<b>ID -</b>	20	N	· N	-	. <del>-</del>		4.25	to an surput lavas (in PiO
T 0	W 0 10 62	5	• -	8	+ N	. N		- 1	8-	4.25	b an autout ward im PiO:
T P	0 w 255		- 00	-	N	N	-	200	_	4.75	o to a flag ovia
T	W 0 to 254	<b>.</b> tr	<b>.</b>		. N	! N	-4	е́г –		4.25	to a flag ward
<u> </u>	A : 0 to 255	28	90 i -	-	• N	N	-	+	-	15.5	to a data byte (nght-hand byte-
<u></u>	L 0 to 255	<u></u>	<b>60</b> :		N	1.				W.25	to a data byte (laft-hand byte)
7 -		73		· -		- 1 <b>N</b>	-   -	+-		10	
	128 10 286			a Jeo	9	1			and the		with updating of the PIO to a samphing tive of the analog suspirts without updating the PIO
	W 0 10 126 138 10 254	78	•••	, 2	• <b>N</b>	N .	:-	and	S. <del>-</del>	<b>9.25</b>	to a perpheral word of the digital evaluation with updating of the PIC to a perpheral word of the analog evaluation without updating the PIC
<u> </u>			- 191879-18								
					s.	2					

www.cbaitomati

www.chaitonati

0	~~~~		-			-	0	-		1	¥ _∎ !	оролан (орола) Станата станата станата Станата станата
		100		·	1	1.1	.1	<u> </u>	·····	-,	12	
S)		<u></u>	81	82	-	111	¦∉₹	ac,	000	1 DV	1.4.4	
	<u></u>					0		<del></del>		- 10		Compare final-comt numbers
1 = F		21	1		Ĩ	N	N	Y	¥	<u> </u>	7.5	for equal to
> <f< td=""><td>A.C.</td><td>21</td><td></td><td></td><td>P</td><td>N</td><td>N</td><td>Y</td><td>18</td><td>이</td><td>7.5</td><td>fer net equal to</td></f<>	A.C.	21			P	N	N	Y	18	이	7.5	fer net equal to
> F	J.	21	20	120	-	N	N	۲.	Y	j	1.5	for greater than
> = F		21	-	1_	i -	N	i N	14	Y	1 -	1.5	for graster than or abuel to
	· · · · · · · · · · · · · · · · · · ·	21	1 40	1-		IN	N	14	Y	-	7.5	for lass than
100	- <u> </u>	21	<b>C0</b>	+	<u>†</u> _		I N	v	V	-	7.5	for ions that or equal to
<u></u>		<u> 30</u>	1	1	_ <u>i</u>	100	1	1			3	100 m
Arithmetic es	erstiens		1-		÷		1.	7	15	1	1	
+F	Sec.	70		- 20					100		4.75	
- •	2		00	4-	! -	N	N	43			1 5.25	Subtrachen, Nets-point
Block solls		de.				les	<u> </u>				- Ar	
8		C ^o				See.			<del></del>		Ś ^o	Unconditional jump
UL	P8 0 to 255	76	00	-	1.3	M N	۲.	-	-	3	40	to a program block
	F8 0 to 255	30	80		S	N	١v.	! -		9°-	40	te a function block
	\$8 0 to 285	1 70	1 80	5	сI	I N	1 ¥	-	12	! -	40	to a sequence block
<u>.</u>	14			27	<u> </u>			1	2			Conditional partici
	PB 0 m 255		. 💼		:	: <b>Y</b>	· •	-		-	1 23.25	te a program block
10 0	FB 0 1 25	10	-			<b>v</b>	<u>_</u>	: <b>-</b>		1 -	73.25	Le a function block
	48 Am 144					2.4	<del>G</del> ₽		· _	<u> </u> _	7.75	
<del></del>		<u></u>		-		100	1				C ^{er}	
	<b>50 6 1 1</b>	· · · · · · · · · · · · · · · · · · ·			5	<u>і</u> .			<u> </u>	150	1 25.25	Cations of a data block
<u> </u>	DB 0 10 286				100					<u></u>		
			1 - 1	NOCE THAT	the the	udecima	(1)		35			
				15		1	1	1	20.			
. BE			1 =	<u> </u>			1 <b>7</b>	-	1-		1 47	
DEC	: 	66		4		· ·	18		<b> -</b>		16.5	Black and, canditional
BEU +)		1.0	10			N	1.	<u> </u>	1-	1-	8	Bleck and, uncondisional **)
•) If a block insertee a factor: the factor: the	is ended with NED, m fter the respective b on the jump beck is m of fallowing the bloc on	a BP state lock call. nos after 1 k call is :	papet a BEB the ekspect	unt in stata-								••) Buly permitted in FB. FB and SB.
Cast Sec.				- 12 - 12 - 12 - 12 - 12 - 12 - 12 - 12			1	-	-	•	1.	Parate - 1
				1	Vilers 1	Ì		-	<b></b> )	_		
			- m		1 80	- 2 I	11	(C)	CC0	1 84		ð <u>ð</u>
NOP	0		1 00			. N.3	9 N	-	-	-	3.5	No operation tell time resert:
NOP	1	<u> </u>	1 54			1.1	N	1_	-	/	3.5	
STP		1 20			1.3	S N		-	-	130	11.5	Stop
			-+ <del>-</del>		300			<u> </u>		d'	1 15	-
810			1 67									

		Jucof						्रो	1-52	2			
	Legic operation	ns. digital (suppleme		retional	Sec. 1	'goo.				and Constant	9 ⁰⁰	 T	Andre Gio
	Carrier .	700000	1444			41 	Ī	2	-	1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 -			2
	all a		3 <b>•</b>			-		21	<b>CC</b> 1	600	0		
	AW	°6.	41	80	-		i N	N	×	X		3.25	ANDing
	OW		•		1	20	N	N	×	x	1	3.5	Offing
. 8° -	XOW	0	\$1	1 00		¢.	1		x	X	1-	3.25	Excusive Offing of ACCU 1
al and a low	XOW	ANN ICT	<b>\$</b> 1	=0	444	(fř	<u> </u>	<u> </u>	X	<b>x</b> ()	-	3.25	Exclusive Offing of ACCU 1

S			<u>de</u>				14			<u>  </u>	10	1 ter ** 13**
AN	•	Fermel epiteric	Ð	<b>90</b>			N	-	-	1	39.5	AND operation with acanning of formal operand for "0"
0	•	Former operand			- Z	Sol N	*	-	5.	82	38.5	OR eperation with scenning of fermal operand for "1"
ON		Formal operand	25		13 ²⁴	N	•	-	12	-	38.5	OR operation with scanning of formal operand for "O"
				- 14					-			

Norma     Norma     Norma     Contract com     Con     Con	
Word 0     Word 0<	
80 81 87 183 2 8 2 CC1 CC0 OV 2 1	
T Gala 177 DA GO SALLY Y Y A A CALL AND AND A COMPANY A COMPANY AND A COMPANY AND A COMPANY A COMPANY AND A COMPANY AND A COMPANY A COMPANY A COMPANY A COMPANY AND A COMPANY AND A COMPANY A COMPANY A COMPANY AND A COMPANY A COMPANY A COMPANY AND A COMPANY	
The operation is executed only at the begins of the RLO and restant of edge of the RLO. The cold restant of arts only when the RLO a "1" at the east operation.	bertwe-going the timer re- turne of the
C 0 to 63 44 60 Y Y 4.5 Enable & counter for cold restart The operation is executed only of the p edge of the RLD The setting, up or do results only if the RLD a "1" at the un results only if the generation.	e of the cor-
Formal eparand 06 00 N N N	ert (For de- formal oper-
Formal operand 35 00 N N N 32.25 Start a timer apacified as a formal oper with the value stored in the accumulate type: T)	and as pulse Parameter
Formal operand 28 00 N N N	and as "On mulator (Pa-
Formal egerand 1E 00 N N N  32.25 Start a smar specified as a formal oper tanded pube with the value stored in t the value subsequently apecified type: T, Cl	end as an es he accumule mai operand di iperamete
U = Formal operand 2E 60 N N = 32.25 Start a timer specified as a formal oper "On" delay with the value stored in it tor an ecoment a counter specified operand (Parameter type, T, C)	end as stored to accumula as a forma
FD = Formal operand 16 00 N N 22.75 Start a timer specified as a formal operand data with the value stored in the sci data with the value stored in the sci decrement a courser specified as a for	and as "Off cumulator o mai operanc
D - Formal operand , 3E 60 N N I 22.75 Reset a formal operand for smarts and sameter type T, Cl.	
- garanteller salations	

www.falautomashari

Sprine		Parameter	- Comme	-		<b>g</b> a			Carrie		•	115	******	
			©``•	-		<b>1</b>	2						C.C.	
	_			<b>8</b> 1				21	851	8				
					1	800				6	So.		Chuck bit for "1" of	
TB	Ť	8.0 m 127.15	70	13	8		N		-	1	- 1	13.75	a terrer word	
<b>TB</b>	C -	8.0 m 63.15	70	1 15	0	. 80	' N	<b>N</b>	- 3	-	-	14.5	· a counter word	27
78	Ð	0.0 to 205.15	. 70	-	: 00	60	: N	N	-	<u> </u>	; -	22.0	e dete word	
<b>TB</b> _ 2	RS	0.0 to 255 15	70	- 57	: 00	: 60	! N	J.	- 1	-	-	19.25		2
TB	Ri	0.0 to 205 15	70	47	0	: =0	1.	I N	-	-	]-	19.75	mentaca data anto	
50		5	0	-			de.				d	S.	Check bit for "0"	
TEN	T	8.8 10 127.15	70	13			E N		- 1	-	S	11.75	et a terrar word	
TBN	C	8.0 10 63.15	: 70	1 16			N		] -	-0	¥-	14.5	of a counter ward	· · · · · · · · · · · · · · · · · · ·
TEN	Ð	8.0 to 285 15	1 70	1 46			1.00	N	-	12	-	22.0	of a data word	
TBN	RS	0.0 10 235.15	70	1.			N	IN	- 7	-	-	19.25	of system data area	1
TBN	- Rt	0.0 10 205 15	1 70	47	: 80	1 80	I N		1-	1-	1 -	19.75	i of interface data area	

operations loug Bit set Set bit unconde math SU T 0.0 to 127.15 · 70 ે 🕿 40 i 💼 N ٧ -12.75 of a senar word -_ Su ٤ 0.0 m 63.15 70 ; 15 80 N ¥ ¥ 13.5 40 of a courser word --Su D 0.0 to 286 15 70 23.0 4 • 60 N ۷ ---, of a data word ÷-SU **A**t 70.25 of interface data area 0.0 to 285.15 70 Ð 40 60 N ۷ ---et bit uncondi onex, Ru Ŧ 0.0 to 127 15 70 i 8 . . . ; 🛚 ۷ . . . -12.75 of a senar word _ Ru C 0.0 m 63.15 - -13.5 of a course word 1 70 15 60 -+ **Y** ---RU D 0.0 to 285.15 1 70 i • 1.00 60 1.1 1. 1_ -1 -23.0 I at a data ward 20.25 of interface data area RU. -0.0 to 255.15 : 70 ] 🗗 - i 📾 80 N Y - 1 - 1 ٠ i ٦ ł

						- 500							
5	•	Formel operand	17	00	- 3	<u>ser</u>	N	¥	-	1	<u> </u>	39.5	Set (breary) a former operand
AB	•	Forma: operand	37		35	-	N	۷	-	See.	-	38.5	Reset (binery) a fermal operand
•	•	Formal operand	1#	60	-	_	N	, <b>Y</b>	-	; -		<b>38.</b> 5	Assign the ratult of the logic operation to a formal element

11-53

							₁ .	1-54	•			at
	MAN ISBAUL			and the second	.1300 ¹³				Saca?	gpaure	~	Market Contraction of the second s
and and ten	for operations (suppl)			tions)								-
Currency Q	~~~~	-	-	1	•	1	32	-	inur anta 1947	•	I.	
			1 m1			2	21	æ	600	OV		
LIR *)	- 0 and 2	40	80		1000	N	N	-	†-	Saute	10.0	Land accs 1 (LIR 0) or accs 2 (LIR 2) with the cantants of the assury word addressed by accs 1.
TIR •)	- 0 and 2	48	80	and		   N	N	-	and	-	9.0	Transfer acco 2 to the assary word addressed by acco 1.
•) System and	rstim			aister a		1						
			2									
Jan Star	- Forma: operand	8 46 -	00		-	•.NO	, N	i <del>-</del>		-	2.75	Load a formal operand
	<u></u>					<u>)</u>	· ·			- 30	1 10.75	(Parameter type, I, Q, data type, BY, W)
	- Pormal operand	ut			80				(	80		: (Parameter type: T, C)
LW	Formal operand	¥	00	344	-	N	• <b>N</b>	· -	and and	-	19.5	Load the bit pattern of a formal operand. Perameter type: D. data type: KF, KH, KM, KY, KS, KT, KC:
T 😞	= Formal operand	- <b>*</b> -	80	-		, N	<b>.</b> .		, -	; <del>-</del>	32.25	Transfer to a formal operand ¹ (Parameter type, I, Q, data type, BY, W)
30 Maria	~	est.	' - pi	rameter	address	S.	K.				Carly	2 ⁻¹
L	RS 0 to 255	62	00	-	-3	N.		-	•	Ś	13.0	Loed a word from the system data area
	Ri 0 to 255	64	00	- ,	S.	N	N	-		8° - 1	13.0	Losd a word from the interface data area
+	RI 0 to 255	68	00	1997 - S. C.	-	¹ N	· N		32	! -	· 12.75	Transfer to a word from the interface data area
Conversion for	nctions laupplements:	and apprent	tions)		2	onat	Yox.			3	Scracht	a. tonauka.
CFW		01	00	-	<u>. 20</u> 7	N	N	-		<u>. 19</u>	1.75	
	AN CONTRACTOR						•	· ·	141 151		3.12	
Shih anarstia			1									
SLW 0	0 to 15	<b>6</b> 1 <	00	-	-	N	N	¥	۷	-	¥	Shift left (16 bris)
SRW	0 to 15	3.	80	-	-	N	N	٧	۷		2	Shift nget (16 brs)
			+ N	mber of	shifts	<u>o</u> nio						
Jump aperatio	s Sumboly	erationa 20	an (1	h	<u> </u>				<u></u>	<u>%                                    </u>	1.2	Unconditional witho
	address			-Star					Star.			Contract of the second
يالي. جالي		FA	00	-	-	• 	• 		-	-		Unite condition. RLO:
<u> </u>	· Symbolic		S. 90	-	-	N	R	-	-	-	6.5	Conditional jump Uump condition: CC1, CC01
JZ S	address			-	-	N	N	-	-	-	7.0	Conditional jump Usimp condition. CC1. CC0)
JZ	= Symbolic address	6 35	80								7.0	<u> </u>
JZ JN JP	address = Symbolic address = Symbolic address	<b>35</b> 15	00 00		1000	N	<u> </u>	-	-	18 T		- Unite condition, CC1, CC0;
JL JN JR JM	symbolic address Symbolic address Symbolic address Symbolic symbolic	35 15 25	00 00 00	-	- Carlon	N	: N   N	-	1. A.	<u></u>	7.0	Conditional jump
SL 9L 9L 0L	address = Symbolic address = Symbolic address = Symbolic address = Symbolic Symbolic	35 15 25 00	00 00 00 00	true True	, doo ^{li}	N	N	-			7.0	Conditional jump - Ulump condition: CC1: CC0; - Conditional jump - Ulump condition: CC1: CC0; - Conditional jump - Conditional jump - Ulump condition: CC1: CC0;
SL NL PL NL OL	address - Symbolic address - Symbolic address - Symbolic address	35 15 25 00	00 00 00 00	-		N N N <u>127</u>	N N	-	-	-	7.8 6.5	Conditional jump Using condition: CC1: CC0; Conditional jump Using condition: CC1: CC0; Conditional jump Conditional jump Using condition: OV;
SL NL PL ML OL	address • Symbolic address • Symbolic address • Symbolic eddress	35 15 25 00	00 00 00	-		N N <u>1</u> 127	N	-		100- 1-	7.0 6.5	Conditional jump Using condition: CC1: CC0; Conditional jump Using condition: CC1: CC0; Conditional jump Using condition: OV;

Other func	tions l		instione)						<del></del>				
Oper Ster		Peranata			-			6		itan anis 1117	•	i i i i i i i i i i i i i i i i i i i	funeur A
			NON			Nord 1		9 Ì		1		- 11	No.X
0		<u> </u>			12				- <del>.</del>		+ *	1.	Factoria commendia con ul
		<u> </u>	PE			-39	<u> </u>					3.7	Precise commente output
		- 38	- DE		- Dò	<u> </u>		· N	÷	120	<u> </u>	1 1 75	Eastis procession of access starting
		AND IN THE REAL PROPERTY OF			- 22					State of the second sec		1 76	
<u> </u>	- di				2				1				
D		0 to 255	19	00		-	- N	N	-		: -	; 3.75	Decrement
108		1 0 10 255	-bn	00			1 N.2	N	1 <u>-</u>	-	<b>-</b>	1 3.5	I increment
2		,ð	S.				Sec.					30	•
		105				- 30	5				30		<u></u>
DO	•	Formai operand	76	80	ī	Ser.	N	N	-	Ś	87	2.0	Process formel operand (parameter type B)
				<u> </u>	aramatar			İ					ALAS AN
DC	Dw	0 m 256	i ee	00		! -	1 N	N	! <b>_</b>	! _	! _	129.5	Process data word
				- 1	Native ad	dress	Nº	2					
DO	F¥	0 m 256	14E	00	-	1 –0	IN	: N	:_	: _	<u>کې</u>	20.0	i Process flag word
				<u>+ n</u>	Native ad	07855							
TAK *)	S.	2	170	E 82	1 ⁴⁰ -	-	i N	: N	- 5	<u> </u>		1 5.0	Susp the contents of acce 1 and acce 2
STS			70	OC.	-	-	N	N	-	_	-	4.25	5T0P
•) System	) ops751		No.9				.Ke	2				No.	2 2 2
J		0 to 31	60	80	-	-	N	T		-	- 6	.42	Unconditions) organization block call
r		0 te 31	40	90	-	1.50	• •	Y		-	39	22	Conditional organisation block call
		180°			38	S				<u>_</u> 8	0		J.C.

Arithmetic operations (supplementary operations)

, obaitomat

000 0-0"	Paramete	Ner		-			0	Care		•	1 L Š	Function	
		Ye .	Ward C	v	Alard 1	1	<u>.</u>		<b>186</b>				
20	^	80		82	: 43	22 I	2 į	CC1	223	Ov		and the second sec	
ADC	-128 to +127	50	00	_	130			-	-	32	3.75	Add byte constant (files-periot) to accu 1	
			<u> </u>	nstant (	(ő máts)		_						
	 Sea.			all'				in the second	E.				80
ABC	-32768 te +32768	56	00	80	60	N	N	-	-	-	7.0	Add word constant (fixed-paint) to accu "	

+ constant (16 bits)

- 12 PC Program Correction through NC Operator Panel
- 12.1 Generalities
- 12.1.1 Using the PC Program Correction
- 12.1.2 Prerequisites and Activation of the Program Correction
- 12.1.3 Key functions
- 12.2 Operation
- 12.2.1 Basic Display
- 12.2.2 Up and Down Load of PC Program
- 12.2.3 PC Initial Clear
- 12.2.4 Editor
- 12.2.5 Search
- 12.2.6 Activation of Blocks Stored in PROMs
- 12.2.7 PC-RAM Compression
- 12.2.8 Information Functions
- 12.3 Examples of Use
- 12.3.1 Display of ISTACK at PC-Stop
- 12.3.2 Program Correction for Test Purposes

#### 12.1 <u>Generalities</u>

#### 12.1.1 Using the PC Program Correction

Functions of the programing unit 670/675 can be partially accomplished on the NC-operator panel, with the aid of the PC correction program (PCdiagnosis). With its use, small program corrections or error analysis (ISTACK, BSTACK) can be executed, especially during service. The program correction can only be used with version 3; the NC software edition does not matter. When using the program correction, the changed user programs are stored in the free RAM memory of the PC. In the case of the 130 WA, care must be taken, that a free RAM range of 0.5k informations be present on the PC CPU for the user, and that it be used by user program.

The RAM memory on the PC CPU of the 130 WB is for 2.9k instructions.

#### 12.1.2 Prerequisites and Activation

Test board 03220 must contain software "PC Program Correction" (10 EPROMs on PROM location 71-80).

The board must be plugged into the location next to the NC power supply. Switch S3 on the NC-CPU board 03100 must be in lower position. Switches Debug and Diagn on test board must be in upper position. Switch NC ON.

The PC program correction is activated with the Diagn switch in lower position. The operator dialog is done via decision menus in text.

With active program correction, the PC program runs normally, but the NC software program is stopped.

If the PC program correction is operated incorrectly,the NC-CPU can go to stop-state (red LED on board 03100 lights up). Return into the PC program correction possible only with NC OFF/ON (hardware reset).

Jump of the PC program correction into normal NC software program: Diagn switch in upper position,then NC OFF/ON (hardware reset)

With the software "PC Program Correction", as described in section 10 of the start-up instructions, the NC engineering panel program is also available. It can be activated with the Debug switch.

# 12.1.3 Key Functions



Clear an input Enter an input Number range decimal O-9 Decimal-hexadecimal conversion Number range hexadecimal A-F Input switch to hexadecimal numbers Minus sign Change of a datum Insertion of a datum Search for a datum

--1 YES NO

Cursor 1 datum to the right Cursor 1 datum to the left Cursor 1 shift line down Cursor 1 shift line up... Answer of questions for operator dialogue Leaving the operating mode

(not possible during inputs in editor operating mode)

### 12.2 <u>Operation</u>

# 12.2.1 Basic Display

LOAD PC PROGRAM OUTPUT PC PROGRAM ERASE PC EDITOR SEARCH ACTIVATE EPROM BLOCKS COMPRESS PC RAM INFORMATION FUNCTIONS Function selection by the operator is generally done with the keys **YES** and **NO** or **1** and **1** 

The display is switched to the next possible operation function with the **NO** key.

The YES

key activates the selected operating mode.

. HOLDE					
12.2.2	Up and Down Load of PC F	o Programs			
	Select the operating mod	te with ope	erator dialogue <b>ve</b>	s and NO	
	CRT display:		onatkan!		
	READ-PUNCH COMBINATION STANDARD READER SELECT BAUD RATE	8	- Siemens PT 80 - Siemens reade - Baud rate sel	300 baud r 9600 baud ection: 200-2400-4800-04	500
		- Claightair	130-380-000-1	200-2400-4600-90	000
	PROGRAM BLOCK ? FUNCTION BLOCK ? ORGANIZATION BLOCK ? DATA BLOCK ? STEP BLOCK ?	at of the second	- Selection thr dialog <b>VES</b>	ough operator and <b>NO</b>	
	ALINO CONTRACTOR	2 ¹⁰			
	INPUT BLOCK NUMBER		-Entering with	the key input	$\Diamond$
	edtered	aller of		and the set	
	READY FOR INPUT (OUTPUT)	ALCON'	Start with key	YES	
	TRANSFER ACTIVE	Nº C	The data are lo	aded up and dowr	ר
	and the series of the series o	autome	through the V-2 of the NC.	4 (RS232) interf	face
	TRANSFER FURTHER BLOCKS (OUTPUT)	?	A A A A A A A A A A A A A A A A A A A	No.	
	<b>YES</b> Select page block select	ion Sel	ect basic page		

Remark: At the output of DBO, a list of the present blocks is outputted, similar to the "Output Director" with the PG 670.

12-5

# 12.2.3 PC Initial Clear

Select the operating mode through operator dialogue

BRING PC IN STOP STATE

PC switch to Stop position

"ERASE PC" NOT COMPLETED !

After approximately 3 seconds

"PC ERASE" COMPLETED PC SWITCH: 2 times from STOP to OPERATION

Switch the PC switch 2 times from Stop to Operation, then select the basic page.

### 12.2.4 Editor

#### Function Range

- Input of a new block with type and number
- Search for an instruction in the PC memory range
- Search for an instruction in the selected program block
- Load into the active memory, and display the block on CRT
- Display of the STEP-5 codes of organization, step, function and program blocks
- Display of the corresponding data values for the data blocks
- Display of the blocks corresponding to the addresses in DBO
- Change,erase and search for a present code (hexadecimal,decimal and mixed), insertion of a new code
- Automatic correction of the datum, block length in the block header, and the jump address at erasing or insertion of a code, if the jump designation is present
- Retransfer into the PC RAM, and change the block address in the address list (DBO)

# Read/Change Block Select via operator dialogue in basic page YES READ/CHANGE BLOCKS ENTER NEW BLOCK Read/change block YES BLOCK TYPE ? PB PROGRAM BLOCK Selection with SB STEP BLOCK FB FUNCTION BLOCK YES OB ORGANIZATION BLOCK DATA BLOCK DB Desired block number with input key BLOCK NUMBER .... (enter)

	-marking Ball		www.chaite	
BLOCK NOT FOUND ! LEAVE "EDITOR" MODE '	?	Selection with	YES / NO	
Select basic page	NO Select editor bloc	k type	www.chaltonia.	
OBOO1 FPROM FR	EE: 00364 WORDS	Supern function	•	
FFF6 7070 D001 80	00 0000 000D			3
0000 86EF 98FB 99	FB 3DOC 2D01			600
000A 3DC8 2D01 65	00	can be erased.c	ted with the cursor hanged,or a new dat	.um
0014		can be inserted		
001E	34 ²	(see the key de 12.1.3)	signation, section	
0028	. tomats	. HORIOL S		
FFF6 7070	18000			
INPLIT (H):				
OBOO1 EPROM FR	REE: 00364 WORDS			
FFF6 7070 D001 80	000 0000 000D			
0000 86EF 98FB 99	FB 3DOC 2D01	- Block header		
000A 3DC8 2D01 65	500	AF239.6 = F25	1.0 = F251.1	and the second
0014	4	JU FB12	ຼິ ງນ FB200	BE
001E	6			
0028	all the	Display of th	e selected datum i	٦
0000 86EF AF 23	39.6	STEP-5 code		
INPUT (H):				

E.g. Change of datum 86EF (AF239.6) in AN F239.6 ANF = AD (see the instruction list PC 130W,section 11.8)

Switch-over decimal/hexadecimal Input (H): H Hex-number A Input (H): A number 0 Input (H): AO Switch-over hexadecimal/decimal Input (D): AO Input (D): A0.2 number 2 Input (D): AD.23 number 3 number 9 Input (D): A0.239 Switch-over hexadecimal/decimal Input (D): A0.239. number 6 Input (D): A0.239.6 Termination of input, prior datum is changed Or newly entered datum is inserted Entered datum is searched for in the selected block

Operating mode terminated

LEAVE "EDITOR" MODE ?



NO

Call of block selected in operating mode editor

STORE BLOCK ?



Operating mode Editor is terminated. The block is stored in the RAM memory of the PC.If, prior to this, the changed block was stored in EPROM, it will now automatically activated in RAM (reactivation in the operating mode, 12.2.6)

NO

Op. mode Editor is terminated. The block is not stored! The basic page is selec-

ted.

Entering a New Block Operation is like in 12.2.4, block read/change Exception: after the block number is entered, a library number must also be entered. The block header of the new block is generated automatically. Alarms: "RAM FULL"- the block cannot be stored. Remedy: compress RAM EPROM LIST FULL 12.2.5 Search Select the operating mode via operator dialogue | YES | NO DATUM (H) SOUGHT As in the Editor operating mode (12.2.4), the input of the sought for datum can be done in hexadecimal, decimal or mixed. The input is completed with the Input key. All valid blocks in the entire PC memory range are searched for the datum. If the datum is not found, If the datum is found, e.q. AF 239.6=86EF (instruction list SEARCH COMPLETED see section 11.8) LEAVE "SEARCH" MODE OBOO1 EPROM FREE:00364 WORDS YES NO FFF6 7070 D001 8000 0000 000D Select basic page Select Search 0000 86EF 98EF 99FB 3D0C 2D01 000A 3DC8 2D01 6500 0000 86EF UM 239.6 DATUM FOUND! FURTHER SEARCH ? YES NO. The PC memory range is searched further for the "EDITOR" MODE"? desired datum YES NO Remark: Op. mode Editor is selected. Select All functions of Editor mode All blocks in the PC user basic page range are available program can be searched with code 6500 (BE). (section 12.2.4)



EPROM list. The block can no longer be generated in the "Activate PROM stored blocks" operating mode. Initial clear

Remedy:

# 12.2.7 Compress PC RAM

Select operating mode via operator dialogue



PC TO STOP STATE !

Push the PC switch from operation to stop



Switch the PC back to operation.



OPERATION

#### 12.3 Application Examples

12.3.1 Reading the I-STACK during PC Stop.

The PC goes from normal operation to stop-state; the red PC LED lights. Procedure:

PC switch to Stop

NC OFF

Board O3220 with program correction; plug EPROMs in. Switch S3 on the front plate of the CPU in lower position Diagnostic switch on board O3220 on lower position NC ON (display shows CS = 7E00 IP = OOEC)

Press key G

Press key **LF** (the menu page of the PC program correction is displayed) Select program test Information Function according to section 12.2.8 Read control bits, I-STACK, B-STACK.

#### 12.3.2 Program Correction for Test Purposes

For testing of the spindle, in the following example the inputs and interlocks of the PC user program are "jumpered". In the example, the spindle enable and hold key on the machine control panel are used on input 11.6, respectively 11.7.

Procedure:

- select operating mode "Editor"
- change block
- call OB1
- search for block end
  - input 6500 |

- cursor to the previous instruction



- Reset flag for the PC - NC interface RF 4.3



- Terminate with the reset key
- Leave the EDITOR operating mode
- STORE the block
- The changed program is in the PC RAM

The changed OB1 can be cleared again, and the original block can later be activated again, with the operating modes:

11

YES

YES

PC initial clear Activate the blocks in EPROM

**Input**
13. Alarm Description

#### Contents

- 13.1 Generalities
- 13.2 Alarm List
- 13.3 Alarm Descriptions

#### 13.1 Generalities

Alarm display for basic versions O and 2

	12	2	3	4	5	•	6	7	8	9	10	) 11	12	13	14	15	16	17	18	19	20	21	22	23	24	25	26	27	28	29	30	31	32
1	2	Γ	Γ	1	Τ	1	ī	0	1	X	c	L	A	M	P	I	N	G	3	С	H	E	c	ĸ			à	P					
			-			_	_		50	_		-					ð.	8			_		•			5	2						
2	L					5	5	0	4		В	L	0	С	K	S.	С	0	N	S	Т	R	U	<u>c</u>	Т	I	0	N		-		1	P
-	<u> </u>	r -	<u>.</u>	<b>T</b>	1	8					r	<b>r</b>			250	) 	_	,	<b>,</b> , , , , , , , , , , , , , , , , , ,	_	<b></b> _		N	$\sim$			<b>.</b>	-			-	25	<u></u>
5	A		A	R	ľ	1	1						- 3	1								3									2		

? The alarm light (nr. 4) can indicate one or several alarms. The first alarm number of the NC's alarms is generally displayed in the last three columns of the 3rd. line.A maximum of 4 current alarms are displayed in text (brief description of cause), in two sequential pages.



Incrementing the display number with the page key to further possible alarms

(Display number, 1 resp. 2)

Alarm numbers 1 - 248 are monitors of the hardware and machine state (also of external devices).

Alarm numbers 250 - 718 are monitors for operation, programing, decoding and processing.

The alarm acts as E-stop, it brings the axes to an immediate standstill (locked control loop)

The alarm can only be cleared with PORESET (power-up)

The alarm can be cleared with the RESET key (with M30 at program end)

The alarm can be cleared with the CLEAR key. (input clear)

BA...

Refers to sections of the operator manual

#### Alarm List

Clear	Nr.: J. Decade 1. Decade		7 Bit 6	6 Bit 5	5 Bit 4	4 8+t 3	3 1) 1 2	2 \$11_1	, Bit D
5	00	(doaue		J. C. Davis	Axis	Good		Axis	Axis
	01		4		Axis		4	Axis	fg Axis
	o 02		3.9		Axis		10 A	d Axis	Axis
	03	NOT OF	• •	KOLOS	Axis	, ofi ³	5	Axis	Axis
	10	Axis 1		1. Ban	Axis	Axis	Axis	Axis	Axis
	11	Axis Axis	1		Axis	Axis	Axis	Axis	Axis
	<u>्र</u> े 12	Axis	29.9		Axis 3	Axis	Axis	88 Axis	Axis
C.	13	Axis		torna	Axis	Axis	Axis	Axis	Axis
2	22	Spindle encoder fault		NIGOS .		Position con troitor of monitor of spindle	Emergency Stop	Control not ready	
	23	Time monitor V24 Interface	Overflow 2 Hardware Reader error	No stap-bit	Overflow Hardware Erro	Parity n	Control which trans- fer Reader	Overflow 1 Herdwere Erro	Stop-bit - parity error
	24	10mar		10mar	5	. one	N	Over- temperature	<u> </u>
	25	(1000)	Block w/o UF or with more then 120 charct.	ALODON'		Operating error V24 interface	Parity error in memory	Program not in mamory	Block net in <b>memor</b> y
	26		à		.d		. d	1)	SNS block not found
	27	Memory overflow	Discrepancy between memory and tape proor.	Tape format error	Tape input not alloued	Block with more than 120 cheract	Block parity error	Irrelevant EIA character	Character parity error
	28	Sub- routine error	Cutter point error	NI ICON		A. GOOL		WI CDOU	General decoding error
	29		False input parameter	False block order	False GO2/GD3	False radius value	False angle value	No intersection	False input value

can be cleared only with PORESET

#### <u>Alarm List</u> (continuéd)

Clear	Nr.: Dec. Dec.	8 811.7	7 Bit 6	6 81t 5	5	4	3	2	ait D
astorie	30	Circle end-point error	5	Joh	60	Zero or tool off set value error	610	Option not present	Circle no in selec- ted plane
50	31	Read - Che	loo many axes to be driven	No F-word or too large		Falsely programed lead		State of the	
	32		, de		1			Functions not allowed with selec ited CRC	La.A
autor		Ker			e and a second	auto	(ard)	alto	Carol.
Se.	34	Marth 1600		CARALLOC		Starte Con		ACAS AND BY	
~	35		2		a di				NC-start without ref.point
HOE.	50	2x axis,or more than 2 axes progr. rep.progr.f.	201	CRC/ contour error	Blueprint programing error	False block structure	Carlos	More than 6 geometry parameters	General
	51	Startan 1. 60°		Complete block can't be displayed	Preselected block nr. carrot be found	Block with more then 120 characters	Memory overflow	Input inhibited	Input only in Reset- state
٣	· 52	Ku-factor not deter- mined	Ky-factor of axas is not equal Monitor —+	•	10.		Hold at ¹⁾ thread		Strobe- input error
R.S.C	53	input error	Last progr. rot termine- ted	niy 2 aveş aliouev at playback	Playback only alloued at axes rest MDA E		Playback in 1st. block not alloued	Program nr. already used	Block with nore than 120 charac- ters
estart	70	and bo		Strand Boo		Stand Color		False activess code in machine detum	
	71		1 a.A		12.2				Battery alarm

- ** Input line (lowest line) must be completely cleared.
  Error 70X cannot be cleared with either RESET or CLEAR !
  Error 71X can be cleared with CLEAR !
- Alarm "Hold at Thread"
   Versions 0 and 2 up to software 05
   Version 3,software 01

## 13.3 Alarm Descriptions

arm numbei	Descriptio	on and Remedy		A ^{re}	4 ^{ch}
, 2,	Limit Swit	ches			
, 12,	These alar	rms are also tri	ggered when a s	software limi	t switch
, 22,	(value ent	ered under TEST	N160-N173) was	s reached (re	ference
, 32	point depe	endent setting).			
	The axis i	s stopped,but t	he position cor	ntroi loop re	mains
	closed and	the following	error moves the	e machine int	o position
	Ser.				
	In ioo mod	e.the machine m	ust be moved in	h the opposit	e direc-
	tion, and	the alarm must	be cleared with	n the red res	et key.
		W.S.	140.9		2
	arrato.		and and a second s		
	100 Merce	1021 July .			
	MALLO .				
	<u>A Moving A</u>	<u>xis has been Di</u>	sabled		
	This alarm	is issued if a	"Control Enabl	le" input sig	nal was no
	received.				
	- alle				
	CH LOD			and the second sec	AN CO
	All axes a	re brought to a	rapid stop and	the position	n control
	loop is lo	cked (EMERGENCY	STOP state).		
	all and a second				
	It should	he investioated	. why the inter	face has rev	oked the
	control en	able (see Inter	face Test. sect	cion 8.1).	
		Argen -	Al alla	And a	
	The alarm	is cleared with	the red reset	key; program	restart i
	afterwards	required.			
	Som.				
	AN .	Str.			

(

	- Ale				
Alarm number	Description and f	Remedy			
101,	Standstill Monito		AND CO	N. C.	and l
111,	The standstill mo	 Donitor is trie	opered when th	e set "Clamp Lin	nit"
121,	has been exceeded	d, or if the a	axis failed to	reach the posi	tion
131,	within the prescr	ribed time int	terval.	automath	
M. Obc	Possible causes:				
tonetykand	- An axis is push ces or faults mechanical port	ed out of pos in the contro ion, or the p	sition due to ol system, tac position contr	high mechanical no generator, mo ol loop hardware	for- otor, e.
M. Ball	- An axis cannot too high.	reach the pos	sition,e.g. be	cause the drift	is .
à	- A mechanically	clamped axis	has been push	ed out of positi	on.
outornatolico.	- see the start-up	p manual, sec	tion 11, N353	. shoreaster.	
MALODO	annet 1500				
12.Q	Remedy:				
. Strand	- The clamp limit	TEST N110 -	N113 must be h	nigher than the	
diam'r	position tolerar	nce TEST N100	- N103.		
end in					
- cashad	- The "Dwell Time enough to allow within the press	for Position for the elim cribed time sp	Monitor" TEST ination of the pan.	N353 must be l following erro	ong r

- The axis is pushed out of position by more than allowed under TEST N110 - N113 (clamping), by another moving axis.

		13 <b>-</b> 6		
	. Hart			
	NOTION NOT			
larm number	Description and Rem	edy	~	en la construcción de la constru
il.	Ster Andrews	Acada .		
02,	<u>Velocity Command Va</u>	<u>lue too High</u>		1 1 1 NAL -
12,	is issued when the	velocity comma	ind value genera	ated in the
22,	control is higher t	han permitted	under the "Com	nand Value Limi-
32	tation" set under T	EST N354.		
4	This can occur when	, for instance	, the motor ca	not follow the
	velocity command va	lue (the maxim	ium following e	rror exceeded).
	and the			
	Check whether the v	alue set under	TEST N354 is	about 20% higher
	than the "Command V	alue Limitatio	in" set under N	140 - N143.
	ALC ALC			
	8			
03,	<u>Contour Monitor</u>			
13,	This monitor trigge	rs alarms 103.	.133, and the	servos. are
23,	stopped by dropping	the velocity	command value	to Ø. Also, the
33	enable signals for	velocity contr	ol are revoked	, and the ma-
	chine is switched t	o follow-up ma	de.	
	The alarms are clea	red with the r	eset key.	
	official of			
	Alarms 103133 in	dicate, that t	he velocity co	ntrol loop opti-
	mization is inadeou	ate. or that t	he position lo	op gain is too
	high - as would cer	tainly be the	case if the va	lues set under
	TEST N351 and N352	ATE ZETOES.		
		Color		
	AD7 477	a thicograph if	) the telerance	band set under
	Alarms 103135 ar	e triggered in	wine accelerat	ions and decele-
	N352 is exceeded, (	or when, du	iring accererat	
			HES DOL FEACH L	ne new sheen
	rations of the driv	e, the axis do		
	rations of the driv within the K _V depen	e, the axis do dent time limi	.t. ala	all
	rations of the driv within the K _V depen	e, the axis do dent time limi	t.	utonable.P
	rations of the driv within the K _V depen	e, the axis do dent time limi	t.	automatike.P

A.A.A.

	AND D				
Alarm number	Description an	d Remedy	10 Mar	LOTTON'	
S	. Spar	S ^{ar}			
104,	Position Contr	ol Loop Hardware	e Fault		
114,	The monitor tr	iggera if the p	position contro	ol cable breaks,	
124,	if the position	n control signal	ls are missing	, or if a ground	
134	short occurred	· · · · · · · · · · · · · · · · · · ·			
	ADOULL .				
	Alarm activatio	on leads to an i	immediate stand	dstill of all axe	S
	(see the start	-up manual secti	ion 11, N353).		
	Check the posi	tion control cat	ole.		
	KOMB.				
	Ser .				
10	Drift too High				
115,	As long as the	drift (temperat	cure influences	s on components)	is
125,	not too high, t	the control is a	able to compens	ate for it	
135	(BA 8.6.5).				
Chourse	AD ^{OUT}	doautr		AD ^{DUTT}	
March 19	The alarms are	triggered if th	ne drift become	es larger than ab	out
¢	500 mV.	4°			
No.S.	NO.R				
Strath.	The "Not in Pos	sition" LED does	s not go out if	` the position co	n- 🥈
100000	trol loop or th	ne drive are not	ready.It also	remains lit in	the solution
ANTA. O	following cases	s: the servo dri	ve is inhibite.	ed, there is a ha	rd-
12 ·	ware fault in t	the position cor	trol loop or s	servo drive, or t	he
10.2	drift for the r	trive is not pro	nerlv adjusted	1	
- Alache			,, <u>-</u>	COLUMN STORE	
walter.	Execute the "di	rift compensatio	on" again (serv	vice manual, sect	ion stor
AND	5.5).				
ft.	(The drift memo	ory is displayed	under TEST N2	230 – N233.)	
29					
-Class.					
1021 HO					
		MAN OF			

Alarm number	Description and Remedy
N. C.	CO MICO MALCON AND AND AND AND AND AND AND AND AND AN
108, 🚿	Contaminated Scales
118,	When linear scales are used by the measuring system, when
128,	the scales are contaminated, the alarm is triggered by a hard-
138	ware signal from EXE (see the Interface Description, chapter 7)
	to the NC.

222

223

Control Loop Not Ready (Position Control - Input Signal) This alarm indicates a fault in the servo drive unit.It indicates if the appropriate input signal "Velocity Control Ready" (collective signal for all axes) is connected, and one drive unit is defective (e.g. fuse blown, overheating, etc). The alarm brings about a rapid stop of the feed drive; the "Control Loop Ready" signal is revoked.

Insert the P-N jumper "Servo Ready Simulation", if the signal is not being used by the interface (see service manual, sec. 9).

#### EMERGENCY STOP (E-STOP)

is issued in the presence of the E-STOP interface input signal. Check whether the E-STOP key was inadvertently pressed, or whether the machine moved to an emergency-stop cam. (Interface test nr. 8, byte 5, bit 7 = "0", * E-STOP is present)

The alarm brings the axes to a stop and inhibits the control.

	adhair adhair adhair
Alarm number	Description and Remedy
jest in the second s	N. C. M. C
224	Spindle Position Control Monitor
	Hardware monitor for the spindle
-utomatyle	It only is active when N4O7, bit 2 = 1 (spindle encoder present)
(GOST	This signal is issued when the input signals A, A*, B, B*, Z,
2	and Z* are faulty or missing.
	The alarm brings the spindle to a stop.
10Mari	softer, softer, softer, softer, softer,
BORD	The corresponding signals must always be of opposite polarity
4	(e.g. A ≠ A*).
-	

Spindle Encoder Contamination not implemented at the present.

ROD encoders have no contamination indicators.

Only for encoders with EXE 600/601 (linear systems, alarms 108...138).

228

# 13-10

	- Hage			
Alarm number	Description and Remedy	Monto	Morris	3
231, 232,	These alarms only react if th Causes:	e Siemens tape	reader is used.	
233 <b>,</b> 237	- reader electronic board MS6 - reader for 231, 237	UU, for 232 an	d 233 especially	
	- machine datum, in case of a	larm 231		
234	<u>Parity Fault</u> This alarm can occur only if	machine datum	411 or 412 bit 4	
	is set. The alarm is triggered when t (8 bit information + 1 parity	he data word f bit) has wron	rom the reader g parity.	
	This fault is totally unrelat characters of the tape (see a Check the machine data and ex	ed to parity e larm 271). ternal de <b>vic</b> e.	rrors of ISO or E	IA
235	<u>Overflow Error</u> This alarm is triggered when	the control re	ceives a new char	ac
	ter before it could store the	previous char	acter.	
	- Check machine data and exte - Error in USART interface - Cable	rnal devices		
236	<u>Stop-Bit Error</u> The alarm is issued when the	wrong number o	f stop-bits have	
	been set.	ternal devices	N.G.Ballott.	
	LNECK THE MACHINE DATA AND EX	CETHET DEVICES	· 344	

ka.A

	and the second				
Alarm number	Description and	d Remedy			
238	Time Monitor fo	or V24 (RS232C)	Interface	ennel C	anan C
~	This alarm is t	triggered if th	e NC is unable	to output or	
and the second	receive a chara	acter within 20	seconds.		
allone	Causes:				
Son	- External devi	lce is not powe	red up		
4	- Incorrectly o	connected cable			
Ś	- The external	device blocks	the CTS signal	for longer than	٦
asho	20 seconds.				
autor.				auton'	
20.	The alarm is al	so issued when	the control s	ignals (DC1 – DC	C4)
4	are used and th	ne NC receives	no DC1 (11 H)	within 20 second	ls
6	at data output.				
and the second					
auton'		auton.			
8.			NIGON CONTRACTOR		
42	Overtemperature	<u> </u>			
6	This alarm is t	riggered if the	e temperature	on the component	ts
and a construction	reaches the lim	it temperature	of 50°C.		
autorn'		- the NC Ready	1 cional in a	eveked The inter	
200	When this occur	s, the NC Ready	blo i o opl	v the currently	ac+
4		es the read to the	able, r.e. one	y the currentry	10
6	tive block is p				
all and a second	Check the fans	and air ducts.			
all office					
200-	If the internal	temperature of	f the control	is lower than 56	5°C,
1	check the tempe	rature switch o	on the CPU.		
Bahr.	If the internal check the tempe	temperature of rature switch o	f the control on the CPU.	is lower than 56	5°[

The trigger temperature range of switch S2 on the CPU is 56°C.

	and the fil				and the second
Alarm number	Description a	nd Remedy			
251	<u>Block not Fou</u>	nd in Memory	Marth C	and the second	and 1000
aska?	e.g. when jum	ping to a block	number.		
JHOM O	automo				
252	Program not F	ound in Memory			
4	The selected	part program ca	nnot be found :	In memory.	
all wall	201 ⁴⁰ .01				
253	<u>Parity Error</u>	in Memory			
, Ber	erent Goo				and the last
254	Operating Err	or for V24 Inte	rface (RS232 C)		
Glauconator	- V24 (RS232) tor panel - The code fo	lock is on, an r Siemens reade	d Data Start fi r is set in mac	com the PC or ope chine datum 412	ra-
20.01	1.0.21				
DET STRATT	Plack Missipo	LE on Containi	no more than 13	20 Characters	
257	BIUCK MISSING	itbout LE	ng more under 12		
	01 1102, 1130 0.				
			· · ·		
261	<u>No Coincidenc</u>	e Found during	Sequence Number	Search (SNS)	
autornu	The alarm is	triggered durin	g SNS if the so	ought for block o	r _{shore}
¹ Contraction of the second se	subroutine can the object of	nnot be found u the search (bl	p to the end of ock or subrouti	` the program, i .ne) is not prese	e. nt
6342.01	in the progra	n memory.			
Ballon	ANNIGROUTORN AND	WIGHERTON'			

• •

)

24

)

	Cally Car				
Alarm number	Description	and Remedy	- allon	Allon .	
262	Fault at The This alarm f occurred due thread (hold For versions	<u>read</u> indicates to t ring thread cu d in feed per s O and 2, sta	the operator, th	that an interrup ing in damage to ftware edition O	tion the ,
A. A	For version	3 starting wi	th software e	dition O2,this i	s dis-
202	played as al	.arm 523.			
SC 351	an all the				
774	Chapaster Dr	nity Ennon			
ALL AND	checked; it 271 is trigo	must correspo ered if dispa	nd to the esta rity is found.	ablished parity.	Alarm
all	EIA is odd p	arity.			
autonne	130 13 60011	purrey.			
х х					
272	Inadmissable	Hole Combina	tion of an EIA	<u>Character</u>	
N.2.2	Alarm is iss	ued despite c	orrect parity,	if a character	is
1CAL	and the second states	FIA code and	it has been r	and S	
NOTON.	nucerined in		IC Has Deen I	eau.	
Butomaci		. dballon	IC Has Deen I	eau.	
allomon with	Dicel: Decitu	Mrdballon	It has been i	eau.	
273	Block Parity	<u>Error</u>	itor (setting	datum) is activ	ated
273	Block Parity When the blo	<u>Error</u> ck parity mon	itor (setting	datum) is activ	ated, s not a
273	Block Parity When the blo all the char even number,	<u>Error</u> ck parity mon acters of a b alarm 273 is	itor (setting lock are count triggered.	datum) is active ed.If the sum is	ated, s not a
273	Block Parity When the blo all the char even number,	<u>Error</u> ck parity mon acters of a b alarm 273 is	itor (setting lock are count triggered.	datum) is activ ed.If the sum is	ated, s not a ity, ir
273	Block Parity When the blo all the char even number, The control dependently	<u>Error</u> ck parity mon acters of a b alarm 273 is always genera of the setting	itor (setting lock are count triggered. tes tapes with q datum; for t	datum) is activ ed.If the sum is even block par his purpose,if p	ated, s not a ity, ir needed,
273 markad	Block Parity When the blo all the char even number, The control dependently blanks will	<u>Error</u> ck parity mon acters of a b alarm 273 is always genera of the setting be issued.	itor (setting lock are count triggered. tes tapes with g datum; for t	datum) is active ed.If the sum is even block par. his purpose,if e	ated, s not a ity, ir needed,

tornatolic.	a some years	t Blonashah			
rm number	Description a	ana kemeuy 			
	Block with mo	ore than 120 C	haracters		
2	If a block wi	ith more than	120 characters	is read, alar	n 274 is
marth	issued.Only t	the stored cha	racters are cou	unted, i.e. CR	,
20.	sprockets, sp	aces (blanks)	with the excep	DTION OF COMME	nts, are
	not countea.				
	Remedy: brake	e the block in	to several bloc	cks.	
140.Q					
offar'	- tornar	La ronal			
5	lape Input Di	<u>sabled</u>	. Bar		
4	Alarm 275 15	triggered i	- Alan		
onar	tum "Key s	witch active	during input of	`part programs	s" being 🦽
tonac	tum "Key s set.	witch active	during input of	`part programs	s" being
Jonac'	tum "Key s set. b) The data s	witch active afety switch	during input of S1 on the CPU i down-loading of	` part programs .s not in "free of machine data	s" being e" a "TE".
ional and	tum "Key s set. b) The data s (upper) po	witch active afety switch osition during	during input of S1 on the CPU i down-loading o	`part programs .s not in "free of machine data	s" being e" a "TE".
Jonacka di	tum "Key s set. b) The data s (upper) po	witch active afety switch sition during	during input of S1 on the CPU i down-loading o	`part programs .s not in "free of machine data	s" being e" a "TE".
onadka pi	tum "Key s set. b) The data s (upper) po <u>Tape Format E</u>	witch active afety switch sition during	during input of S1 on the CPU i down-loading o	`part programs .s not in "free of machine data	s" being e" a "TE".
conadiana a	tum "Key s set. b) The data s (upper) po <u>Tape Format E</u> Alarm 276 is	witch active afety switch osition during	during input of S1 on the CPU i down-loading o	`part programs .s not in "free of machine data	e" being e" a "TE".
conaction and	tum "Key s set. b) The data s (upper) po <u>Tape Format E</u> Alarm 276 is	witch active afety switch osition during <u>frror</u> triggered whe	during input of S1 on the CPU i down-loading o n :	` part programs .s not in "free of machine data	e" being e" "TE".
Jonatika pi	tum "Key s set. b) The data s (upper) po <u>Tape Format E</u> Alarm 276 is a) The allowe	witch active afety switch osition during <u>frror</u> triggered whe ad number of d	during input of S1 on the CPU i down-loading o n : ecades followir	` part programs .s not in "free of machine data ng an address :	s" being e" a "TE". is in-
conational stand	tum "Key s set. b) The data s (upper) po <u>Tape Format E</u> Alarm 276 is a) The allowe correct.	witch active afety switch osition during <u>trigg</u> ered whe ed number of d	during input of S1 on the CPU i down-loading o n : ecades followir	`part programs .s not in "free of machine data	s" being e" a "TE". is in-
conational s	tum "Key s set. b) The data s (upper) po <u>Tape Format E</u> Alarm 276 is a) The allowe correct. b) A decimal	witch active afety switch osition during <u>triggered whe</u> ed number of d point appears	during input of S1 on the CPU i down-loading o n : ecades followin in the wrong p	` part programs .s not in "free of machine data ng an address : position.	s" being e" a "TE". is in-
Jonaskan	tum "Key s set. b) The data s (upper) po <u>Tape Format E</u> Alarm 276 is a) The allowe correct. b) A decimal c) Part progr	witch active afety switch osition during triggered whe ed number of d point appears ams or subrou	during input of S1 on the CPU i down-loading o n : ecades followin in the wrong p tines are termi	` part programs .s not in "free of machine data og an address : position. .nated or incom	s" being e" a "TE". is in-
Jonathan	tum "Key s set. b) The data s (upper) po <u>Tape Format E</u> Alarm 276 is a) The allows correct. b) A decimal c) Part progr defined.	witch active afety switch osition during <u>triggered whe</u> ad number of d point appears ams or subrou	during input of S1 on the CPU i down-loading o n : ecades followin in the wrong p tines are termi	` part programs .s not in "free of machine data ng an address : position. .nated or incom	s" being e" a "TE". is in-
ionadian	tum "Key s set. b) The data s (upper) po <u>Tape Format E</u> Alarm 276 is a) The allows correct. b) A decimal c) Part progr defined. d) False form	witch active afety switch osition during <u>triggered whe</u> ad number of d point appears ams or subrou	during input of S1 on the CPU i down-loading o n : ecades followin in the wrong p tines are termi earing of progr	part programs s not in "free of machine data og an address : position. nated or incom	s" being e" a "TE". is in-
	tum "Key s set. b) The data s (upper) po <u>Tape Format E</u> Alarm 276 is a) The allowe correct. b) A decimal c) Part progr defined. d) False form	witch active afety switch osition during <u>triggered whe</u> ad number of d point appears ams or subrou	during input of S1 on the CPU i down-loading o n : ecades followin in the wrong p tines are termi earing of progr	part programs s not in "free of machine data og an address : position. nated or incom	s" being e" a "TE". is in-

cito?

143.9

arm number	-   Description ar	nd Remedy			
<u></u>		Start .	Span of the second seco	dunde.	.80 ⁰
7	Stored Program	m ≠ Tape Progra	m		
	If a tape is 1	read more than	once, its conte	ent is compared	d block
	by block to the	ne program stor	ed during the f	first reading.	The
	alarm 277 is t	triggered if an	y disparities a	are found.	
	This alarm is	also issued at	any attempt to	o store a prog	ram
	under a progra	am number, unde	r which another	r program has a	already
	been stored.			·	
	In such cases	, the previousl	y stored progra	am must be clea	ared.
	Unlike during	storage of pro	grams, when pro	ograms are com	oared
	the "Available	e Memory" numbe	r remains uncha	anged.	
	1. 1				
8	Memory Overflo	ow			
*Offab	If the memory		insufficient d	uring down-loa	ding,
	alarm 278 is	triggered.	. Baue		
	The memory spa	ace still avail	able for stora	ge can be chec	ked
	via the "Avai.	lable Memory" n	umber.		
	If necessary,	irrelevant pro	grams may be e	rased and the	pro-
	gram must be :	read anew.			
	Span.				
	en e				
	Irreparable P	rograming Error	<u>`S</u>		
1.000	General				
	The error is	displayed in th	ne "Display of	the Correction	Block"
		additional com	ima underneath	each character	· seren ·
	page, with an				
	ALC .				
	NOTOC'				
		W.			

Alarm number	Description	and Remedy			ROUND SAL
AN.C	N.C.	CHILDER CONTRACT	and the second s	NN. C	WI GDOL
287	Path Inter	rsection Error	- 312		
	The alarm is	s triggered at	errors in the	programing of t	he
	stock remova	al cycle L94,	when the parame	et <b>ers</b> entered ar	re false.
288	Subroutine E	FFOF			
	– M17 in par	rt program 🔬			
bastomatile	- Excessive	nesting depth	halfonante		halfonatik
	. Be				

## Errors in Blueprint Programing

False Input Value

The programed values lead to overflow during calculation, or cannot be calculated due to false dimensions or procedure.

291

# 13**-**17

)

)

)

er	Description	n and Remedy	Roperton Contraction	- Andrew -	
Sara O					
550					
10	<u>No Intersec</u>	ction			
	When calcul	lating the el	ements of the c	lescribed contou:	r, the
	programed v	alues fail t	o result in an	intersection.	
ŏ					
332	False Angle	e Value			
	Alarm indic	ates angles	larger or equal	to 360°, or va	lue un <del>-</del>
	reasonable	for the desc	ribed contour.		
ŏ					
-3535	<u>False Radit</u>	is Value			
	The input v	alue is too	large or inadmi	ssable for the o	describe
	contour.			. Aller	
		autonin			
č,					
4520	False GO2/G	03			
	Circular di	rection is n	ot possible for	the described	contour.
ja La constante La	False Block	Sequence			
35	Several blo	ocks are nece	ssary for calcu	lations :	
	The sequence	e is incompa	tible, or there	is insufficient	t infor-
	mation for	the calculat	ion.		
6 					
355	False Input	Parameters			
	The program	ned parameter	sequence is in	admissable or i	ncomplet
	in view of	the describe	d contour.		
, all		CNIG.			
	Anna C	False AngleAlarm indicreasonableFalse RadiuThe input wcontour.False GO2/CCircular diFalse BlockSeveral blocThe sequencemation forFalse InputThe programin view of	False Angle ValueAlarm indicates anglesreasonable for the descFalse Radius ValueThe input value is toocontour.False G02/G03Circular direction is nFalse Block SequenceSeveral blocks are neceThe sequence is incompamation for the calculatFalse Input ParametersThe programed parameterin view of the describe	False Angle Value         Alarm indicates angles larger or equal         reasonable for the described contour.         False Radius Value         The input value is too large or inadmi         contour.         False G02/G03         Circular direction is not possible for         False Block Sequence         Several blocks are necessary for calcu         The sequence is incompatible, or there         mation for the calculation.         False Input Parameters         The programed parameter sequence is in         in view of the described contour.	False Angle Value         Alarm indicates angles larger or equal to 360°, or value reasonable for the described contour.         False Radius Value         The input value is too large or inadmissable for the or contour.         False G02/G03         Circular direction is not possible for the described of false Block Sequence         Several blocks are necessary for calculations :         The sequence is incompatible, or there is insufficient mation for the calculation.         False Input Parameters         The programed parameter sequence is inadmissable or in in view of the described contour.

	asked asked asked asked
Alarm number	Description and Remedy
301	<u>Circle not in the Selected Plane</u> 3M: the interpolation parameters are incorrect for the selec-
	ted plane.
302	<u>Non-existent Option</u> This alarm is issued under the following circumstances:
	- Option "Feed per Revolution" is not present, <u>and</u> G95/G96 is programed
	- Option "Thread" is not present <u>and</u> 633 is programed Remedy: Check software extent and machine data
304	Zero Offset or Tool Offset: Inadmissable Value Double word overflow is possible with six or more decade values.
308	<u>Circle End-point Error</u> The programed circle end-point lies outside the circle. The alarm is triggered when the end-point lies outside the tolerance specified by the machine datum N355 ("Circle End-point Monitor").

314

## Thread Lead Incorrectly Programed

The thread lead is programed under I, J, <u>or</u> K, and <u>always</u> refers to the <u>leading</u> axis, because alarm 314 is otherwise triggered.

(e.g. X 20 000 Z 10 000 K 1000)

Warn. Obi

mashart	I omenkan	nankan			
Alarm number	Descrip	tion and Remedy		- B ^{alle}	
RAC MARK	2			And all	
316		not Programed	or programed a	t feed per rev	olution.
	L.Y.	o i yec in program	or programed a		
	autonic				
240	So T	Auge Decement			
317	This al	arm is issued when	more than two	axes should mo	ve.as
	unuld h	e the case with 64	1/642.	2,02 0,1222	,-
	a contra -	Calific	- Calife		
	102110.	20 ²⁰¹⁰⁰	doauto.		
-22	Inadmis	sable Block at CRC	resp. TRC		
	With a	selected CRC or TR	C, G92, G33, M1	9 or G59 may n	ot be
	program	ed.			
	KOLLOL.	tomat.			
	Remedy:	Program G40 or	- Melan		
	2	G41/G42, DOO (CR	C cancelled)		
	, à				
	Carlyle				
	10auton				
	10 ⁵				
	AKO.P				
	tor out	tomac	atomat.		
351	NC Star	t without Referenc	<u>e Point</u>		
	~				
	6				
	A - 3				
1844a.0	all				
waitemastra.pl	ballonad ka.				

**F**à

		13-	20		
	asha?				
larm number	Description	and Remedy			
	3 ¹				- Contractor
4	ALARMS 501.	.538 CAN BE C	LEARED WITH THE	CLEAR KEY	4
120.01	19.9	. A			
anatol.	<u>Reparable P</u>	rograming Erro	rs		
10aure					
U1	<u>Leneral</u>		the "Display o	f the Correction	
41	Ine error 1	s indicated in	nc) with an ad	ditional charact	eristic
-N2.9	page (brock	Delote decour	ng), wron an au		
xorrans	xoff ^{als}				
.02	More than 6	Geometry Para	meters		
102 A		ed in one bloc	k.		
~	(Geometry n	arameters are:	axes. interpol	ation parameters	δ,
- SHO.P.	(decine cr) p	all and the second	radii, angles)	. Here	
tonio	tornia		Morne		
So.	1. Bar	16000			
504 📣	False Block	Structure			
8	E.g. N10 GO	2 X1000 LF	(missing inte	rpolation parame	eters)
all Marin	N20 GO:	2 Z2000 I20 LF	(I2O not allo	wed in this bloc	sk)
autornu					
05	Error in Bl	ueprint Progra	ming		
4		14 M			
	CRC resp TI	R Contour Err	or s		
	The interme	diate block is	too small for	the selected off	`set, or
wall of	the travell	ing direction	resulting from	the offset calcu	lation
<u>8</u>	is opposite	to that progr	amed.		
4	- F F	W.			
12.2	Rey				
08	Axis Progra	med Twice or	more than 2 Axe	s Programed	
10auto	abauto.			abauto.	
$\otimes$					

dia.p

aka di

CHQ.P

dra.d

arka A

#### 13-21

Alarm number (	Description and Remedy		
Solution and the solution of t			
511	Input only in Reset State	AND AND	
6	The alarm is issued if inadmissab	le functions are sel	lected in
- STA	automatic mode.These functions ca	n be employed only i	if the
autonia	reset key is first pressed.		
S ^{oo}	The alarm can be cleared with the	key.	
à			
512	Input Inhibited		
autor.	Input is only possible if the key	switch is in "open'	' position
So.c.	or if the S1 data safety switch o	n the CPU 03 100 is	in upper
and a second	position.		
N2R	The alarm is also triggered if in	put is attempted in	the wrong

513

## Memory Overflow

This alarm indicates that the program memory is full. Programs not in current use may have to be erased (see BA 7.1). These procedures can be undertaken only while the control is in reset state.

- 514

## Block with more than 120 Characters

During editing, the number of characters in a block is checked before storing. (The 120 characters must include LF) The only remedy is to split the large block into several smaller ones.

					STIRE.
Alarm number	Description a	nd Remedy	- dballe	dhaute	
515	Selected Bloc	<u>k Number not F</u>	ound		
	This alarm in	dicates that t	the selected bl	ock number cann	ot be
	found in the	program during	SNS.		
	The cursor is	set to the pi	rogram start an	d the program i	s John
	displayed.				
	14 A A				
	6				
516	Block cannot	be Displayed i	n Its Entirety		
	Even if the a	llowed block l	ength was not	exceeded (120 ci	harac-
	ters), in ce	rtain configu	urations , not	all the charac	ters 🖉
	can be displa	yed.			33 ⁵⁵¹
	Remedy. Tt i	e possible to	oenerate tuob	locks (the seco	nd of 💧
	Nemedy. It I	b bas po block	pumber), by i	nserting an "LE	н. З ^{кол}
		tuo blocks car		d so that a fu	nctio-
. Close					
	TIAL		produced.		
E01	Ennon et Stro	be Ipput			
521	This planm is	trippered at	external data	input if the co	de is
		rd is too loor	or %input is	made durino a	running
		10 13 000 100	, ur <i>minput</i> re	made daring -	
	program.				
507 Mag	The second		m nn 262)		Je?
523	Hold at Inrea				
500		at any of the A	AND ADDRESS		
527	Unequal KV Fa	ctors of the r	nxes	otors of the av	es are
	Ine alarm 15				
	not equal, be	cause this ine	equality can le		EVIA-
	tions.				
	10 alle				
	and C.	Marilo.			
528	Kv Factors no	t Calculated	24-		24
	Occurs as a r	emainder after	power-up and	machine data ch	ange.

		13-23			
	- Walt				
Alarm number	Description a	nd Remedy	Hornals	Homats	utomati
and Constanting and	MDA Alarms	ANICE STREET	And M. ODO		
531	<u>Block with mo</u>	re than 40 Ch	aracters		
	N. dball				
532	Program Numbe	r already in (	Memory		
	aller a				
533	Playback Inad	missable as F.	<u>irst Block</u>		-automia
	A program num	ber must firs	t be opened in d	teach-in mode	, and
·	it must be co	rrectly store	<b>.</b>		
	The Sharps			anatyles	
534	Playback only	Allowed if M	e stored in MD	A.	
			Alman		
535	Playback only	Allowed when	the Axes are	Stopped	
	No axis motio	n may take pl	ace while the	block is being	stored.
	10 ¹				
536	More than 2 A	xes in a Bloc	<u>k</u>		
- addent	More than two	axes cannot	interpolate.		
	108110C			aballon.	
	all and				
	3				
		diante			
		la.			

Alarm number	Description and Remedy	10 ^{21/10}	- Alle	100 ⁰⁰⁰⁰			
S. W. M. C.	State Stat						
537	Last Program not Terminated						
1. C. R.	appears as a warning when a	new program i	s opened under	MDA,			
anals.	before the old program has b	been terminate	ed.				
AD ^{RUIC}							

538

702

## General Input Error in MDA

False Address Code in Machine Datum Machine data must be changed. (Name, axis.)

#### Battery Alarm

Check the voltage of the battery on 03500 (power supply); if necessary, change the battery.

<u>Important</u>: The battery must be exchanged under voltage (while the control is powered up), so that the C-MOS Memory 03210,resp. 03260, is not erased.

Note: In version 2 (with PC), when the battery fails the PC goes into stop state, and thus the NC also stops. The alarm is therefore not triggered under these circumstances (see the service manual, section 3.3.12).

711

#### 14 Basic Version O

#### Brief Start-up Instructions

#### Contents

- 14.1 Prerequisites
- 14.2 Setting the Standard Machine Data
- 14.3 Adaption to the Machine
- 14.4 Adaptions of the Velocity Related Machine Data Before the First Travel
- 14.5 Setting the Control Sense for the Axes
- 14.6 Moving the Axes
- 14.7 Functional Adaptions
- 14.8 Remarks Concerning Erroneous Inputs and Erasing the Memory Ranges
- 14.9 Conclusion
- 14.10 List 1: Axis Specific Machine Data
- 14.11 List 2: Common Machine Data
- 14.12 List 3: Machine Data Bits
- 14.13 List 4: Possible Devices for Data Input and Output

#### 14.1 Prerequisites

Check - 24 V input voltage on power supply 03500 (+24 V, 0 V)

- 24 V input voltage on operator panel (+24 V, O V)
  - 24 V input voltage on machine control panel (+24 V, O V)

14-2

The position control cables (command and actual value cables) should be unplugged.

The tacho should be adjusted for maximum velocity according to 8 V velocity command value.

Check on the installation of position control boards 03310 and 03320.

14.2 Setting the Standard Machine Data

Operati	ng mode		M	DI-SE-TE
3T: 3M:	key key	$\widehat{\diamondsuit}$	and and	$\frac{3}{k}$

both keys pressed simultaneously, NC power-up

#### 14.3 Adaptions to the Machine

Only machine data whose adaption to the machine is absolutely necessary are handled.For standard values, maximum values and units, see lists 1 and 2.

Operating mode  $\diamondsuit$  MDI-SE-TE Upper position for the switch on board 03100, select  $\fbox$  Test Input e.g. >160 ..... Input figures 0..9 e.g. >403 ..... figures 0.1

#### Cursor

If the machine's output system is in inch (ball screw, position coder, machine data), see the start-up instructions in section 11.4 .

#### 14.4 Adaptions of the Velocity Related Machine Data, before the First Travel

In the presence of deviations from the standard values, the input of machine related values is required.

nautomatika	hautonatikan		14-3						
14.4.1	Maximum Axis Velo	city Stand	ard: 1000	)0 mm/mir	ı				
	3T >130 S	··· 🗖 X-a	xis	3M	>130 S	•••		X-axis	
	>131 5	🚺 Z-a	xis		>131 S	[	$\mathbf{O}$	Y-axis	
					>132 S	۲	¥.	Z-axis	
14.4.2	Software Limit Sw	itch Stand	ard: limi	t switch	n inact	ive, +	direc	tion	
	<b>3T ≥160 S</b>	···· [ ] X-a	xis	3M	>16D_5	454		X-axis	
	>161 S		xis	011	>161	[	$\overline{\mathbf{A}}$	Y-axis	
					>162 S		$\mathbf{\nabla}$	Z-axis	
- ADBILLON			Minus	Directi	.on				
		Sch ¹ O		700	N 470 C			V	
<u>)</u>	31 >170 5	··· 🕹 🕺	X1S 📣	311	>170 5			X-axis	
<u> </u>	- 71/13	•••• •••• 2-8	X15		×172 G	[	M (		
California -					2112 3	•••		2-ax15	
14.4.3	Reference Point V	alues Stand	ard: O						North Chi
	liber the reference		pppochod	the set	`~~~~~~	o anint	volue	.ie .	and Or
	transferred into	the actual v	alue.		erence	μοτιις	varue	18 4	
	3T >180 S	\Lambda X-a	xis	3M	>180 S			X-axis	
	×181 S	文 _{Z-a}	xis	, Ĉ	>181 S	[	$\mathbf{N}^{(n)}$	Y-axis	
					>182 S		$\mathbf{\nabla}$	Z-axis	
14.4.4	Velocity Adaption	Standard:	8 V, comm	and valu	ie = 10	000 mm	/min		
Je star			maximum a	xis velo	city a	t 8 V	comman	d value	. Hand
and Constantion of	V max (m/min) at 8 V command value	15 12	10 8	6	5	4	3	1	antibationac.
2	Input Value 🔬	1600 2000	2400 300	0 4000	4800	6000	8000	12000	2
	3T >220 S >221 S	X-axis Z-axis	¹⁹ 29	3M	>220 S >221 S >222 S	 	$\overline{\mathbf{O}}$	X-axis Y-axis Z-axis	weit of Dautomaster
www.chailonayka.c				and dealterne					



#### 14.6 Moving the Axes

Plug in all the cables! The following signals must be present: Control Enable, Feed Release, No Emergency Stop, No Axis Lock (check via the interface diagnosis, see the operator manual).

#### 14.6.1 Drift Compensation

see the operator manual section 8.6.5 . The value is entered automatically in 📈 nr. 230 through 232.

#### 14.6.2 Adaption of Travel Distance to Command Distance

Move 10 mm i	n _o perating	mode "In	ncrement"
The distance	travelled	by the a	kis is :
automic	10 mm	20 mm	5 mm
Bit 3	٥ 🖉	1	0
Bit 4	0	O	1
L			

Standard

43 Bit

Nr. 403 through 405 S____

3T >403 S... X-axis >404 S... X-axis

3M >403 S... X-axis >404 S... Y-axis >405 S... Z-axis

14.7 Functional Adaptions

>408 S 00.01001 14.7.1 Hand Wheel Activation

14.7.2 Device Coding for Data Input/Output Standard: unit with 300 baud and 2 stop bits

For the adaption of other devices, see list 4.

>411 S_____ Data Input >412 S_____ Data Output

14.7.3 Functional Options

Individual bits according to start-up list or control data sheet.

#### 14.7.4 Automatic Determination of the Position Loop Gain ( $K_V$ Factor)

Move each axis in manual mode (M) (1), 30G, 100%, for about 4 seconds. The calculated K_V factor is displayed under test (M) nr. 850 - 852. In continuous path control, the values of the participating axes must be equal. Any deviation of more than 50 leads to alarm 827. In such cases, all the K_V values entered as machine data nr. 150 - 152, or the tacho adjustment or machine data 220 through 223 are wrong.

The Kv factors are cleared each time the machine data are changed.

#### 14.8 Remarks

The machine can be optimized exactly by following the extensive instructions given in the "Service Manual SINUMERIK System 3". The input of erroneous machine data may lead to activation of the red LED on board 03100; in such cases, return to section 14.2 . If power supply 03501 is disconnected, the battery voltage will be lost. The following cancel operations are necessary for the reactivation of the control:



Clear Machine Data Clear User Program Clear Setting Data

The NC should be powered, and the cancel and number keys should be pressed simultaneously.

The machine data must then be entered anew.

#### 14.9 <u>Conclusion</u>

The toggle switch on board 03100 should be in lower position. Generate a machine data list and/or machine data tape and place it next to the control.

Test all functions of the machine and all operating modes.

14.10 List 1 : Axis Specific Machine Data

Nr.	Standard values set with "Input"	Axis	Entered values 3T	Axis	Entered values 3M	Explanation	Maxìmum value [ unit ]
100	50	X		×	L.X.X.X.X.X.X.X.X.X.X.X.X.X.X.X.X.X.X.X	Position	32000
101	50	z		Y	<u></u>	tolerance	[um]
102	50		~	3		10	S.
103	50			4	2	e de la companya de la compa	1000
110	200	X	and the second s	X	- All		32000
111	200	Z	20	Y	34	L rraub trute	[47]
112	200		$\sim$	2	· · · · · · · · · · · · · · · · · · ·	1	
113	200	-		4	0	6	6
120	50	X		X	Ke .	Acceleration	6000
121	50	≥ Z		্রু			[0.01m/s ² ]
122	50		-	Z		30	30
123	50			4		< °	
130	10000	X	A.	X	all in the second se	Max:Velocity	15000
131	10000	2	385	Y			[mm/min]
132	10000			2		-	
133	10000		$\leq$	4		<u></u>	<u> </u>
140	2048	X		X	N.C.	Velocity	2048
141	2048	<u> </u>		<u> </u>		command	[AET O]
142	2048					limitation	
143	2048					<u></u>	<u></u>
150	1666	X		X	11	K, factor	10000 En es-17
151	1666					4	[0.015 ]
152	1665					-	
153	1000				-		
100	+9999999	2			5	Limit switch	[m]
161	• 3333333	<u> </u>				+	[ <b>1</b> ]
162	- 9999999			-		- 300	
170	-9999999	¥ I	<u> </u>	Y	3.	1	0000000
171	-9999999	7		Y	1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 -	Limit switch	[un]
172	- 9999999			z		*	
173	- 9999999			4		(minus)	
180	0			X	18 ⁸	D. C. NOV	+9999999
181	0	Z		۲ă	5	Reference	[H]
182	0		/	ČZ.		point	. S
183	0			4		S. S	and the
190	0	X		X		Backlash	255 ±255
191	0	2	de la companya de la comp	Y	3	Dackidsh	[#]
192	0			2	1	Compensarion	
193	0		$\sim$	4			
210	0	X		X	. A.Y.	Reference	±9999
211	0	Z		۲.	5	noint shift	[47]
212	0			2			100
213	0			£.		N	
220	2400	X		×	<u></u>	Mult gain	32000
221	2400	Z		¥		- 2	[CXmin/m]
222	2400		$ \rightarrow $		~~	-	
223	2400			+ •		Deter	
230	U 10	-		+	<u></u>		+500
231	0				<u></u>	_ compensation	lverg
222	10			1		- 10 m	201
1 633	- N			P 7	ł	N.	10 N

Nr.	Standard data set via "Input"	Input values 3T	Input values 3M	Explanations	Max. Value [ units ]
350	500	, K	UTO T	Cut-off velocity	15000 [m/min]
351	0	A. A. A.	1 ³¹	Threshold for contour mon.	<b>fm/</b> min]
352	0	Had.	Jan Star	Contour tolerance	32000 [mm.Test850] [T25.1000]
353	500	S.C.B.	Homan	Dwell time for position monit.	16001 [ms]
354	2400 .	N. S		Velocity comm. value limit	3000 [VEL 0]
355	10	14	44	Circle end-pt. monitor	32000 [m]
356	10	×2.9	H.	Compensation motion limit	32000 [Jm]
357	0	S. C.	tottar	Spindle drift	+ 500 [VELO]
358	0	N. S.		Thread dynamic smoothing exp.	5
359	500	<u></u>		Maximum	<b>99</b> 99
360	1000			sneed	[t/min]
361	2000	6	8	for	6
362	4000	L.	NO.	8 near	NO."
363	4000	A.	100		S. S. S.
364	4000	5	<u>,0``</u>	Tanges	10
365	4000	20	2	Ser .	100
366	4000		2	<u>82</u>	8
371	2000		al de la calendaria de la c	Manual feed	15000
372	10000			Man.rapid trav.	<b>[m/min]</b>
373	10000		<u> </u>	Kef.appr.vel.	6
374	500	NO.	NO.X	INC speed	N ^{OX}
375	2000	18 Mar 19 Mar	- A.	URY feed	20
376	1000	5	10 ⁵	Dwell time for	16000
	Ser.	2	۲. 	spindle inhibit	[#5]
377	0	and the second sec		Minimum spindle motor speed	2048 [VELO]
381				Software editio	ah 3200
385	- 9999999	-Hall		2nd.limit switc X-	• <b>9999</b> 999 (m)

## 14.11 List 2: Common Machine Data

#### 14.12 List 3: Machine Data Bits

## Standard values set with "Input"

3T								
8	Ma	chi	ne	Da	ta	Bit	s	
Nr.	7	5	5	4	3	2	1	0
N 400 S	:	1	1	0	0	1	0	ņ
N 4015	1	1	16	1	0	0	1	1
N 402 S	1	1	1	1	C	0	0	0
N 403S	0	0	0	C	0	С	2	1
N 4045	0	3	0	0	n i	0	0	7
N 405 S	C	0	J)	C	0	Û.	0	
N 406 S	0	0	0	0	0	0	C	0
N 407 S	0	0	0	0	0	1	0	0
N 408 \$	0	0	0	0	1	0	0	1
N 4095	1	0	1	0	0	1	0	0
N 4105		1	1	1	1	1	1	-
N 4115	1	1	0	0	0	0	0	0
N 412S	1	1	0	0	0	0	0	0
N 4135	0	0	0	0	C	0	0	0
N 4145	0	0	0	0	0	0	0	Ô
N 415S	1	0	21	0	1	0	1	0
N 4165	0	0	0	0	0	0	1	10
N 4175	0	0	0	Ó		0	0	0
3M -								

~		Bit									
Nr.	7	6	5	4	3	2	1	0			
				1							
N 4305	1				1	0					
N 4015	1	1	1		0	C	1				
N 4025	1	1	1	1	0	1	1	ŋ			
N 4035	0	0	0	0	Û.	0	0	0			
N 404 S	0	0	0	0	0	0	0	0			
N 4055	0	0	0	0	0	0	0	0			
N 406S	0	0	0	0	0	0	0	0			
N 4075	0	0	0	0	0	0	0	0			
N 4085	0	0	0	0	1	0	0	1			
N 4095	1	0	0	10	0	0	0	0			
N 4105	1	1	1	1	18	1	1	1			
N 4115	1	1	Ō	0	n	0	0	0			
N 412S			0	9		Ū.	0	7			
N 4135	0	0	0	0	0	0	0	0			
N 4145	0	0	0	C	0	0	C	0			
N 4155	1	0	0	0	1	0	0	0			
N 4165	0	C	1	0	0	0	1	10			
N 4175	0	0	0	0	0	0	0.3	0			

See section 2.1

# Enter the set values (do not change preset values)

3	Т	
r-	_	

	_ r	lact	nin	e C	)at	a B	it	ş .
Nr.	Ľ_	6	15	4	3	2	1	<u>_</u>
N 400 S	2		<u> </u>	1	<b>†</b>		1.2	0
N 401 S	5				1	1	.0	
N 4025				1	10	0	1	10
N 4035	0	0	0			201		1
N 4045	0	0	Û,	1		9	1	<b>†</b>
N 4055	0	0	0	10	0	10	0	0
N 4065	0	0	0	0	0	10	1	0
N 4075		0	0	0	0			
N 408S	•		0		11	1	1	
N 409S	1	0	2		Ō		0	
N 4105			1			<u> </u>	1	
N 4115		1/100						N
N 4125	1							011
N 4135	5					Ι	0	
N 4145	0	0	0	0	0	0	0	0
N 4155	1	0	1			200		
N 4165			0	0	0	11	T	
N 4265	0	0	0		220	1	0	

 Bit

 Nr.
 7
 6
 5
 4
 3
 2
 1
 0

 N 400S
 ...
 ...
 ...
 ...
 ...
 ...
 ...

 N 401S
 ...
 ...
 ...
 ...
 ...
 ...
 ...

 N 401S
 ...
 ...
 ...
 ...
 ...
 ...
 ...

 N 401S
 ...
 ...
 ...
 ...
 ...
 ...
 ...

 N 402S
 ...
 ...
 ...
 ...
 ...
 ...
 ...

 N 403S
 0
 0
 0
 ...
 ...
 ...
 ...

 N 405S
 0
 0
 0
 0
 0
 0

 N 405S
 0
 0
 0
 0
 0

 N 405S
 0
 0
 0
 0
 0

 N 405S
 0
 0
 0
 0
 0

 N 413S
 ...
 ...
 ...
 ...

 N 413S
 ...
 ...
 ...
 ...

 N 414S
 0
 0
 0
 0
 ...

 N 416S
 0
 0
 0
 0
 0

,	6	5	Bi	t 3	:2	1	n	Hexa	Designation	Baud rate	Туре
0	0	0	0	1	1	1	1	0 F	Sienens reader	9601	Special devices
0	0 1 1 1	00000	1 0 0 0	) 0 0 0	0	() 1 1 1	0 0 1 1	C 0 C 2 C 3 C 3	TELETYPE ASR 33 SIEMENS PTRO FACIT 4040 FACIT 4070 with M177	110 300 600 600	Universal units
1 1 1 0	1 1 1	0000	000	0 1	1	000	000	C 4 C 4 C C	FACIT 4030 SANYO M2502U FACIT 4208 (cassette)	1209 1209 1200	



